

McGlashan, S. (2002) String rewriting systems and associated finiteness conditions. PhD thesis

http://theses.gla.ac.uk/6190/

Copyright and moral rights for this thesis are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This thesis cannot be reproduced or quoted extensively from without first obtaining permission in writing from the Author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the Author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given.

Glasgow Theses Service http://theses.gla.ac.uk/ theses@gla.ac.uk

String Rewriting Systems and Associated Finiteness Conditions

S. MCGLASHAN

A thesis submitted for the degree of Doctor of Philosophy



UNIVERSITY of GLASGOW

GLASGOW, SCOTLAND, UK JANUARY 2002

Contents

Li	st of	Figures	vi
A	bstra	t v	ii
St	atem	ent vi	iii
A	cknov	ledgements	ix
N	otati	n	x
In	trod	ction	1
1	Pre	minaries	7
	 1.1 1.2 1.3 1.4 	Presentations and the word problem1.1.1Group presentations1.1.2Rewriting systemsRelated complexes	$ \begin{array}{r} 7 \\ 7 \\ 9 \\ 14 \\ 16 \\ 20 \\ 21 \\ 22 \\ 23 \\ 25 \\ 31 \\ 35 \\ 39 \\ 39 \\ \end{array} $
2	λ-cc 2.1 2.2	mplete rewriting systems λ -solvable word problems 2.1.1 λ -Dehn presentations Producing λ -complete rewriting systems from complete rewriting systems 2.2.1 HNN-extensions 2.2.2 Some examples 2.2.3 Bemarks concerning Open Question 2.1.8	41 45 49 49 52
			55

3	A 3-	dimensional complex and its homology	56			
	3.1	A 3-dimensional complex	56			
	3.2	The 2-dimensional homology of the 3-complex $\overline{\mathcal{D}^{\mathbf{p}}}$	58			
	3.3	The basic short exact sequence	61			
	3.4	Killing the left or right action	74			
4	Son	ne new finiteness conditions	81			
	4.1	The properties FDT_2 and FHT_2	81			
	4.2	Invariance properties	83			
		4.2.1 Invariance of finite presentation and trivializer	83			
		4.2.2 Invariance under retraction	84			
	4.3	The properties FDT_2 and FHT_2 for groups $\ldots \ldots \ldots$	85			
	4.4	Monoids with finite complete rewriting systems are FDT_2	85			
		4.4.1 2-cycles at critical triples	86			
		4.4.2 Proof of Theorem 4.4.1	89			
	4.5	An alternative proof of Theorem 4.4.1 using homotopy	90			
		4.5.1 Pictures	91			
		4.5.2 3-cells at critical triples	93			
		4.5.3 Operations on pictures	97			
		4.5.4 A 2-dimensional Dehn-type algorithm	98			
5	Alte	ernative viewpoints	104			
	5.1	Competing finiteness conditions	104			
		5.1.1 An alternative definition of FHT_2	104			
		5.1.2 The property bi- FP_n	105			
		5.1.3 Homological finite derivation type	105			
		5.1.4 A finiteness condition for higher dimensional rewriting systems	106			
	5.2	A complete homotopy reduction system for complete rewriting systems	106			
		5.2.1 Critical pairs of the homotopy reduction system	110			
		5.2.2 Homology	117			
A Appendix			110			
	A.1	Abels' group	119			
Bi	Bibliography 125					

List of Figures

1.1	The Diamond Lemma
1.2	Construction used in Lemma 1.4.17
1.3	Proof of Lemma 1.4.18
_	
3.1	Boundary of the 3-cell $[f, \sigma]$
3.2	A 2-cell of \mathcal{D}
3.3	A cube
3.4	Homologous 2-chains
3.5	Typical 3-cells in $\mathcal{D}^{\mathbf{p}}$
3.6	The subcomplex in Example $3.3.7$
41	A spherical nicture
42	Spherical picture at triple (e_1, e_2, e_3)
43	Construction used in the proof of Lemma 452
4 4	Second construction used in the proof of Lemma 4.5.2
4.5	Spherical picture at a critical triple $\frac{96}{100}$
46	Cancelling nair
47	Bridge move
 1 Q	Insertion of \mathbf{X}_{-} nicture
10	Subplicture after performing bridge movies
4 10	Subjecture after performing bridge moves $\dots \dots \dots$
1 11	Subpliciture composed of the disce labelled σ and σ
4 19	Subjicture composed of the discs labelled δ and $-\delta$
4.12	The maximal region bounded by more than two diase
4.10	The maximal region bounded by more than two discs
4.14	After performing bridge means on Figure 414
4.10	After performing bridge moves on Figure 4.14
4.10	The maximal region is the outer region (case 2) $\ldots \ldots \ldots$
5.1	Single-step homotopy reduction (type 2)
5.2	Resolution of type 1 overlap
5.3	Resolution of type 1 overlap (continued)
5.4	Resolution of type 2 overlap
5.5	Resolution of type 3 overlap
5.6	Critical overlap
5.0	116

Abstract

We begin with an introduction which describes the thesis in detail, and then a preliminary chapter in which we discuss rewriting systems, associated complexes and finiteness conditions. In particular, we recall the graph of derivations Γ and the 2-complex \mathcal{D} associated to any rewriting system, and the related geometric finiteness conditions FDT and FHT. In §1.4 we give basic definitions and results about finite *complete* rewriting systems, that is, rewriting systems which rewrite any word in a finite number of steps to its normal form, the unique irreducible word in its congruence class.

The main work of the thesis begins in Chapter 2 with some discussion of rewriting systems for groups which are confluent on the congruence class containing the empty word. In §2.1 we characterize groups admitting finite λ -complete rewriting systems as those with a λ -Dehn presentation, and in §2.2 we give some examples of finite rewriting systems for groups which are λ -complete but not complete.

For the remainder of the thesis, we study how the properties of finite complete rewriting systems which are discussed in the first chapter are mirrored in higher dimensions. In Chapter 3 we extend the 2-complex \mathcal{D} to form a new 3-complex $\overline{\mathcal{D}P}$, and in Chapter 4 we define new finiteness conditions FDT_2 and FHT_2 based on the homotopy and homology of this complex. In §4.4 we show that if a monoid admits a finite complete rewriting system, then it is of type FDT_2 .

The final chapter contains a discussion of alternative ways to define such higher dimensional finiteness conditions. This leads to the introduction, in §5.2, of a variant of the Guba-Sapir homotopy reduction system which can be associated to any complete rewriting system. This is a rewriting system operating on paths in Γ , and is complete in the sense that it rewrites paths in a finite number of steps to a unique "normal form".

Statement

Chapter 1 covers some basic material including group presentations and rewriting systems, related complexes and finiteness conditions, and also a discussion of complete rewriting systems. Apart from a new proof of Theorem 1.3.4, the material in this chapter can be found elsewhere - references are given in the text.

Chapters 2, 3, 4 and 5 are the original work of the author, with the exception of instances mentioned in the text. The main results of Chapters 3, 4 and 5 are contained in a paper which has been submitted for publication.

Glasgow January, 2002

Stuart McGlashan

Acknowledgements

I have very much enjoyed my time at Glasgow University, and I would particularly like to thank my supervisor, Professor Steve J. Pride FRSE, for giving me the opportunity to work with him, and for his friendly help and guidance.

I would also like to thank my parents and my sister for their love and support.

This work was funded by an Engineering and Physical Sciences Research Council studentship, reference number 98700176.

Notation

Z	the set of integers	page
\mathbb{N}	the set of natural numbers	
\mathbb{N}_0	the set of non-negative integers	
$F = F(\mathbf{x})$	the free monoid on the set \mathbf{x}	7
$\mathcal{G} = \langle \mathbf{x}; \mathbf{r} \rangle$	group presentation	8
$\mathcal{P} = [\mathbf{x}; \mathbf{r}]$	rewriting system	9
$\rightarrow_{\mathbf{r}}$	the single-step relation induced by ${f r}$	9
$\rightarrow^*_{\mathbf{r}}$	the reflexive, transitive closure of $\rightarrow_{\mathbf{r}}$	9
$\longleftrightarrow_{\mathbf{r}}^*$	the Thue congruence generated by \mathbf{r}	9
\overline{w}	the congruence class of a word w	9
$S = S(\mathcal{P})$	the monoid defined by ${\cal P}$	9
$>_{llex}$	the length-lexicographic reduction ordering	10
$\mathcal{F} = \mathcal{F}(\mathbf{x})$	the free group with basis \mathbf{x}	11
$\succ_{\mathbf{r}}$	the reduction ordering induced by ${f r}$	11
$\mathbb{Z}T$	the integral monoid ring of a monoid T	12
J	the kernel of the ring map $\mathbb{Z}F \rightarrow \mathbb{Z}S$ induced by \mathcal{P}	12
$\Gamma = \Gamma(\mathcal{P})$	the graph of derivations	14
e ⁺	the set of all positive edges in Γ	14
$star^+(w)$	the set of positive edges at a vertex w	15
$<_w$	the order of edges in $star^+(w)$	15
$artheta=artheta_{\mathcal{P}}$	the height function associated to ${\cal P}$	16
$\mathcal{D}=\mathcal{D}(\mathcal{P})$	the 2-complex of monoid pictures	16
[e,f]	$\text{2-cell in } \mathcal{D} \ (e,f{\in}\mathbf{e^+})$	16
$\Pi = \Pi(\mathcal{P})$.	the homology bimodule of ${\cal P}$	3. 18
$\Pi^l(\mathcal{P})$	the left homology bimodule of ${\cal P}$	-, 10
$\mathcal{M}=\mathcal{M}(\mathcal{P})$	the relation module of ${\cal P}$	4. 18
\mathcal{M}^l	the left relation module of ${\cal P}$	-, 10

n	ิล	ø	e
2	~	\sim	0

$\mathbb{Z}T.\mathbf{X}.\mathbb{Z}T$	the $(\mathbb{Z}T,\mathbb{Z}T)$ -bimodule with basis X	18
$\mathcal{D}^{\mathbf{p}}$	the 2-complex based on the pair $(\mathcal{D}, \mathbf{p})$,	
	where \mathbf{p} is a set of closed paths	20
Irr(w)	the normal form of w	25
[w; (e, f)]	2-cell in $\mathcal{D}^{\mathbf{p}}$, where \mathbf{p} is a set of closed paths	
	composed of resolutions of critical pairs	35

$G *_{A \cong B}$	an HNN group with base G and associated	
	subgroups A and B	49
$G_{(p,q)}$	the (p,q) -Baumslag-Solitar group	53

[w, p, w']	2-cell in $\mathcal{D}^{\mathbf{p}}$ with boundary $w.p.w'$,	
	where $p \in \mathbf{p}$ and $w, w' \in F$	56
$\overline{\mathcal{D}^{\mathbf{p}}}$	the 3-complex based on the pair (\mathcal{D},\mathbf{p})	56
$[\sigma, e], [e, \sigma]$	3-cells in $\overline{\mathcal{D}^{\mathbf{p}}}$ (σ a 2-cell and $e \in \mathbf{e}^+$)	56
p	the set of 2-cells $\{\hat{p} = [1, p, 1] : p \in \mathbf{p}\}$ in $\mathcal{D}^{\mathbf{p}}$	59
$\Pi_2 = \Pi_2(\mathcal{P},\mathbf{p})$	the second homology bimodule of the pair $(\mathcal{P}, \mathbf{p})$	61
$K^{\mathbf{p}}$	the kernel of the bimodule homomorphism (3.6)	62
Π_2^l	the left $\mathbb{Z}S$ -module $\Pi_2 \otimes_{\mathbb{Z}S} \mathbb{Z}$	-74
$p_l(w)$	the leftmost path from a vertex w	74
P	a picture over $\overline{\mathcal{D}^{\mathbf{p}}}$	90

single-step homotopy reduction	
the reflexive, transitive closure of the relation \rightsquigarrow	107

Introduction

Every group is the homomorphic image of a free group, and as such can be presented as a quotient of a free group by some normal subgroup. Specifying defining relations as elements of the free group which generate this kernel will then give a concise description of the group in terms of *generators and relations*. Combinatorial group theory is concerned with the study of such presentations which may in fact provide very little group theoretic information. The *word problem* asks whether we are even able to determine whether or not two different words in the generators of the group represent the same group element.

1

It was the problem of deciding whether or not two knots are the same that led Dehn to the first statement of the word problem [19], and he was able to give a positive solution for the fundamental groups of closed orientable surfaces by using what is now known as the *Dehn algorithm*. This result has two well-known generalizations. Firstly, Magnus [46] showed that the word problem was solvable for all one-relator groups, and later Greendlinger [25] began the study of more general groups allowing a Dehn algorithm solution of their word problem - these are the *word-hyperbolic* groups of Gromov [26].

Solutions to the word problem of other important classes of groups have also been given, for example for abelian groups and for braid groups [62], but the most famous result is that of Novikov [50] and, independently, Boone [9] that there are groups with unsolvable word problem. Moreover, it is possible to write down presentations of such groups - a presentation involving 10 generators and 29 relations with unsolvable word problem is given in [15, §7.2].

. 1

In his original paper, Dehn had shown that a knot was trivial if and only if the corresponding knot group is abelian¹, a fact which can be easily ascertained if a solution to the word problem is given. Much later Waldhausen [64] showed that knot groups have solvable word problem, and so it is indeed possible to decide whether a given knot is trivial.

String rewriting systems have been much studied in theoretical computer science in semigroup and group theory they can be used to give presentations of monoids. Of particular interest are finite *complete* rewriting systems, that is to say finite rewriting systems that are both terminating and confluent, and so have normal form algorithms which solve their word problem. For many years it was an open question as to whether, conversely, any finitely presented monoid with solvable word problem had a finite complete rewriting system, but Squier [59] proved this to be false. He showed that monoids with a finite complete rewriting system were of type left and right FP_3 , and then gave an example of a finitely presented monoid with solvable word problem which was not of type left FP_3 . In fact, it was implicit in an earlier paper of Anick [3] that monoids admitting finite complete rewriting systems satisfied the stronger finiteness conditions left and right FP_{∞} . This result has also been proved by Groves [27], Kobayashi [36], and Brown [12].

This thesis begins with a preliminary chapter, Chapter 1, which reviews these concepts in detail, together with other finiteness conditions for rewriting systems which are discussed below. Proofs of results in this chapter are given either where similar ideas are used later in the thesis, or where references are not easily available. Chapter 2 is a short chapter which contains some new results on λ -complete rewriting systems, which contain a solution to their word problem when they define groups. In particular, Proposition 2.1.14 characterizes groups with finite λ -complete rewriting systems as those with λ -Dehn presentations, a generalization of the standard notion of a Dehn presentation. We also use Britton's Lemma to construct some interesting

¹Dehn's result relied on an incorrect proof of what is now called Dehn's Lemma, and which was finally proved much later by Papakyriakopoulos [53].

examples of such rewriting systems.

However, the main aim of this thesis is to develop the following ideas one dimension higher. Squier [60] introduced the class of monoids of *finite derivation type (FDT)* which contains each monoid with a finite complete rewriting system. It was then shown by various authors [17, 43, 54] that *FDT* monoids were necessarily of type left and right FP_3 , and also [18, 55] that for finitely presented groups these conditions were equivalent. A related property *finite homological type* (*FHT*) was introduced by Wang and Pride [66].

Roughly speaking, the properties FDT and FHT are defined as follows. If \mathcal{P} is a finite rewriting system for a monoid S consisting of a set of rules \mathbf{r} on an alphabet \mathbf{x} , then there is a certain 2-complex $\mathcal{D} = \mathcal{D}(\mathcal{P})$ (the Squier complex or 2-complex of monoid pictures) derived from \mathcal{P} on which the free monoid F on the set \mathbf{x} acts on both the left and the right. We require that the first homotopy or first homology of \mathcal{D} is finitely based, that is to say, there is a finite set \mathbf{p} of closed paths in \mathcal{D} such that attaching 2-cells to the paths $F.\mathbf{p}.F$ gives a 2-complex $\mathcal{D}^{\mathbf{p}}$ which is simply-connected (FDT) or has trivial first homology (FHT).

Both these properties are monoid invariants, that is, they are independent of the choice of finite rewriting system for S [60, 66]. Also, a retract of an FDT or FHT monoid will have the same property [66]. (Recall that a retract R of a monoid S is (up to isomorphism) a submonoid of S such that there is a homomorphism (a retraction) of S onto R which fixes R elementwise.) From the definitions, it is immediate that an FDT monoid is FHT. An FHT monoid is also of type left and right FP_3 [66], but, on the other hand, Kobayashi and Otto [42] have given an example of a finitely presented monoid of type left and right FP_3 which is not FHT. It remains a major open question as to whether FHT is equivalent to FDT^2 [41, 66] - for groups the two are equivalent.

The first homology group of \mathcal{D} is in fact a natural $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodule (the homology bimodule of \mathcal{P}) which we denote by $\Pi = \Pi(\mathcal{P})$, and the FHT property is just the

²Pride and Otto have recently found an example of a monoid which is FHT but not FDT.

assertion that this bimodule is finitely generated. There is an important short exact sequence

$$0 \longrightarrow \Pi \xrightarrow{\eta} \mathbb{Z}S.\mathbf{r}.\mathbb{Z}S \longrightarrow \mathcal{M}(\mathcal{P}) \longrightarrow 0 \tag{1}$$

of $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodules involving II. Here $\mathbb{Z}S.\mathbf{r}.\mathbb{Z}S$ denotes the free $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodule with basis \mathbf{r} , and $\mathcal{M}(\mathcal{P})$ is the relation bimodule of \mathcal{P} introduced by Ivanov [32]. The sequence (1) was introduced by Pride [54], apart from the injectivity of η which was proved by Guba and Sapir [29]. An alternative proof has been given by Otto and Kobayashi [40], where it is also shown that the sequence remains exact upon killing either the right or left S-action. To be more specific, applying the tensor $-\otimes_{\mathbb{Z}S}\mathbb{Z}$ to (1) will preserve the injectivity of η , that is, the left $\mathbb{Z}S$ -module homomorphism

$$(\eta \otimes 1) : \Pi^{l} \longrightarrow \mathbb{Z}S.\mathbf{r}$$
⁽²⁾

is injective, where $\Pi^l = \Pi \otimes_{\mathbb{Z}S} \mathbb{Z}$ is the *left homology module* of \mathcal{P} and $\mathbb{Z}S.\mathbf{r}$ denotes the free left $\mathbb{Z}S$ -module with basis \mathbf{r} which is naturally isomorphic to $\mathbb{Z}S.\mathbf{r}.\mathbb{Z}S \otimes_{\mathbb{Z}S} \mathbb{Z}$ [40]. We therefore have a short exact sequence

$$0 \longrightarrow \Pi^{l} \xrightarrow{\eta \otimes 1} \mathbb{Z} S.\mathbf{r} \longrightarrow \mathcal{M}^{l} \longrightarrow 0$$
(3)

of left ZS-modules, where $\mathcal{M}^{l}(\mathcal{P}) = \mathcal{M}^{l} = \mathcal{M} \otimes_{\mathbb{Z}S} \mathbb{Z}$ is the *left relation module* [32]. An analogous exact sequence of right ZS-modules is obtained by applying $\mathbb{Z} \otimes_{\mathbb{Z}S} -$.

For finitely presented groups the properties FDT, FHT and left- FP_3 are equivalent, and in §1.3.4 we give a new proof of the equivalence of FHT and left- FP_3 .

Main results

We begin Chapter 3 by constructing a new 3-complex $\overline{\mathcal{D}^{\mathbf{p}}}$ by attaching 3-cells to certain obvious spherical subcomplexes in $\mathcal{D}^{\mathbf{p}}$. It turns out that the second homology group of this 3-complex is also a $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodule, which we denote by $\Pi_2 = \Pi_2(\mathcal{P}, \mathbf{p})$, and this module will play the role of Π one dimension higher.

We obtain a short exact sequence analogous to (1):

Theorem 3.3.1 If the homology classes of the paths in \mathbf{p} give rise to a set of bimodule generators of Π then there is a short exact sequence

$$0 \longrightarrow \Pi_2 \xrightarrow{\Phi} \mathbb{Z}S.\mathbf{p}.\mathbb{Z}S \xrightarrow{\nu} \Pi \longrightarrow 0 \tag{4}$$

of $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodules.

Furthermore, we shall show in $\S3.4$ that the analogue of (2) is injective, giving the following short exact sequence analogous to (1)

Theorem 3.4.1 If the homology classes of the paths in \mathbf{p} give rise to a set of bimodule generators of Π then there is a short exact sequence

 $0 \longrightarrow \Pi_2^l \longrightarrow \mathbb{Z}S.\mathbf{p} \longrightarrow \Pi^l \longrightarrow 0 \tag{5}$

of left $\mathbb{Z}S$ -modules, where $\Pi_2^l = \Pi_2 \otimes_{\mathbb{Z}S} \mathbb{Z}$.

In Chapter 4 we introduce new finiteness conditions analogous to FDT and FHT. Roughly speaking, S is FDT_2 (respectively FHT_2) if it is FDT (respectively, FHT) and if for some finite rewriting system \mathcal{P} and finite homotopy (respectively, homology) trivializer **p** the second homotopy (respectively, homology) is finitely based. There are certain subtleties involved in these definitions which we shall discuss in §5.1, where we also describe alternatives.

In §4.2.1 and §4.2.2 respectively, we prove some invariance properties:

Theorem 4.2.1 The properties FDT_2 and FHT_2 are monoid invariants (that is, they are independent of the choice of finite rewriting system and finite trivializer).

Theorem 4.2.2 Any retraction of an FDT₂ or FHT₂ monoid has the same property.

It is clear that an FDT_2 monoid is also of type FHT_2 , and it turns out that for FDT monoids the two properties FDT_2 and FHT_2 are in fact equivalent (see Remark 4.1.3). Consequently, if the properties FHT and FDT turn out to be equivalent, then the properties FDT_2 and FHT_2 will be also.

Note that for finitely presented groups the properties FDT_2 and FHT_2 are equivalent. In fact (see §4.3),

Theorem 4.3.1 For finitely presented groups the properties FDT_2 , FHT_2 and FP_4 are all equivalent.

This result is obtained by repeating the proof of the analogous result of the equivalence of the properties FHT and left- FP_3 given in §1.3.4: here we use the fact that for an FDT group G, Π_2 is finitely generated as a ($\mathbb{Z}G, \mathbb{Z}G$)-bimodule if and only if Π_2^l is finitely generated as a left $\mathbb{Z}G$ -module.

In §4.4 we give proofs of the following result.

Theorem 4.4.1 A monoid S which admits a finite complete rewriting system is of type FDT_2 (and of type FHT_2).

Thus our 3-complex $\overline{\mathcal{D}^{\mathcal{P}}}$ and the correspondingly defined finiteness conditions would seem to be correct, in that the results mirror exactly those one dimension lower, which concern the properties FDT and FHT. However, there are certain subtleties involved in these definitions of FDT_2 and FHT_2 which we discuss in §5.1, where we also describe alternatives. We also describe how the properties FDT_2 and FHT_2 relate to finiteness conditions introduced elsewhere, namely *n*-dimensional homological finite derivation type [2] and the property bi- FP_n [41]. In particular, Kobayashi and Otto have characterized FHT monoids as finitely presented monoids of type bi- FP_3 [41], and there is an analogous characterization of FHT_2 monoids as finitely presented monoids of type bi- FP_4 .

Finally, the possibility of defining FDT_2 by studying the graph of derivations of Γ leads us to a discussion in §5.2 of a *complete homotopy reduction system* which we associate to complete rewriting systems. The critical pairs of this higher dimensional rewriting system (a variant of the Guba-Sapir reduction system [29, 39]) are shown to correspond to the critical triples of the original rewriting system.

Chapter 1 Preliminaries

1.1 Presentations and the word problem

A word w on an alphabet x is a finite sequence of elements of x. The length |w| of w is just the length of the sequence, and we shall use 1 to denote the word of length zero (the *empty word*). The concatenation of two words w_1 and w_2 is written w_1w_2 , and the *free monoid* F = F(x) on x consists of all words on x together with this multiplication.

1.1.1 Group presentations

In combinatorial group theory, to an alphabet \mathbf{x} of generators we associate a set

$$\mathbf{x}^{-1} = \{x^{-1} : x \in \mathbf{x}\}$$

of formal inverses, and consider words on the alphabet $\mathbf{x} \cup \mathbf{x}^{-1}$. For any such word

$$w = x_1^{\varepsilon_1} x_2^{\varepsilon_2} \dots x_n^{\varepsilon_n}$$

where $x_i \in \mathbf{x}$ and $\varepsilon_i = \pm 1$ for each j = 1, 2, ..., n, we use w^{-1} to denote the word

$$x_n^{-\varepsilon_n}x_{n-1}^{-\varepsilon_{n-1}}\ldots x_1^{-\varepsilon_1}.$$

Also, we shall say that w is *freely reduced* if it contains no subword of the form $x^{\epsilon}x^{-\epsilon}$ $(x \in \mathbf{x}, \epsilon = \pm 1)$, and cyclically reduced if moreover $x_1^{\epsilon_1}$ and $x_n^{-\epsilon_n}$ are distinct. A group presentation is a pair $\mathcal{G} = \langle \mathbf{x}; \mathbf{r} \rangle$ consisting of a set of generators \mathbf{x} and a set \mathbf{r} of words in \mathbf{x} , the relators. Two words w_1 and w_2 are said to be equivalent if one can be converted to the other by a finite sequence of operations of the following type, together with their inverses:

1. replace a word uv with $ux^{\varepsilon}x^{-\varepsilon}v$ where $x \in \mathbf{x}$ and $\varepsilon = \pm 1$;

2. replace a word uv with urv where $r \in \mathbf{r}$.

The equivalence classes form a group (see [47, Theorem 1.1]), the group defined by \mathcal{G} , where the multiplication is that induced by concatenation of words. We say that \mathcal{G} is a presentation for any group which is in the same isomorphism class as the group it defines.

A group presentation is called finite if both \mathbf{x} and \mathbf{r} are finite sets, and its *word* problem is described as follows:

for any pair of words w_1 and w_2 in the generators \mathbf{x} , decide whether they are in the same equivalence class.

A Tietze transformation of \mathcal{G} takes one of the following forms:

type I: Replace \mathcal{G} by $\langle \mathbf{x} ; \mathbf{r} \cup \{w\} \rangle$, where w is a word on $\mathbf{x} \cup \mathbf{x}^{-1}$ equivalent to 1.

type II: Replace \mathcal{G} by $\langle \mathbf{x} \cup \{z\}$; $\mathbf{r} \cup \{z^{-1}w\}\rangle$, where z is a new letter not in x, and w is some word on $\mathbf{x} \cup \mathbf{x}^{-1}$.

The Tietze Theorem [33, p.49] states that two finite group presentations define isomorphic groups if and only if they are *Tietze equivalent*, that is one can be obtained from the other by a finite number of Tietze transformations and their inverses. If the word problem is solvable for some finite presentation of a group G, then it is also solvable for any other finite presentation of G [15, Corollary 1.1.10], and so we can speak of (finitely presented) groups with solvable word problem.

In this thesis, however, we shall mostly use *rewriting systems* to present groups and, more generally, monoids.

1.1.2 Rewriting systems

A rewriting system is a pair $\mathcal{P} = [\mathbf{x}; \mathbf{r}]$ where \mathbf{x} is a set of letters and \mathbf{r} is a set of rules of the form $r = (r_{+1}, r_{-1}) \in F \times F$, where $F = F(\mathbf{x})$ denotes the free monoid on \mathbf{x} . When displaying a rewriting system we shall generally write r in the form $r_{+1} = r_{-1}$. We say that \mathcal{P} is finite if both \mathbf{x} and \mathbf{r} are finite. The corresponding single-step reduction relation

$$\rightarrow_{\mathbf{r}} = \{(wr_{+1}w', wr_{-1}w') : r \in \mathbf{r} \text{ and } w, w' \in F\}$$

on elements of F is given by rewriting words according to the set of rules. The reflexive and transitive closure of this relation is denoted by $\rightarrow_{\mathbf{r}}^{*}$, and the reflexive, transitive and symmetric closure $\leftrightarrow_{\mathbf{r}}^{*}$ is the *Thue congruence* generated by \mathbf{r} . Two rewriting systems on the same alphabet are said to be *equivalent* if the two sets of rules generate the same Thue congruence.

We denote the congruence class of $w \in F$ by \overline{w} , so that the monoid $S = S(\mathcal{P})$ defined by \mathcal{P} consists of these congruence classes with the multiplication

 $\overline{u}.\overline{v} = \overline{u}\overline{v} \qquad (u, v \in F).$

We say that a monoid T is presented by \mathcal{P} if T is isomorphic to S, and is finitely presented if it can be presented by some finite rewriting system.

The Tietze Theorem is easily extended to rewriting systems. A *Tietze transformation* of \mathcal{P} takes one of the following forms:

type I: Replace \mathcal{P} by $[\mathbf{x}; \mathbf{r} \cup \{u = v\}]$, where $u, v \in F$ satisfy $u \leftrightarrow_{\mathbf{r}}^* v$.

type II: Replace \mathcal{P} by $[\mathbf{x} \cup \{z\}; \mathbf{r} \cup \{z = w\}]$ where z is a new symbol not in x, and $w \in F(\mathbf{x})$.

Again, two finite rewriting systems present the same monoid if and only if they are *Tietze equivalent*, that is one can be obtained from the other by a finite number of Tietze transformations and their inverses (see, for example, [60, Proposition 4.2]).

Reduction orderings

We call an irreflexive and transitive relation \succ on a set A a partial order. A total order of A is a partial order such that for distinct elements $a_1, a_2 \in A$ one of $a_1 \succ a_2$ or $a_2 \succ a_1$ holds. We say that a partial order is noetherian if there is no infinite sequence

 a_1, a_2, a_3, \ldots

of elements of A satisfying $a_i \succ a_{i+1}$ for each i = 1, 2, 3, ...

Let \succ be some partial ordering of the free monoid F which respects the multiplication of F, so that for all $w, u, v, z \in F$ with $u \succ v$ we have $wuz \succ wvz$. If \succ is noetherian then we say that it is a *reduction* order on F. We shall say that a set of rules \mathbf{r} is *compatible* with a reduction ordering \succ if for each $r \in \mathbf{r}$ the relation $r_{+1} \succ r_{-1}$ holds.

We shall need the following reduction orders on free monoids:

Definition 1.1.1 The *length-reducing order* is the reduction order on F given by writing $u \succ v$ if and only if |u| > |v| for any $u, v \in F$.

Definition 1.1.2 Let \triangleright be a total order on the alphabet **x**. The corresponding *length-lexicographical order* is the reduction ordering of F defined as follows. For $u, v \in F$ write $u >_{llex} v$ if and only if |u| > |v|, or |u| = |v|, u = au' and v = bv', where $a, b \in \mathbf{x}$ and $u', v' \in F$, and one of the following holds:

1. *a*⊳*b*

2. a = b and $u' >_{llex} v'$

Note that $>_{llex}$ is a total order.

Definition 1.1.3 (Dershowitz [21]) Let \triangleright be a noetherian order on a set \mathbf{x} . The corresponding *recursive path ordering from the left* is the reduction order on F defined as follows. For $u, v \in F$ write $u \succ v$ if and only if $u \neq 1$ and v = 1, or u = au' and v = bv', where $a, b \in \mathbf{x}$ and $u', v' \in F$, and one of the following holds:

1. $a \triangleright b$ and $au' \succ v'$

2.
$$a = b$$
 and $u' \succ v'$

3. $u' \succeq bv'$

Noetherian and complete rewriting systems

A rewriting system $\mathcal{P} = [\mathbf{x}; \mathbf{r}]$ is said to be *noetherian* if there are no infinite chains $w \rightarrow_{\mathbf{r}} w' \rightarrow_{\mathbf{r}} \cdots$, and such a rewriting system is said to be *complete* if each congruence class contains a unique irreducible word, sometimes called a *normal form*. We can then view the monoid defined by the rewriting system as the set of normal forms, where the multiplication is given by concatenation followed by a reduction to the irreducible.

Example 1.1.4 (Newman [49]) The rewriting system

$$[x, x^{-1} (x \in \mathbf{x}); xx^{-1} = 1, x^{-1}x = 1 \quad (x \in \mathbf{x})]$$
(1.1)

is a complete rewriting system presenting the free group $\mathcal{F}(\mathbf{x})$ with basis \mathbf{x} . The normal forms are just the set of words which are freely reduced.

It is clear that any monoid with a finite complete rewriting system has solvable word problem, as any two words are in the same congruence class if and only if they have the same normal form¹.

Definition 1.1.5 If \mathcal{P} is noetherian, then the partial order $\succ_{\mathbf{r}}$ on F given by writing

$$wr_{+1}z \succ_{\mathbf{r}} wr_{-1}z$$

for each $r \in \mathbf{r}$ and $w, z \in F$ is a reduction order, the reduction ordering induced by \mathbf{r} .

¹It was remarked in the Introduction that Knot Theory provided important motivation for the study of the word problem. It is interesting to note, therefore, that the Freyd-Yetter proof of the existence of the knot invariant known as the HOMFLY polynomial [23] uses a complete set of rewrite rules to obtain the polynomial "normal form".

Integral monoid rings, and the ideal J

For any monoid T, the integral monoid ring $\mathbb{Z}T$ consists of all formal sums

$$\sum_{t \in T} \alpha_t t \qquad (\alpha_t \in \mathbb{Z} \text{ non-zero for only finitely many } t \in T),$$

with the natural multiplication

$$(\sum_{t\in T} lpha_t t)(\sum_{t\in T} eta_t t) = \sum_{t,t'\in T} lpha_t eta_{t'} tt'$$

induced by that of T.

Let F be the free monoid on the alphabet of some rewriting system defining the monoid S. The natural epimorphism

$$F \longrightarrow S \qquad w \mapsto \bar{w}$$

extends linearly to a ring epimorphism

$$\mathbb{Z}F \longrightarrow \mathbb{Z}S$$

of the associated integral monoid rings. We will use J to denote the kernel of this epimorphism. The following lemma is well-known.

Lemma 1.1.6

1. As an abelian group, J is generated by all elements of the form

$$u(r_{+1} - r_{-1})v$$
 $(u, v \in F, r \in \mathbf{r}).$ (1.2)

2. As a $(\mathbb{Z}F,\mathbb{Z}F)$ -bimodule J is generated by the elements $(r_{+1} - r_{-1})$, where r is in the set of rules **r**.

Proof ([40]): Part 2 follows immediately from part 1.

$$\xi = \sum_{i=1}^n \alpha_i w_i \; ,$$

where $w_i \in F$ and α_i is a non-zero integer for each i = 1, 2, ..., n, and if $i \neq j$ then $w_i \neq w_j$. We shall proceed by induction on the value

$$T(\xi) = \sum_{i=1}^{n} |\alpha_i| \, .$$

Firstly, $T(\xi) = 0$ if and only if $\xi = 0$. Suppose that $T(\xi) > 0$. Since

$$\sum_{i=1}^{n} \alpha_i \overline{w}_i = 0$$

we can choose a pair of integers $0 < j, k \le n$ such that $\alpha_j > 0, \alpha_k < 0$ and $\overline{w}_j = \overline{w}_k$, and so there is a sequence of the form

$$w_j = u_1(r_1)_{\epsilon_1}v_1, \ u_1(r_1)_{-\epsilon_1}v_1 = u_2(r_2)_{\epsilon_2}v_2, \ \dots, \ u_m(r_m)_{\epsilon_m}v_m = w_k$$

describing a derivation from w_j to w_k , where $u_i, v_i \in F$, $r_i \in \mathbf{r}$ and $\varepsilon_i = \pm 1$ for i = 1, 2, ..., m. Then

$$w_j - w_k = \sum_{i=1}^m \varepsilon_i u_i ((r_i)_{+1} - (r_i)_{-1}) v_i$$

is in the subgroup generated by elements of the form (1.2). So is $\xi - (w_j - w_k)$ by induction, since $T(\xi - (w_j - w_k)) = T(\xi) - 2$, and therefore ξ is also. This completes the proof.

In the special case where S is the trivial monoid we denote by I the kernel of the augmentation mapping

$$\mathbb{Z}F \longrightarrow \mathbb{Z}$$
 $w \mapsto 1$ $(w \in F)$.

1.2 Related complexes

1.2.1 The graph of derivations

The graph of derivations $\Gamma = \Gamma(\mathcal{P})$ associated to \mathcal{P} is a geometric interpretation of the rewriting system, and is constructed as follows. The vertex set is the free monoid F, and the edge set consists of all quadruples of the form

 $e = (w, r, \varepsilon, w')$ $w, w' \in F, r \in \mathbf{r} \text{ and } \varepsilon = \pm 1$

with initial, terminal and inverse functions

$$\iota e = wr_{\varepsilon}w', \quad \tau e = wr_{-\varepsilon}w' \text{ and } e^{-1} = (w, r, -\varepsilon, w').$$

The edge e is called *positive* if $\varepsilon = 1$, and we shall denote by e^+ the set of all positive edges. A path p of length n in Γ is a sequence $p = e_1 e_2 \dots e_n$ of edges with $\iota e_{i+1} = \tau e_i$ $(i = 1, \dots, n-1)$, and it is positive if it is composed of positive edges. We say that p is closed if $\iota e_1 = \tau e_n$.

Sometimes it will be convenient to depict the positive edge e as

$$w \underline{r}_{+1} w' \to w r_{-1} w' ,$$

where the subword being rewritten is underlined. As an illustration, in the graph Γ of the rewriting system (1.1) in Example 1.1.4 the edge $(x, (xx^{-1} = 1), +1, x)$ is more clearly described by writing

$$x \underline{x} \underline{x}^{-1} x \to x^2$$

There is a natural two-sided action of F on Γ . The action on the vertices is given by left and right multiplication in F, and for an edge $e = (w, r, \varepsilon, w')$ and $z, z' \in F$ we define

$$z.e.z' = (zw, r, \varepsilon, w'z'),$$

and this action extends to paths.

Note that there is a positive edge from u to v $(u, v \in F)$ if and only if $u \to_{\mathbf{r}} v$, and a path in Γ from u to v describes a *derivation* in F rewriting u to v using the rules in \mathbf{r} and their inverses. Thus u and v lie in the same connected component of Γ if and only if $u \leftrightarrow_{\mathbf{r}}^* v$, and so there is a one-to-one correspondence between the connected components of Γ and the elements of S. Note also that the abelian group generators (1.2) of J are precisely the elements $\iota e - \tau e$ $(e \in \mathbf{e}^+)$.

We will denote the set of positive edges with initial vertex $w \in F$ by $star^+(w)$, and we order these edges as follows: If e = (u, r, +1, v) and f = (u', r', +1, v') where $u, u', v, v' \in F$ and $r, r' \in \mathbf{r}$ with $\iota e = \iota f = w$, then write $e <_w f$ if:

- 1. u is a proper prefix of u'; or
- 2. u = u' and r_{+1} is a proper prefix of r'_{+1} ; or
- 3. $u = u', r_{+1} = r'_{+1}$ and $r_{-1} <_{llex} r'_{-1}$ in some chosen length-lexicographical order $<_{llex}$ on F.

In particular, each non-irreducible vertex has a least outgoing edge which, following Guba and Sapir [29, Definition 9.1], we call the *left principal* edge. We say that two positive edges e_1, e_2 with the same initial vertex are *disjoint* if they can be written in the form

$$e_1 = f_1 \iota f_2, \qquad e_2 = \iota f_1 \cdot f_2$$

for some positive edges f_1, f_2 . Also, we say that an edge e is *left-reduced* (respectively, *right-reduced*) if it cannot be written in the form u.f (respectively, f.u) for some nonempty word $u \in F$ and edge f.

Example 1.2.1 Consider the vertex $w = x^2 x^{-1} x^3 x^{-1}$ in the graph Γ of the rewriting system (1.1) in Example 1.1.4. The set $star^+(w)$ consists of the edges

$$x\underline{xx}^{-1}x^3x^{-1} \rightarrow x^4x^{-1}, \ x^2\underline{x}^{-1}\underline{x}x^2x^{-1} \rightarrow x^4x^{-1}, \text{ and } x^2x^{-1}\underline{x}^2\underline{xx}^{-1} \rightarrow x^2x^{-1}x^2$$

which for convenience we label e, f and g respectively. These edges are ordered as $e <_w f <_w g$, so that e is the left principal edge. The edge g is disjoint from both e and f, and is, moreover, right-reduced.

The following well-known result is related to König's Lemma:

Lemma 1.2.2 If \mathcal{P} is noetherian, then for any vertex w in Γ there is a bound on the length of any positive path originating at w.

Proof: Otherwise, we may choose $w_1 \in F$ with $w \to_{\mathbf{r}} w_1$ and with no bound on the lengths of paths originating at w_1 . In the same way we may choose some $w_2 \in F$ with $w_1 \to_{\mathbf{r}} w_2$ and no bound on the lengths of paths originating at w_2 , and so on, producing an infinite reduction sequence $w \to_{\mathbf{r}} w_1 \to_{\mathbf{r}} w_2 \to_{\mathbf{r}} \ldots$, a contradiction.

Definition 1.2.3 By Lemma 1.2.2, we can define the following height function $\vartheta = \vartheta_{\mathcal{P}}$ (called the *disorder function* in [28]), which we use in §4.4 and §5.2.

 $\vartheta : F \longrightarrow \mathbb{N}$

 $w \mapsto \max$. length of all paths originating at w.

1.2.2 The 2-complex \mathcal{D} and its homology

The 2-complex $\mathcal{D} = \mathcal{D}(\mathcal{P})$ associated to \mathcal{P} is the combinatorial 2-complex with 1-skeleton Γ , to which, for each pair of positive edges e and f, a 2-cell [e, f] is attached along the closed path

$$\partial[e,f] = (e.\iota f)(\tau e.f)(e^{-1}.\tau f)(\iota e.f^{-1}).$$

Remark: In the usual definition of the Squier complex, for each pair of positive edges e and f we also attach 2-cells $[e, f^{-1}]$, $[e^{-1}, f]$ and $[e^{-1}, f^{-1}]$ along closed paths

$$\begin{array}{lll} \partial [e,f^{-1}] & = & (e.\tau f)(\tau e.f^{-1})(e^{-1}.\iota f)(\iota e.f) \\ \\ \partial [e^{-1},f] & = & (e^{-1}.\iota f)(\iota e.f)(e.\tau f)(\tau e.f^{-1}) \\ \\ \partial [e^{-1},f^{-1}] & = & (e^{-1}.\tau f)(\iota e.f^{-1})(e.\iota f)(\tau e.f). \end{array}$$

But the additional cells here are attached along paths which are cyclic permutations of $(\partial [e, f])^{\pm 1}$, and since we are interested in the homology and homotopy of this complex it is unnecessary to include them: In the presence of [e, f] only, the attaching paths of the others are already null-homotopic.

Squier [60] in fact introduced a homotopy relation on Γ which coincides with homotopy of this 2-complex: \mathcal{D} is properly "the 2-complex of monoid pictures" introduced independently by both Pride [54] and Kilibarda [34]. The term "Squier complex" was introduced by Guba and Sapir [29, Introduction].

The two sided action of F on Γ extends naturally to the 2-cells by

w.[e, f].w' = [w.e, f.w'] $(w, w' \in F, e, f \text{ positive edges}).$

Homology

We have the chain complex

$$\mathbf{C}(\mathcal{D}): \qquad 0 \longrightarrow C_2 \xrightarrow{\partial_2} C_1 \xrightarrow{\partial_1} C_0 \longrightarrow 0 \tag{1.3}$$

where the chain groups C_0 , C_1 and C_2 are, respectively, the free abelian groups with bases F, the set of all positive edges, and the set of all 2-cells, and the boundary maps are

 $\partial_1 e = \iota e - \tau e$ (e a positive edge)

and

 $\partial_2[e, f] = e.(\iota f - \tau f) - (\iota e - \tau e)f$ (e, f positive edges).

The chain groups are $(\mathbb{Z}F, \mathbb{Z}F)$ -bimodules where the *F*-action is inherited from the two-sided action of *F* on \mathcal{D} and, moreover, the boundary maps respect this action. It follows that the first homology group $H_1(\mathcal{D})$ is also a $(\mathbb{Z}F, \mathbb{Z}F)$ -bimodule. It can be shown [54, Lemma 4.1] that

$$J.H_1(\mathcal{D}) = H_1(\mathcal{D}).J = 0 \tag{1.4}$$

and so $H_1(\mathcal{D})$ has an induced $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodule structure. This is the homology bimodule of \mathcal{P} , denoted by $\Pi = \Pi(\mathcal{P})$. In the short exact sequence (1) of $(\mathbb{Z}S, \mathbb{Z}S)$ bimodules Π is exhibited as the kernel of a presentation of the relation bimodule $\mathcal{M}(\mathcal{P}) = J/J^2$ of \mathcal{P} .

If we use (\mathcal{D}, s) to denote the connected component of \mathcal{D} corresponding to the element s of S, and write $\Pi_s = H_1(\mathcal{D}, s)$, then we have the (abelian group) decomposition

$$\Pi = \bigoplus_{s \in S} \Pi_s. \tag{1.5}$$

For $a, b \in S$ we then have $a.\Pi_s.b \subseteq \Pi_{asb}$, with equality if a and b are units.

In particular if S is a group then we have

$$a.\Pi_{s.}b = \Pi_{asb} \text{ for all } a, s, b \in S.$$

$$(1.6)$$

We shall require this in our proof of Theorem 1.3.4.

Bimodules

For a set X and monoid T, we shall use $\mathbb{Z}T.X.\mathbb{Z}T$ to denote the free $(\mathbb{Z}T,\mathbb{Z}T)$ bimodule with basis X. The underlying abelian group is free with basis

 $\{t.x.t':t,t'\in T \text{ and } x\in \mathbf{X}\},\$

on which the two-sided T-action is given by writing

s.(t.x.t').s' = st.x.t's'

for any $s, s' \in T$.

The following lemma (and proof) is from a paper of Kobayashi and Otto [40, Lemma 6.1].

Lemma 1.2.4 Let S be the monoid defined by some rewriting system, let F be the free monoid on its alphabet, and let $\mathbb{Z}F.\mathbf{X}.\mathbb{Z}F$ and $\mathbb{Z}S.\mathbf{X}.\mathbb{Z}S$ be the free $(\mathbb{Z}F,\mathbb{Z}F)$ and $(\mathbb{Z}S,\mathbb{Z}S)$ bimodules on some set \mathbf{X} . If we consider $\mathbb{Z}S.\mathbf{X}.\mathbb{Z}S$ as a $(\mathbb{Z}F,\mathbb{Z}F)$ -bimodule, with F acting via the epimorphism $F \longrightarrow S$, then we have a natural epimorphism

 $\nu: \mathbb{Z}F.\mathbf{X}.\mathbb{Z}F \longrightarrow \mathbb{Z}S.\mathbf{X}.\mathbb{Z}S \qquad w.x.z \longmapsto \overline{w}.x.\overline{z} \qquad (x \in \mathbf{X}, w, z \in F)$ (1.7)

of $(\mathbb{Z}F,\mathbb{Z}F)$ -bimodules. The kernel $Ker(\nu)$ of (1.7) can be written as

 $J.\mathbf{X}.\mathbb{Z}F + \mathbb{Z}F.\mathbf{X}.J$.

Proof: It is clear that the inclusion $J.X.\mathbb{Z}F + \mathbb{Z}F.X.J \subseteq Ker(\nu)$ holds, so we must show that the converse inclusion is also true.

If we choose for each congruence class \overline{w} a representative element \hat{w} , then we can define homomorphisms

$$\theta: \mathbb{Z}S.\mathbf{X}.\mathbb{Z}S \longrightarrow \mathbb{Z}F.\mathbf{X}.\mathbb{Z}F \qquad \overline{w}.x.\overline{z} \longmapsto \hat{w}.x.\hat{z} \qquad (x \in \mathbf{X}, w, z \in F)$$

and

 $\phi: \mathbb{Z}F.\mathbf{X}.\mathbb{Z}F \longrightarrow \mathbb{Z}F.\mathbf{X}.\mathbb{Z}F \qquad w.x.z \longmapsto w.x.z - \hat{w}.x.\hat{z} \qquad (x \in \mathbf{X}, w, z \in F).$

of abelian groups. Note that $(\theta \nu + \phi)$ is the identity on $\mathbb{Z}F.\mathbf{X}.\mathbb{Z}F$, mapping a basis element w.x.z $(w, z \in F, x \in \mathbf{X})$ to

$$\theta(\overline{w}.x.\overline{z}) + w.x.z - \hat{w}.x.\hat{z} = \hat{w}.x.\hat{z} + w.x.z - \hat{w}.x.\hat{z} = w.x.z.$$

Now because, for any $x \in \mathbf{X}$ and $w, z \in F$,

$$\phi(w.x.z) = w.x.z - \hat{w}.x.\hat{z} \\ = (w - \hat{w}).x.z + \hat{w}.x.(z - \hat{z})$$

we see that $Im(\phi) \subseteq J.\mathbf{X}.\mathbb{Z}F + \mathbb{Z}F.\mathbf{X}.J$. But for any $\xi \in Ker(\nu)$,

$$\xi = (\theta \nu + \phi)\xi = \phi \xi ,$$

and therefore $\xi \in J.X.\mathbb{Z}F + \mathbb{Z}F.X.J$.

Ц

1.3 Finiteness conditions

1.3.1 Finite derivation type (FDT) and finite homological type (FHT)

Let **p** be a set of closed paths in \mathcal{D} . We shall denote by $\mathcal{D}^{\mathbf{p}}$ the 2-complex obtained by attaching 2-cells [w, p, w'] along all closed paths of the form

$$\partial[w, p, w'] = w.p.w'$$
 $(p \in \mathbf{p}, w, w' \in F).$

A rewriting system \mathcal{P} is said to be of finite derivation type (FDT) (respectively, finite homological type (FHT)) if it is finite, and if there is some finite set \mathbf{p} (a homotopy (respectively, homology) trivializer) of closed paths in \mathcal{D} such that the 2complex $\mathcal{D}^{\mathbf{p}}$ has trivial fundamental groups (respectively, has trivial first homology). An equivalent formulation of the FHT property is that there is a finite set of closed paths in \mathcal{D} whose homology classes generate Π as a (ZS, ZS)-bimodule. Note that a homotopy trivializer is also a homology trivializer, and so if \mathcal{P} is FDT then it is also FHT; as remarked in the introduction, whether the converse holds is an open question² [42, 66].

The properties FDT and FHT are in fact monoid invariants in the sense that if \mathcal{P} and \mathcal{P}' are two finite rewriting systems presenting isomorphic monoids, if one of them has the property FDT (respectively, FHT) then so does the other [60, 66]; thus we can talk about FDT and FHT monoids. These properties are also closed under retraction [66].

Squier [60] showed that monoids with finite complete rewriting systems are of type FDT (see §1.4.2), and hence also FHT [54, 60]. More recently, Kobayashi [38] has shown that monoids with a finite presentation containing one relation only are of type FDT; it is an open question as to whether such monoids have finite complete rewriting systems.

²Pride and Otto have recently found an example of a monoid which is FHT but not FDT.

1.3.2 Homological finiteness conditions

A monoid S is left- FP_n (respectively, right- FP_n) if Z, regarded as a left (respectively, right) ZS-module with trivial S-action

$$s.n = n$$
 (resp. $n.s = n$) $(s \in S, n \in \mathbb{Z}),$

has a partial resolution

 $A_n \xrightarrow{\partial_n} A_{n-1} \xrightarrow{\partial_{n-1}} \cdots \xrightarrow{\partial_1} A_0 \xrightarrow{\varepsilon} \mathbb{Z} \longrightarrow 0$

by free left (respectively, right) $\mathbb{Z}S$ -modules of finite rank.

The following is a standard result of homological algebra.

Lemma 1.3.1 (generalized Schanuel's Lemma [11, p.193]) Let

$$0 \longrightarrow K \longrightarrow P_n \longrightarrow P_{n-1} \longrightarrow \cdots \xrightarrow{} P_0 \longrightarrow M \longrightarrow 0$$

and

$$0 \longrightarrow K' \longrightarrow P'_n \longrightarrow P'_{n-1} \longrightarrow \cdots P'_0 \longrightarrow M \longrightarrow 0$$

be exact sequences of modules (or bimodules) where P_i and P'_i are projective for each i = 0, 1, ..., n. If n is even then there is an isomorphism

$$P_0 \oplus P_1' \oplus P_2 \oplus P_3' \oplus \cdots \oplus P_n \oplus K' \cong P_0' \oplus P_1 \oplus P_2' \oplus P_3 \oplus \cdots \oplus P_n' \oplus K,$$

otherwise there is an isomorphism

$$P_0 \oplus P_1' \oplus P_2 \oplus P_3' \oplus \cdots \oplus P_n' \oplus K \cong P_0' \oplus P_1 \oplus P_2' \oplus P_3 \oplus \cdots \oplus P_n \oplus K'$$

Consequently, if P_i and P'_i are finitely generated for $i \le n$, then K is finitely generated if and only if K' is finitely generated.

If we splice together the sequence (3) with sequences found in [54] we obtain the following complex

$$0 \longrightarrow \Pi^{l} \longrightarrow \mathbb{Z}S.\mathbf{r} \longrightarrow \mathbb{Z}S.\mathbf{x} \longrightarrow \mathbb{Z}S \longrightarrow \mathbb{Z} \longrightarrow 0, \tag{1.8}$$

giving a partial resolution of \mathbb{Z} up to dimension 2, and so if S is finitely presented, then by the generalized Schanuel Lemma (Lemma 1.3.1 above) S is of type left- FP_3 if and only if Π^l is finitely generated as a left $\mathbb{Z}S$ -module. There is also the corresponding sequence of right $\mathbb{Z}S$ -modules.

For groups the properties left- and right- FP_n are equivalent for all n. However, this is not the case for monoids in general. Cohen has given an example of a monoid which is right- FP_n for all $n\geq 0$ but is not left- FP_1 [13]; also a monoid of type left- FP_n for all $n\geq 0$ introduced by Squier [59] was later shown not to be right- FP_3 by Pride and Wang [56].

Another standard result of homological algebra is the Snake Lemma, which we shall use in Chapter 3.

Lemma 1.3.2 (Snake Lemma) Suppose we have the commutative diagram



where both rows are exact.

Then the map $D: ker\gamma \longrightarrow coker\alpha$ defined by $c \longmapsto \iota^{-1}\beta \rho^{-1}c + im\alpha$ is a homomorphism. Moreover, there is an exact sequence

$$ker \alpha \longrightarrow ker \beta \longrightarrow ker \gamma \xrightarrow{D} coker \alpha \longrightarrow coker \beta \longrightarrow coker \gamma.$$

1.3.3 FDT and FHT monoids are left and right FP_3

Let \mathbf{p} be a set of closed paths in the 2-complex \mathcal{D} , and consider the sequence

$$\mathbb{Z}S.\mathbf{p} \xrightarrow{\partial_3} \mathbb{Z}S.\mathbf{r} \xrightarrow{\partial_2} \mathbb{Z}S.\mathbf{x} \xrightarrow{\partial_1} \mathbb{Z}S \xrightarrow{\partial_0} \mathbb{Z} \longrightarrow 0$$
(1.9)

of left ZS-modules, where the homomorphisms are defined as follows. Firstly, ∂_0 is the *augmentation homomorphism*, mapping each $s \in S$ to 1, and ∂_1 is the homomorphism defined by the map

$$x \mapsto (x-1)$$
 $(x \in \mathbf{x}).$

If for $w = x_1 x_2 \dots x_n \in F$ we use $\langle w \rangle$ to denote the element

$$\overline{x_1\ldots x_{n-1}}.x_n+\overline{x_1\ldots x_{n-2}}.x_{n-1}+\ldots+x_1$$

of $\mathbb{Z}S.\mathbf{x}$, then ∂_2 is the homomorphism defined by the map

$$r \longmapsto \langle r_{-1} \rangle - \langle r_{+1} \rangle \qquad (r \in \mathbf{r}).$$

Lastly, ∂_3 is the homomorphism defined by the map

$$p \longmapsto \sum_{i=1}^{n} \varepsilon_i \overline{u}_i \cdot r_i$$

for $p = (u_1, r_1, \varepsilon_1, v_1) \dots (u_n, r_n, \varepsilon_n, v_n) \in \mathbf{p}$, where, for $i = 1, 2, \dots, n, u_i, v_i \in F, \varepsilon_i = \pm 1$ and $r_i \in \mathbf{r}$.

It is straightforward to see that (1.9) is a complex, and in fact

Theorem 1.3.3 ([17, 54]) If \mathbf{p} is a homology trivializer then the complex (1.9) is exact.

A partial resolution of right $\mathbb{Z}S$ -modules can be constructed in the same way, and therefore FDT and FHT monoids are of type left and right FP_3 .

1.3.4 The properties *FDT* and *FHT* for groups

It has been shown that for finitely presented groups, the properties FDT, FHT and FP_3 are equivalent. This follows from the equivalence of FDT and FP_3 , shown by Cremanns and Otto [18] and Pride [55], together with the following result, for which we give a new proof:

Proof: We use the following general result. Let G be a group and let A be a $(\mathbb{Z}G, \mathbb{Z}G)$ -bimodule with an abelian group decomposition

$$A = \bigoplus_{g \in G} A_g$$

such that $h_1 A_g h_2 = A_{h_1gh_2}$ for all $h_1, g, h_2 \in G$. Then A_1 has a left ZG-module structure with G-action

$$g * a = g.a.g^{-1}$$
 $(g \in G, a \in A_1).$

Lemma 1.3.5 1. $A \otimes_{\mathbb{Z}G} \mathbb{Z}$ and A_1 are isomorphic as left $\mathbb{Z}G$ -modules.

2. If A_1 is finitely generated as a left ZG-module then A is finitely generated as a bimodule.

Proof:

1. Regarding \mathbb{Z} as a left $\mathbb{Z}G$ -module with trivial G-action, the map

$$\psi: A \times \mathbb{Z} \longrightarrow A_1 \qquad (\sum_{g \in G} a_g, n) \longmapsto n \sum_{g \in G} a_g.g^{-1} \qquad (a_g \in A_g, n \in \mathbb{Z})$$

is bilinear, since in particular for any $h \in G$,

$$\psi((\sum_{g\in G} a_g).h, n) = n \sum_{g\in G} a_g.h(gh)^{-1} = n \sum_{g\in G} a_g.g^{-1} = \psi(\sum_{g\in G} a_g, h.n).$$

Therefore ψ induces a homomorphism

$$\Psi: A \otimes_{\mathbb{Z}G} \mathbb{Z} \longrightarrow A_1 \qquad \sum_{g \in G} a_g \otimes 1 \longmapsto \sum_{g \in G} a_g g^{-1}$$

which is the inverse of the homomorphism

$$A_1 \longrightarrow A \otimes_{\mathbb{Z}G} \mathbb{Z} \qquad a \longmapsto a \otimes 1$$

induced by the inclusion of A_1 into A. It is easily seen that these isomorphisms respect the left G-actions.

2. Suppose that the left module A_1 has finite generating set $B = \{b_1, \ldots, b_n\}$. Then for any $a \in A_g$ we can express $a.g^{-1} \in A_1$ as a sum

$$a \cdot g^{-1} = \sum_{i=1}^{k} \varepsilon_i (h_i * b_{j_i}) = \sum_{i=1}^{k} \varepsilon_i h_i \cdot b_{j_i} \cdot h_i^{-1},$$

where, for $1 \le i \le k$, $\varepsilon_i = \pm 1$, $h_i \in G$ and $1 \le j_i \le n$, and so in the bimodule A we can write

$$a = \sum_{i=1}^{k} \varepsilon_i h_i b_{j_i} (h_i^{-1}g).$$

Therefore B generates A as a bimodule.

Theorem 1.3.4 now follows by applying the Lemma to the decomposition (1.5) of Π , taking account of (1.6): the bimodule Π is finitely generated if and only if Π^l is finitely generated as a left-module, and this is the case if and only if S is FP_3 (applying the generalized Schanuel's Lemma (Lemma 1.3.1) to the exact sequence (1.8)).

1.4 Complete rewriting systems

General references for complete rewriting systems are [8, 7, 14]. In §1.1.2, we gave the following definition: a rewriting system \mathcal{P} is complete if it is noetherian and it has precisely one irreducible element in each congruence class. Therefore we can denote by Irr(w) the irreducible word obtained by rewriting any word w, and if z is another word in the same congruence class, then Irr(z) = Irr(w). We now give two equivalent definitions.

For any $w \in F$ we shall say that \mathcal{P} is confluent (respectively, locally confluent) at w if whenever $u, v \in F$ satisfy $w \to_{\mathbf{r}}^* u$ and $w \to_{\mathbf{r}}^* v$ (respectively, $w \to_{\mathbf{r}} u$ and $w \to_{\mathbf{r}} v$) then there is some $z \in F$ such that $u \to_{\mathbf{r}}^* z$ and $v \to_{\mathbf{r}}^* z$. We shall say that \mathcal{P} is confluent (respectively, locally confluent) if it is confluent at w (respectively locally confluent at w) for all $w \in F$.

┛

In geometric terms, \mathcal{P} is confluent (respectively, locally confluent) if whenever there is a pair of positive paths (respectively, positive edges) in Γ from $w \in F$ to vertices u and v, there are also positive paths in Γ from u and v to some common terminal vertex.

Lemma 1.4.1 (Newman [49]) Any congruence class in a confluent rewriting system contains at most one irreducible word.

Proof: For any pair of words u and u' lying in the same congruence class, there is a path in Γ from one to the other. We shall use an inductive argument on the path length to show that there is a pair of positive paths from u and u' to some common vertex.

If the path has length one, then the pair of positive paths consists of either the path itself or its inverse, together with the corresponding empty path at u or u'. If the path has length n > 1, it can be considered as the product of two non-empty paths of length < n: a path from u to some vertex v, say, together with a path from vto u'. By induction, there are positive paths p and q from u and v respectively to some common vertex w, say, and also positive paths q' and p' from v and u' to some common vertex w'. Because \mathcal{P} is confluent, there are positive paths r and r' from w and w' to some common vertex z, say, and then pr and p'r' are a pair of positive paths from u and u' to z.

The lemma is now straightforward to prove: if u and u' are irreducible words in the same congruence class, then there is a pair of positive paths from these words to some common vertex, and because these paths must be empty we see that u and u' are the common vertex, that is, u = u'.

We therefore have the following characterization, due to Newman [49]:

a rewriting system is complete if and only if it is noetherian and confluent.

(1.10)
Lemma 1.4.2 (Principal of noetherian induction) Let P be a predicate on a set A with a noetherian order \succ . Suppose that whenever $a \in A$ has the property that P(a') holds for every $a' \in A$ with $a \succ a'$, then P(a) holds. Then P(a) holds for every $a \in A$.

Proof. For suppose that P(a) does not hold for some $a \in A$. Then by our supposition there will be some $a_1 \in A$ such that $a \succ a_1$ and $P(a_1)$ is false. Continuing this argument will give an infinite sequence

 $a \succ a_1 \succ a_2 \succ a_3 \succ \cdots$,

a contradiction to our assumption that \succ is a noetherian order.

This method of induction will be used in proving Proposition 3.3.4 and Theorem 4.4.1; we illustrate its use here by giving the well-known proof of the so-called Diamond Lemma:

Lemma 1.4.3 (Diamond Lemma [49]) A noetherian and locally confluent rewriting system \mathcal{P} is confluent.

Proof: We use noetherian induction on F, where the order is the reduction order $\succ_{\mathbf{r}}$ induced by the rules.

Let w be an element of F, and inductively assume the rewriting system is confluent at any vertex $w' \in F$ with $w \succ_r w'$. Suppose that $w \rightarrow_r^* u$ and $w \rightarrow_r^* v$ hold for some $u, v \in F$, so there are positive paths p and q in Γ from w to u and v respectively. We want to deduce that \mathcal{P} is also confluent at w, that is, there is some $z \in F$ such that $u \rightarrow_r^* z$ and $v \rightarrow_r^* z$.

Firstly, if w is irreducible, then u = v = w and we can just take z to be w.

Suppose on the other hand that w is not irreducible. If either of the paths is empty, say w = u, then we can again just take z = v, so we can assume that both paths are non-empty and can therefore be written as products p = ep' and q = fq', where e and f are positive edges and p', q' are positive paths. Since \mathcal{P} is locally confluent, there is some $w' \in F$ such that $\tau e \to_{\mathbf{r}}^* w'$ and $\tau f \to_{\mathbf{r}}^* w'$.



Figure 1.1: The Diamond Lemma

Because $\tau e \to_{\mathbf{r}}^* u$ and $\tau e \to_{\mathbf{r}}^* w'$ hold, by inductive assumption there is some $u' \in F$ such that $u \to_{\mathbf{r}}^* u'$ and $w' \to_{\mathbf{r}}^* u'$; similarly, there is some $v' \in F$ such that $v \to_{\mathbf{r}}^* v'$ and $w' \to_{\mathbf{r}}^* v'$. Also, because $w' \to_{\mathbf{r}}^* u'$ and $w' \to_{\mathbf{r}}^* v'$ hold, there is some $z \in F$ such that $u' \to_{\mathbf{r}}^* z$ and $v' \to_{\mathbf{r}}^* z$. We therefore have (see Figure 1.1)

 $u \rightarrow_{\mathbf{r}}^{*} u' \rightarrow_{\mathbf{r}}^{*} z$ and $v \rightarrow_{\mathbf{r}}^{*} v' \rightarrow_{\mathbf{r}}^{*} z$;

thus \mathcal{P} is confluent at w, and the lemma follows by noetherian induction (Lemma 1.4.2).

Thus we have the alternative characterization:

a rewriting system is complete if and only if it is noetherian and locally confluent. (1.11)

The class of monoids which have finite complete rewriting systems has no known alternative characterization, although as we have seen such monoids must be finitely presented and have solvable word problem. We also have the following result:

Theorem 1.4.4 (Anick-Groves-Squier [3, 27, 59]) A monoid with a finite complete rewriting system is of type left and right FP_{∞} .

In fact Squier [59] showed that such monoids were of type left and right FP_3 , and Groves [27] gave the more general result; later it was realized that this result was contained in work already published by Anick [3]. Other proofs have been given by Kobayashi [36], and Brown [12].

Corollary 1.4.5 (Squier [59]) There are finitely presented monoids with solvable word problem which do not have finite complete rewriting systems.

Squier proved this by appealing to known examples of finitely-presented groups with solvable word problems which are not of type FP_3 , for example groups studied by Stallings [61] and Abels [1]. Abels' example is a group of matrices, which we shall consider in §A.1, and Stallings' group is shown to have solvable word problem in [4]. Squier also gave his own examples of finite rewriting systems with solvable word problem for monoids which are not left FP_3 .

Critical pairs

The characterization (1.11) prompts the study of local confluence.

Definition 1.4.6 A pair of positive edges with the same initial vertex form a critical pair if either:

- 1. One of the pair is both left- and right-reduced (a critical pair of inclusion type); or
- 2. One of the pair is left-reduced but not right-reduced, the other is right-reduced but not left-reduced, and they are not disjoint (a critical pair of overlapping type).

Remark 1.4.7 We are trying here to emphasize the geometric interpretation of rewriting systems - the usual definition of a critical pair would only be the terminal vertices of one of our critical pairs (of edges), that is, the two possible results of rewriting the initial word.

Example 1.4.8 Consider the rewriting system (1.1)

$$[\mathbf{x} \cup \mathbf{x}^{-1} ; xx^{-1} = 1, x^{-1}x = 1 \quad (x \in \mathbf{x})]$$

of Example 1.1.4 defining the free group with basis **x**. Each critical pair is of overlapping type and is in the star of a vertex $xx^{-1}x$ or $x^{-1}xx^{-1}$ ($x \in \mathbf{x}$):



Note that any pair of edges in $star^+(w)$ $(w \in F)$ are either disjoint or are a translate of a critical pair by the two-sided action of F. A critical pair as defined above is a pair of edges corresponding to the two different ways of rewriting a word composed of the (non-disjoint) left-hand-sides of two rules.

Definition 1.4.9 A resolution of a critical pair (e, f) is a pair of positive paths from τe and τf to some common vertex; we say that a critical pair is resolvable if it has a resolution.

Example 1.4.8 is particularly simple as its critical pairs are resolved immediately (that is, with empty paths).

Example 1.4.10 The rewriting system given by the rules

 $ac = ca, \quad caaa = aa, \quad bacb = 1$

on the alphabet a, b, c has one critical pair of inclusion type,



٦

and also three of overlapping type, for example



(As mentioned previously (Remark 1.4.7), elsewhere these critical pairs would usually be written as the pairs (1, bcab) and (a^2c, ca^2ca) respectively.)

The following two paths give a resolution of the critical pair of overlapping type, to the vertex ca^2 :



The critical pair of inclusion type cannot be resolved, since the terminal vertices are distinct irreducibles.

Proposition 1.4.11 (Knuth and Bendix [35]) A noetherian rewriting system is complete if and only if all the critical pairs are resolvable.

Proof: A noetherian rewriting system is locally confluent if and only if each critical pair is resolvable.

Hence the result that the rewriting system (1.1) given in Example 1.1.4 is complete, since the critical pairs are resolved.

1.4.1 The Knuth-Bendix completion procedure

Let $\mathcal{P} = [\mathbf{x}; \mathbf{r}]$ be a finite rewriting system where \mathbf{r} is compatible with some total reduction order \succ on F - note that \mathcal{P} is always equivalent to such a rewriting system,

for example if we choose some length-lexicographical ordering of F, then we can replace each rule $r \in \mathbf{r}$ for which $r_{+1} \not\geq_{llex} r_{-1}$ with the rule (r_{-1}, r_{+1}) . If \mathcal{P} is not complete then applying the Knuth-Bendix procedure [35] will produce an equivalent complete rewriting system \mathcal{P}_{∞} , but which in general will have an infinite set of rules.

The procedure is as follows. For each non-resolvable critical pair (e, f) we can choose a pair of positive paths p_e and p_f from τe and τf respectively to different irreducibles. Let \mathbf{r}' denote the set of rules obtained from \mathbf{r} by adding for each such critical pair (e, f) a rule $(\tau p_e, \tau p_f)$ if $\tau p_e \succ \tau p_f$, otherwise adding the rule $(\tau p_f, \tau p_e)$. Since \mathcal{P} is not complete, $\mathbf{r}' \neq \mathbf{r}$, but the rewriting system $[\mathbf{x}; \mathbf{r}']$ is equivalent to \mathcal{P} . (More generally, we can add any finite set of rules

$$\{u_1 = v_1, u_2 = v_2, \ldots\}$$

which allows us to resolve the critical pairs, provided that $u_i \succ v_i$, and of course that $u_i \leftrightarrow_r^* v_i$ for each i = 1, 2, ...)

Put $\mathbf{r} = \mathbf{r}_0$, and then for all $n \ge 1$ we define \mathbf{r}_n to be \mathbf{r}'_{n-1} , and

$$\mathbf{r}_{\infty} = \bigcup_{n>0} \mathbf{r}_n.$$

The rewriting system $\mathcal{P}_{\infty} = [\mathbf{x}; \mathbf{r}_{\infty}]$ is equivalent to \mathcal{P} , and is compatible with the original noetherian order on F. But it is also complete, for any critical pair arises from a pair of rules found in \mathbf{r}_n for some $n \ge 0$, and is therefore resolvable using a rule in $\mathbf{r}'_n = \mathbf{r}_{n+1} \subseteq \mathbf{r}_{\infty}$.

Example 1.4.12 Consider the rewriting system consisting of the set of rules

$$\mathbf{r} = \{ xx^{-1} = 1, x^{-1}x = 1, xx = 1 \}$$

on the alphabet x, x^{-1} , which is compatible with the length-lexicographic ordering based on the ordering $x^{-1} \triangleright x$. There are four critical pairs, all of overlapping type: two arising from the rules $xx^{-1} = 1$ and $x^{-1}x = 1$ which are immediately resolved as in Example 1.4.8, and the critical pairs



Applying the Knuth-Bendix procedure, we obtain the set \mathbf{r}_1 of rules by adding the rule $x^{-1} = x$ to \mathbf{r} . There are two new critical pairs (of inclusion type) which arise from the introduction of this rule, namely



and they are both easily resolved by the path $\underline{xx} \rightarrow 1$.

Thus in this case the Knuth-Bendix procedure terminates with a complete rewriting system after adding a single rule. This rewriting system is a presentation of the group of order two, which in fact has the simpler complete rewriting system

$$\left[\vartheta ; \vartheta \vartheta = 1\right].$$

Example 1.4.13 It has been shown that the rewriting system consisting of the set of rules

$$\mathbf{r} = \mathbf{r}_0 = \{ xa = atx, xt = tx, xy = 1, xb = bx, ab = 1 \}$$

on the alphabet a, b, t, x, y has no equivalent finite complete rewriting system [60, Corollary 6.8]. Therefore in this case the Knuth-Bendix procedure will not terminate in a finite number of steps. The rewriting system is compatible with the recursive path ordering \succ from the left (Definition 1.1.3) induced by the ordering $x \triangleright a \triangleright b \triangleright t \triangleright y$:

- 1. $xa \succ atx$ holds because $x \triangleright a$ and $xa \succ tx$ ($xa \succ tx$ holds because $x \triangleright t$ and $xa \succ x$, and $xa \succ x$ because $a \succ 1$);
- 2. $xt \succ tx$ holds because $x \triangleright t$ and $xt \succ x$ ($xt \succ x$ holds because $t \succ 1$);
- 3. $xb \succ bx$ holds because $x \triangleright b$ and $xb \succ x$ ($xb \succ x$ holds because $b \succ 1$).

For n = 1, 2, 3, ..., let

$$\mathbf{r}_n = \mathbf{r}_{n-1} \cup \{at^n b = 1\}.$$

By [60, Lemma 6.2(b)], the relation $at^n b \leftrightarrow_{\mathbf{r}}^* 1$ holds for each n, and so the rules $\mathbf{r}, \mathbf{r}_1, \mathbf{r}_2, \ldots$ all generate the same Thue congruence. We use \mathcal{P}_n to denote the rewriting system with the rules \mathbf{r}_n on the alphabet a, b, t, x.

For $n = 0, 1, 2, 3, \ldots$, the rules \mathbf{r}_n give rise to the single critical pair



which (for $n \ge 1$) does not arise from the rules \mathbf{r}_{n-1} . In each case the critical pair can be resolved by the path

$$atxt t^{n-1}b \to at^2 \underline{xt} t^{n-2}b \to \cdots \to at^{n+1} \underline{xb} \to \underline{at^{n+1}b} x \to x$$
.

in $\Gamma(\mathcal{P}_{n+1})$.

Thus this sequence of rewriting systems can be produced by applying the Knuth-Bendix procedure to \mathcal{P}_0 . It is only the union

$$\mathcal{P}_{\infty} = [a, b, t, x ; \bigcup_{n \ge 0} \mathbf{r}_n]$$

which is complete.

34

1.4.2 The propertyFDT

Throughout this section we shall assume that for a complete rewriting system \mathcal{P} , **p** is a set of closed paths in Γ obtained by choosing resolutions of each critical pair. The following observation is fundamental to our study of complete rewriting systems.

Remark 1.4.14 The boundary of each 2-cell σ in $\mathcal{D}^{\mathbf{p}}$ has a unique maximal vertex w_{σ} and minimal vertex z_{σ} with respect to the reduction ordering $\succ_{\mathbf{r}}$ on F, and the boundary of σ consists of two positive paths from w_{σ} to z_{σ} . These paths will come from resolutions of either critical or disjoint pairs, according as to whether σ arises from the trivializer \mathbf{p} or not.

Moreover each 2-cell is uniquely determined by the two edges at its maximal vertex, and we can therefore denote each 2-cell σ by $[w_{\sigma}; (e, f)]$ where $e <_{w_{\sigma}} f$ are the corresponding edges in $star^{+}(w_{\sigma})$. We shall use this notation when studying complete rewriting systems in §3.3, §4.4 and §5.2.

Example 1.4.15 Consider the vertex $w = xxx^{-1}x$ in the graph Γ of the complete rewriting system in Example 1.4.12 consisting of the rules \mathbf{r}_1 on the alphabet x, x^{-1} . The set $star^+(w)$ consists of the three edges

 $\underline{xx}x^{-1}x \rightarrow x^{-1}x, \qquad x\underline{xx}^{-1}x \rightarrow xx, \qquad xx\underline{x}^{-1}\underline{x} \rightarrow xx$

which for convenience we label e, f and g respectively. The 2-cell [w; (e, g)] is in the 2-complex \mathcal{D} , and is attached along the closed path



If **p** is chosen to contain the closed paths



then the 2-cells [w; (e, f)] and [w; (f, g)] will be attached along the closed paths



in $\mathcal{D}^{\mathbf{p}}$.

Theorem 1.4.16 (Squier [60]) The set \mathbf{p} is a homotopy trivializer, and therefore a finite complete rewriting system has the property FDT.

We need the following two lemmas.

Lemma 1.4.17 Any pair of positive paths p, q from a single vertex to its irreducible will give rise to a closed path pq^{-1} which is null-homotopic.

Proof: This is proved by noetherian induction on F, where the order is the reduction order $\succ_{\mathbf{r}}$ induced by the rules.

Suppose that $p = e_1 \dots e_m$ and $q = f_1 \dots f_n$ are two positive paths from $w \in F$ to Irr(w), and assume inductively that for any $w' \in F$ satisfying $w \succ_r w'$ and positive paths r, sfrom w' to Irr(w') (= Irr(w)) we have that rs^{-1} is null-homotopic. We want to deduce that the closed path pq^{-1} is also homotopically trivial.

If w is irreducible, then both paths are empty, and the result is obvious. Otherwise, both paths are non-empty. If $e_1 = f_1$ then by inductive assumption the closed path $e_2 \dots e_m f_n^{-1} \dots f_2^{-1}$ is null-homotopic and the result follows. If not, then there is a 2-cell

ا...



Figure 1.2: Construction used in Lemma 1.4.17

 $[w; (e_1, f_1)]$ in $\mathcal{D}^{\mathbf{p}}$ with boundary $e_1 p' q'^{-1} f_1^{-1}$, say, where p' and q' are positive paths. Let us choose some positive path r from $\tau p' = \tau q'$ to Irr(w). If we use \simeq to denote the relation "homotopic with endpoints fixed" then we can write (see Figure 1.2)

$$p \simeq f_1 q' p'^{-1} e_2 \dots e_m$$

$$\simeq f_1 q' p'^{-1} p' r \qquad \text{(by inductive assumption)}$$

$$\simeq f_1 q' r$$

$$\simeq q \qquad \text{(by inductive assumption)}.$$

Thus any pair of positive paths from w to Irr(w) give rise to a null-homologous path, and the lemma now follows by noetherian induction (Lemma 1.4.2).

Lemma 1.4.18 For any path p from u to w $(u, w \in F)$,

 $p \simeq q_u q_w^{-1}$

holds for any pair q_u and q_w of positive paths from u and w respectively to Irr(u) = Irr(w).

Proof: We use induction on the length n of p. If p is empty, then by Lemma 1.4.17 $p \simeq q_u q_w^{-1}$. If n > 0 then we can write $p = e^{\varepsilon}p'$, where $e \in e^+$ and $\varepsilon = \pm 1$, and if we write $v = \tau e^{\varepsilon}$ and choose some positive path q_v from v to the irreducible (z, say), then we have



Figure 1.3: Proof of Lemma 1.4.18

1. $e^{\varepsilon} \simeq q_u q_v^{-1}$, because

(a) if $\varepsilon = +1$ then by Lemma 1.4.17 we have $q_u \simeq eq_v$;

(b) if $\varepsilon = -1$ then again by Lemma 1.4.17 we have $eq_u \simeq q_v$; and

2. (by the inductive hypothesis) $p' \simeq q_v q_w^{-1}$.

Therefore (see Figure 1.3)

$$p = e^{\varepsilon} p' \simeq q_u q_v^{-1} q_v q_w^{-1} \simeq q_u q_w^{-1},$$

and the result follows by induction.

Proof of Theorem 1.4.16 If p_1 and p_2 are paths with the same initial and terminal vertices (u and v, say) then we can choose positive paths q_u and q_v to the irreducible. By Lemma 1.4.18 we have

$$p_1 \simeq q_u q_v^{-1} \simeq p_2.$$

It follows immediately that any closed path in $\mathcal{D}^{\mathbf{p}}$ is homotopically trivial, and so \mathbf{p} is indeed a homotopy trivializer.

Example 1.4.19 Let \mathcal{P} be the complete rewriting system in Example 1.4.13 with the infinite set of rules. Each critical pair is of the form



and can be resolved by the path

$$at\underline{xt} t^{n-1}b \to at^2\underline{xt} t^{n-2}b \to \dots \to at^{n+1}\underline{xb} \to \underline{at^{n+1}b} x \to x$$
.

By Theorem 1.4.16, the set of closed paths of the form



for $n \ge 0$ is a homotopy trivializer of \mathcal{D} .

1.4.3 Groups with finite complete rewriting systems

There are many groups which are known to have finite complete rewriting systems, for example free groups of finite rank (Example 1.4.8), finitely generated abelian groups, the Heisenberg group [33], and a group studied by Greendlinger [51]. Dekov [20] has shown that torus-knot groups have such rewriting systems, as do one-relator groups with a finite presentation of the form

$$\langle \mathbf{x} \cup \{a, b\}; a^{-1}b^{-1}abw \rangle \tag{1.12}$$

where w is a word on the generators $\mathbf{x} \cup \mathbf{x}^{-1}$.

Groves and Smith [28] have investigated how the property of being presented by a finite complete rewriting system behaves under various group-theoretic constructions

(subgroups, quotient groups, group extensions and HNN-extensions) and used the results to prove:

Theorem 1.4.20 ([28]) Constructible solvable groups have finite complete rewriting systems.

The following questions have so far been unanswered:

Open Question 1.4.21 ([28]) If a group G has a finite index subgroup with a finite complete rewriting system, then G itself has a finite complete rewriting system. Is the converse true?

Open Question 1.4.22 ([65]) Suppose that H_1 and H_2 are groups and that the free product $H_1 * H_2$ has a finite complete rewriting system. Is it always true that H_1 and H_2 themselves have finite complete rewriting systems?

Because of the following result, a positive answer to the second question would also give a positive answer to the first:

Theorem 1.4.23 (Pride and Wang [56]) Let H be a subgroup of finite index n in a group G. If G has a finite complete rewriting system, then so does the free product $H * \mathcal{F}_{n-1}$ of H with the free group of rank n-1.

We also mention the following open questions, regarding two much-studied classes of groups which are of type FP_{∞} and have solvable word problems:

Open Question 1.4.24 ([38]) Do all finitely generated one-relator groups have a finite complete rewriting system?

Open Question 1.4.25 ([52]) Do all hyperbolic groups have finite complete rewriting systems? Hyperbolic groups are particular examples of automatic groups, which were introduced by Cannon and Thurston [22]. Do all automatic groups have a finite complete rewriting system?

Chapter 2

λ -complete rewriting systems

Abstract

We introduce λ -complete rewriting systems which give solutions to λ -word problems. In §2.1.1 we characterize monoids which have a finite λ -complete rewriting system as those monoids with a λ -Dehn presentation, and in §2.2 we describe a method of obtaining λ -complete rewriting systems for certain groups which are HNN-extensions whose base groups have complete rewriting systems, and we use this to obtain some interesting examples.

2.1 λ -solvable word problems

Definition 2.1.1 The λ -word problem of a rewriting system will be to decide for any word w whether or not w is congruent to the empty word. We shall say that a rewriting system has a λ -solvable word problem if there is an algorithm which solves its λ -word problem.

The following proposition shows that this property is a monoid invariant, so we can extend the above definition to say that a finitely presented monoid has λ -solvable word problem if each of its finite rewriting systems does.

Proposition 2.1.2 If \mathcal{P} and \mathcal{Q} are finite rewriting systems which define isomorphic monoids then \mathcal{P} has λ -solvable word problem if and only if \mathcal{Q} does.

Proof: By the Tietze Theorem (Theorem 1.1.1), it is enough to show that the result is true whenever Q is obtained from \mathcal{P} by a single Tietze transform.

- 1. Suppose that a type 1 Tietze transformation is applied to $\mathcal{P} = [\mathbf{x}; \mathbf{r}]$ to give the rewriting system $\mathcal{Q} = [\mathbf{x}; \mathbf{r}']$, where \mathbf{r}' is obtained from \mathbf{r} by adjoining a rule u = v with $u \leftrightarrow_{\mathbf{r}}^* v$. Because the congruences $\leftrightarrow_{\mathbf{r}}^*$ and $\leftrightarrow_{\mathbf{r}'}^*$ are the same, any algorithm solving the λ -word problem for \mathcal{P} also solves the λ -word problem for \mathcal{Q} , and vice versa.
- 2. Suppose that a type 2 Tietze transformation is applied to $\mathcal{P} = [\mathbf{x}; \mathbf{r}]$ to give the rewriting system \mathcal{Q} by adding a new letter y and rule y = z for some word z on the alphabet \mathbf{x} . Recall from §1.1.2 that the homomorphism of free monoids on the alphabets of the respective rewriting systems which identifies the letters in \mathbf{x} and maps y to z induces an isomorphism

 $S(\mathcal{Q}) \longrightarrow S(\mathcal{P}).$

A word w on the alphabet $\mathbf{x} \cup \{y\}$ of \mathcal{Q} is therefore congruent to the empty word (with respect to the rules of \mathcal{Q}) if and only if $w' \leftrightarrow_r^* 1$ holds, where w' denotes the word obtained from w by rewriting each instance of the letter y according to the rule y = z.

It follows easily that \mathcal{Q} has λ -solvable word problem if and only if \mathcal{P} does. \Box

Definition 2.1.3 We shall say that a rewriting system is λ -complete if it is noetherian, and the empty word is the unique irreducible in its congruence class.

Proposition 2.1.4 If a monoid has a finite λ -complete rewriting system then it has λ -solvable word problem.

Proof: Suppose that \mathcal{P} is a finite λ -complete rewriting system. Because \mathcal{P} is noetherian, any word w can be rewritten in a finite number of steps to an irreducible word

 w^* , say, and then since the empty word is the unique irreducible in its congruence class, $w \leftrightarrow_r^* 1$ if and only if $w^* = 1$.

Obviously, if a rewriting system has solvable word problem then it has λ -solvable word problem. Example 2.1.5 shows that the converse does not hold in general; it is true, however, if we restrict our attention to the class of groups (Proposition 2.1.6 below).

Example 2.1.5 The following rewriting system was shown by Tsejtin [63] to have unsolvable word problem (the congruence class containing the word *aaa* is a non-recursive subset of the free monoid on the given alphabet).

```
alphabet: a, b, c, d, e

rules: ac=ca, ad=da, de=edb, caaa=aa, cdca=cdcae,

bc=cd, dc=cd, ce=eca, daaa=aa.
```

Because there are no rules whose left or right hand side is the empty word, the empty word is in the congruence class consisting only of itself; an algorithm to solve the λ -word problem is then just to check whether or not a word is the empty word.

Proposition 2.1.6 A group G with λ -solvable word problem has solvable word problem.

Proof: Suppose that G has a rewriting system \mathcal{P} with some algorithm solving the λ -word problem. For each letter x in the alphabet x of \mathcal{P} we can choose a word $w_x \in F$ such that $\overline{xw}_x = 1$: since G is a group we know that such words exist, and we can check using the algorithm whether or not any word $w \in F$ satisfies $\overline{xw} = 1$.

We wish to describe an algorithm which solves the word problem for G. We claim that any pair of words u and $v = x_1 x_2 \dots x_n$ (where $x_i \in \mathbf{x}$ for $i = 1, 2, \dots n$) are congruent with respect to the rules \mathbf{r} if and only if $uw_{x_n} \dots w_{x_1} \leftrightarrow_{\mathbf{r}}^* 1$. We can check this fact, since we have assumed an algorithm recognizing words that are congruent to the empty word. For suppose that u and v are congruent. Then

$$uw_{x_{n}} \dots w_{x_{1}} \quad \leftrightarrow_{\mathbf{r}}^{*} \quad vw_{x_{n}} \dots w_{x_{1}}$$

$$= x_{1} \dots x_{n} w_{x_{n}} \dots w_{x_{1}}$$

$$\leftrightarrow_{\mathbf{r}}^{*} \quad x_{1} \dots x_{n-1} w_{x_{n-1}} \dots w_{x_{1}}$$

$$\vdots$$

$$\leftrightarrow_{\mathbf{r}}^{*} \quad x_{1} w_{x_{1}}$$

$$\leftrightarrow_{\mathbf{r}}^{*} \quad 1.$$

On the other hand, suppose that $uw_{x_n} \dots w_{x_1}$ is congruent to the empty word. Since G is a group, xw_x is congruent to the empty word if and only if $w_x x$ is also, and so

 $v \leftrightarrow_{\mathbf{r}}^{*} uw_{x_{n}} \dots w_{x_{1}}x_{1} \dots x_{n}$ $\leftrightarrow_{\mathbf{r}}^{*} uw_{x_{n}} \dots w_{x_{2}}x_{2} \dots x_{n}$ \vdots $\leftrightarrow_{\mathbf{r}}^{*} uw_{x_{n}}x_{n}$ $\leftrightarrow_{\mathbf{r}}^{*} u.$

This proves our claim.

Thus, as regards the word problem, finite complete rewriting systems for groups seem to give unnecessary information. Indeed, as mentioned in §1.4, there are known examples of finitely presented groups with solvable word problem but without any finite complete rewriting system. The following questions present themselves:

Open Question 2.1.7 ([8, 16]) Does every finitely presented group with solvable word problem have a finite λ -complete rewriting system?

Open Question 2.1.8 ([8, 16]) Is there a group with a finite λ -complete rewriting system but with no finite complete rewriting system?

We could also rephrase Open Questions 1.4.24 and 1.4.25 as follows.

Open Question 2.1.9 Do all finitely generated one-relator groups have a finite λ -complete rewriting system?

Open Question 2.1.10 Do all hyperbolic groups have finite λ -complete rewriting systems? Do all automatic groups have finite λ -complete rewriting systems?

2.1.1 λ -Dehn presentations

In the next section we discuss some examples of groups with finite λ -complete rewriting systems, but first we give an alternative description of such groups.

Definition 2.1.11 ([44]) A group presentation $\langle \mathbf{x}; \mathbf{r} \rangle$ is called a Dehn presentation if it is finite, and if the following holds:

whenever a word w on the alphabet $\mathbf{x} \cup \mathbf{x}^{-1}$ is non-empty and satisfies $\overline{w} = 1$, then either w is not freely reduced, or w contains a subword u for which there is a word v of shorter length such that uv^{-1} is a cyclic conjugate of an element of $\mathbf{r} \cup \mathbf{r}^{-1}$.

For any group presentation $\mathcal{G} = \langle \mathbf{x}; \mathbf{r} \rangle$ we can construct a length-reducing (and therefore noetherian) rewriting system

$$\hat{\mathcal{G}}_{lr} = [\mathbf{x} \cup \mathbf{x}^{-1} ; \hat{\mathbf{r}}]$$
(2.1)

whose rules consist of all free reductions $xx^{-1} = 1, x^{-1}x = 1$ ($x \in \mathbf{x}$) together with all rules of the form u = v where uv^{-1} is a cyclic permutation of some relator or its inverse, and v has strictly shorter length than u. If \mathcal{G} is a Dehn presentation, the empty word is the only irreducible element in its congruence class, and therefore $\hat{\mathcal{G}}_{lr}$ is λ -complete. This rewriting system performs the Dehn algorithm [44] and, of course, any group with a Dehn presentation has a solvable word problem. The fact that groups with Dehn presentations have λ -complete rewriting systems has been noted before (see [8, 45] and the references therein). In [45] some results on the classes of groups which have particular types of length-reducing λ -complete rewriting systems are given. Dehn [19] originally showed that the fundamental groups of closed orientable surfaces of genus $n \ge 2$ have such presentations. More recently, Gromov [26] introduced the notion of word hyperbolic groups, and stated that this class of groups coincides with the class of groups which admit a Dehn presentation. Proofs of this result have been given elsewhere, for example in [58].

A characterization theorem for groups with a finite λ -complete rewriting system.

We shall generalize the idea of a Dehn presentation as follows.

Definition 2.1.12 We shall call a group presentation

$$\mathcal{G} = \langle \mathbf{x} \, ; \, \mathbf{r} \rangle$$

a λ -Dehn presentation if it is finite, and if for some reduction ordering \succ on the free monoid $F(\mathbf{x} \cup \mathbf{x}^{-1})$ the following holds:

- 1. $xx^{-1}, x^{-1}x \succ 1$ for all $x \in \mathbf{x}$; and
- if w = 1, where w∈F(x∪x⁻¹) is non-empty, then either w is not freely reduced, or w contains a subword u for which there is a word v such that u≻v and uv⁻¹ is a cyclic conjugate of an element of r∪r⁻¹.

Thus a Dehn presentation is a λ -Dehn presentation where we can choose the length reducing order (Definition 1.1.1). On the other hand the next example shows that there are groups which have a λ -Dehn presentation for some non-length reducing reduction order but which do not have any Dehn presentation.

Example 2.1.13 Consider the group with presentation

 $\mathcal{G} = \langle a, b, c ; abc = cba \rangle.$

Greendlinger [25] gave this as an example of a presentation that was not a Dehn presentation. In fact it is known that this group has no Dehn presentation; Gromov observed that any abelian subgroup of a word hyperbolic group containing an element of infinite order is finite-by-cyclic ([26] - for a proof of this see [58, Corollary 3.6]), whereas the subgroup generated by \overline{ab} and \overline{cb} is free abelian of rank two. However, it does have a finite complete rewriting system, discovered by Otto [51]:

alphabet:
$$a, a^{-1}, b, b^{-1}, c, c^{-1}$$

rules: $aa^{-1} = 1, a^{-1}a = 1, a^{-1}cb = bca^{-1}, abc = cba,$
 $bb^{-1} = 1, b^{-1}b = 1, ac^{-1} = b^{-1}c^{-1}ab,$
 $cc^{-1} = 1, c^{-1}c = 1, a^{-1}b^{-1} = c^{-1}b^{-1}a^{-1}c.$

Dekov [20] gave another proof that this rewriting system is noetherian by showing that it is compatible with the recursive path ordering from the left \succ induced by the partial ordering on the alphabet given by writing

 $a \triangleright b^{-1} \triangleright c, \qquad a \triangleright c^{-1}, \qquad a^{-1} \triangleright c^{-1} \triangleright b, \qquad a^{-1} \triangleright b^{-1}$

of the alphabet. For example $ac^{-1} \succ b^{-1}c^{-1}ab$ holds because $a \triangleright b^{-1}$ and $ac^{-1} \succ c^{-1}ab$; the latter holds because $a \triangleright c^{-1}$ and $ac^{-1} \succ ab$; this time the latter holds because $c^{-1} \triangleright b$.

It is easily checked that each rule is either a free reduction or can be obtained from a cyclic permutation of abc = cba, and it follows that \mathcal{G} is a λ -Dehn presentation.

Proposition 2.1.14 The class of groups with finite λ -complete rewriting systems coincides with the class of groups which have a λ -Dehn presentation.

Proof: In the same way that a finite (length-reducing) λ -complete rewriting system (2.1) can be constructed from a Dehn presentation, we can construct a finite λ -complete rewriting system from any λ -Dehn presentation.

Conversely, suppose that G is a group with finite λ -complete rewriting system $\mathcal{P} = [\mathbf{x}; \mathbf{r}]$. Since G is a group, for each $x \in \mathbf{x}$ we can choose words w_x on \mathbf{x} such that $\overline{xw_x} = \overline{w_xx} = 1$. Introducing new monoid generators $\mathbf{x}^{-1} = \{x^{-1} : x \in \mathbf{x}\}$ by type 2 Tietze transformations gives the equivalent finite rewriting system

 $\mathcal{P}' = [\mathbf{x} \cup \mathbf{x}^{-1}; \mathbf{r}'],$

where $\mathbf{r'} = \mathbf{r} \cup \{x^{-1} = w_x : x \in \mathbf{x}\}.$

Then \mathcal{P}' is also noetherian, since the new rules only operate on the new generators, replacing them with words on x: any hypothetical infinite sequence

$$w_1 \rightarrow_{\mathbf{r}'} w_2 \rightarrow_{\mathbf{r}'} w_3 \rightarrow_{\mathbf{r}'} \cdots$$

involving words on $\mathbf{x} \cup \mathbf{x}^{-1}$ would involve only a finite number of applications of the new rules, and so we could easily derive a corresponding infinite reduction sequence for \mathcal{P} , a contradiction.

It is also λ -complete, because any word w on $\mathbf{x} \cup \mathbf{x}^{-1}$ can be rewritten in a finite number of steps to a word w' on \mathbf{x} , and then $w \leftrightarrow^*_{\mathbf{r}'} 1$ holds if and only if $w' \leftrightarrow^*_{\mathbf{r}} 1$. We can then rewrite w to the empty word as follows:

$$w \rightarrow^*_{\mathbf{r}'} w' \rightarrow^*_{\mathbf{r}} 1.$$

(We are again using the fact that the homomorphism of free monoids which identifies \mathbf{x} and maps each $x^{-1} \in \mathbf{x}^{-1}$ to the word w_x on \mathbf{x} induces an isomorphism

$$S(\mathcal{P}') \longrightarrow S(\mathcal{P})$$

-see §1.1.2.)

Now write

 $\mathcal{G} = \langle \mathbf{x} ; \hat{\mathbf{r}} \rangle,$

where $\hat{\mathbf{r}} = \{r_{+1}r_{-1}^{-1} : r \in \mathbf{r}\} \cup \{w_x x : x \in \mathbf{x}\}$, and let $\succ_{\mathbf{r}'}$ be the reduction ordering on $F(\mathbf{x} \cup \mathbf{x}^{-1})$ induced by the set of rules \mathbf{r}' of \mathcal{P}' (Definition 1.1.5). If some non-empty word w on $\mathbf{x} \cup \mathbf{x}^{-1}$ is equivalent to the empty word, then, as \mathcal{P}' is λ -complete, w can be rewritten to the empty word, and so:

- 1. $w \succ 1$: in particular, $xx^{-1} \succ_{\mathbf{r}'} 1$ and $x^{-1}x \succ_{\mathbf{r}'} 1$;
- 2. w must contain a subword r_{+1} , the left-hand side of some rule in **r**, or a letter x^{-1} ($x \in \mathbf{x}$), where there is a relator $r_{+1}r_{-1}^{-1}$ or $x^{-1}w_x^{-1}$ in \mathcal{G} with $r_{+1} \succ_{\mathbf{r}'} r_{-1}$ or $x^{-1} \succ_{\mathbf{r}'} w_x$ respectively.

Thus \mathcal{G} is λ -Dehn, and we are done.

_

2.2 Producing λ -complete rewriting systems from complete rewriting systems

In this section we introduce a method of constructing λ -complete rewriting systems for HNN groups whose base groups have certain complete rewriting systems with nice properties. It relies on Britton's Lemma (Lemma 2.2.2), which describes the structure of words that are equivalent to 1 in HNN groups.

We use this method to construct some interesting examples, and remark on the possibility of using similar constructions to search for an answer to Open Question 2.1.8.

2.2.1 HNN-extensions

Let G be a group with group presentation $\mathcal{G} = \langle \mathbf{x}; \mathbf{r} \rangle$, and suppose that A and B are isomorphic subgroups. Let a_1, a_2, \ldots be words on \mathbf{x} representing a generating set of A, and let b_1, b_2, \ldots be words representing the images of this generating set under some isomorphism $\varphi : A \rightarrow B$. If $t \notin \mathbf{x}$ is some new letter, then the group presented by

$$\langle \mathbf{x} \cup \{t\} ; \mathbf{r} \cup \{t^{-1}a_1t = b_1, t^{-1}a_2t = b_2, \ldots\} \rangle$$
 (2.2)

is an HNN-extension of G with stable letter t, associating subgroups A and B (by the isomorphism φ). We call G the base group, and we denote the HNN-extension by $G_{*\varphi:A\cong B}$, or just $G_{*A\cong B}$ if the isomorphism is understood. We shall generally abbreviate the presentation (2.2) to

$$\langle \mathcal{G} | t ; t^{-1}a_1t = b_1, t^{-1}a_2t = b_2, \ldots \rangle.$$

Lemma 2.2.1 ([57, p.412]) The group homomorphism induced by the inclusion of the presentation $\langle \mathbf{x}; \mathbf{r} \rangle$ of the base group into (2.2) is injective.

This group construction has become very important since it was first introduced by Higman, Neumann and Neumann [30] (where the above Lemma was first proved, and then used to give some embedding theorems). In particular, the combinatorial arguments used by both Boone and Novikov to give groups with unsolvable word problems were later seen to be contained in the more general theory of HNN-extensions, and Britton [10] was able to use the following lemma to considerably simplify Boone's construction:

Lemma 2.2.2 (Britton's Lemma [10]) Consider the presentation (2.2). If w is a word involving the stable letter t which is equivalent to 1, then w contains a subword of the form $t^{-1}ut$ or of the form tvt^{-1} , where u and v are words on $\mathbf{x}\cup\mathbf{x}^{-1}$ representing elements of the subgroups A and B respectively.

Let \mathcal{P}_H be a finite complete rewriting system for a group H, with words u, v on the alphabet \mathbf{x} such that

 $\{\ldots, u^{-1}, 1, u, u^2, \ldots\}$ and $\{\ldots, v^{-1}, 1, v, v^2, \ldots\}.$

are sets of irreducible words. Then these sets are sets of normal forms of the infinite cyclic subgroups generated by \overline{u} and \overline{v} are, respectively.

By associating the cyclic subgroups $\langle \overline{u} \rangle$, $\langle \overline{v} \rangle$ under the isomorphism induced by the map $\overline{u} \mapsto \overline{v}$, we can form the HNN-group

$$G = H *_{\langle \overline{u} \rangle \cong \langle \overline{v} \rangle}.$$

Theorem 2.2.3 The rewriting system Q obtained from \mathcal{P}_H by adding new letters t and t^{-1} together with the additional rules

$$tt^{-1} = 1, \quad t^{-1}u = vt^{-1}, \quad t^{-1}u^{-1} = v^{-1}t^{-1}, \\ t^{-1}t = 1, \quad tv = ut, \quad tv^{-1} = u^{-1}t$$

is a λ -complete rewriting system for G.

Proof: We first show that the new rewriting system is noetherian. Let $F(\tilde{F})$ be the free monoid on the alphabet

 $\widetilde{F} = \{\widetilde{w}: \text{ w is a word on the alphabet } \mathbf{x}\}.$

(Note that for two words w_1 and w_2 on \mathbf{x} , $(\widetilde{w}_1)(\widetilde{w}_2)$ is not equal to $\widetilde{w_1w_2}$.)

We can use the reduction order on $F(\mathbf{x})$ induced by the rules of \mathcal{P}_H to give a noetherian partial order \triangleright on the alphabet $\widetilde{F} \cup \{t, t^{-1}\}$ as follows:

- 1. $t^{\epsilon} \triangleright \widetilde{w}$ for $\epsilon = \pm 1$ and each letter $\widetilde{w} \in \widetilde{F}$;
- 2. $\widetilde{w}_1 \triangleright \widetilde{w}_2$ if $w_1 \rightarrow * w_2$ but $w_1 \neq w_2$.

Any word w on the alphabet $\mathbf{x} \cup \{t, t^{-1}\}$ of \mathcal{Q} is of the form

$$w = w_1 t^{\varepsilon_1} w_2 t^{\varepsilon_2} \dots w_n t^{\varepsilon_n} w_{n+1}$$

where $\varepsilon_i = \pm 1$ and w_i is a word on **x** for i = 1, 2, ..., n. Let \widetilde{w} be the corresponding word

$$\widetilde{w} = \widetilde{w}_1 t^{\varepsilon_1} \widetilde{w}_2 t^{\varepsilon_2} \dots \widetilde{w}_n t^{\varepsilon_n} \widetilde{w}_{n+1}$$

on the alphabet $\widetilde{F} \cup \{t, t^{-1}\}$.

For any two words w and z on the alphabet of Q we then write $w \succ z$ if and only if $\widetilde{w} \succ_{\triangleright} \widetilde{z}$, where \succ_{\triangleright} is the recursive path ordering from the left (Definition 1.1.3) on $F(\widetilde{F} \cup \{t, t^{-1}\})$ induced by the partial order \triangleright . It remains to show that the rules of Qare compatible with this reduction ordering on $F(\mathbf{x} \cup \{t, t^{-1}\})$. This is easily checked. Firstly,

 $r_{+1} \succ r_{-1}$ holds for each rule r in \mathcal{P}_H , since $\tilde{r}_{+1} \triangleright \tilde{r}_{-1}$ immediately gives $\tilde{r}_{+1} \succ_{\triangleright} \tilde{r}_{-1}$.

The additional rules are seen to be compatible as follows.

- 1. For $\varepsilon = \pm 1$, $t^{\varepsilon}t^{-\epsilon} \succ 1$ holds because $t^{\varepsilon}t^{-\epsilon} \succ 1$ by definition.
- 2. The relation $t^{-1}u \succ vt^{-1}$ holds because $t^{-1}\widetilde{u} \succ_{\triangleright} \widetilde{v}t^{-1}$, since $t^{-1} \triangleright \widetilde{v}$ and $t^{-1}\widetilde{u} \succ_{\triangleright} t^{-1}$; the relation $t^{-1}\widetilde{u} \succ_{\triangleright} t^{-1}$ holds because $\widetilde{u} \succ_{\triangleright} 1$.

- 3. The relation $t^{-1}u^{-1} \succ vt$ holds because $t^{-1}\widetilde{u} \succ_{\triangleright} \widetilde{v}t^{-1}$, since $t^{-1} \triangleright \widetilde{v}$ and $t^{-1}\widetilde{u} \succ_{\triangleright} t^{-1}$; this latter holds because $\widetilde{u} \succ_{\triangleright} 1$;
- 4. The relation $tv \succ ut$ follows from the fact that $t\tilde{v} \succ_{\triangleright} \tilde{u}t$, because $t \triangleright \tilde{u}$ and $t\tilde{v} \succ_{\triangleright} t$; this latter holds because $\tilde{v} \succ_{\triangleright} 1$.
- 5. The relation $tv^{-1} \succ u^{-1}t$ holds because $\widetilde{tv^{-1}} \succ \widetilde{u^{-1}t}$, because $t \triangleright \widetilde{u^{-1}}$ and $\widetilde{tv^{-1}} \succ t$; this latter holds because $\widetilde{v^{-1}} \succ 1$.

Therefore Q is indeed noetherian, and it remains to show that the empty word is the unique irreducible in its congruence class.

Suppose that w is an irreducible word which is congruent to the empty word. If w contains the letter t or t^{-1} then by Britton's Lemma it must contain a subword of the form $t^{-1}wt$ or $tw't^{-1}$ where w, w' are words on $\mathbf{x} \cup \mathbf{x}^{-1}$ representing elements of $\langle \overline{u} \rangle$, $\langle \overline{v} \rangle$, respectively, and so it will not be irreducible - a contradiction. If on the other hand w does not contain any instances of the letter t or the letter t^{-1} , then by Lemma 2.2.1 we see that it is the empty word, which is therefore the unique irreducible in its congruence class.

2.2.2 Some examples

Baumslag-Solitar groups

The infinite cyclic group has a finite complete rewriting system consisting of the rules

 $xx^{-1} = 1$ and $x^{-1}x = 1$

on the alphabet x, x^{-1} . For each pair of integers p and q, the normal forms of the subgroups generated by \overline{x}^p and \overline{x}^q are, respectively,

$$\{x^{kp}: k \in \mathbb{Z}\}$$
 and $\{x^{kq}: k \in \mathbb{Z}\},\$

and so by Theorem 2.2.3, the HNN-group with the presentation

$$\mathcal{G}_{p,q} = \langle x, t \; ; \; tx^p t^{-1} = x^q \rangle \tag{2.3}$$

obtained by associating these subgroups has the following λ -complete rewriting system:

alphabet:
$$x, x^{-1}, t, t^{-1}$$

rules: $xx^{-1} = 1, \quad t^{-1}x^p = x^q t^{-1}, \\ x^{-1}x = 1, \quad t^{-1}x^{-p} = x^{-q}t^{-1}, \\ tt^{-1} = 1, \quad tx^q = x^p t, \\ t^{-1}t = 1, \quad tx^{-q} = x^{-p}t.$

This rewriting system is not complete, since for example a critical pair of the form

$$\underbrace{\frac{t^{-1}x^{p}x^{-1}}{\overset{\parallel}{t^{-1}x^{p-1}}}_{t^{-1}x^{p-1}} x^{q}t^{-1}x^{-1}}_{t^{-1}x^{p-1}}$$

will not generally be resolved.

In fact $\mathcal{G}_{(p,q)}$ is the standard presentation of the Baumslag-Solitar group $G_{(p,q)}$, for which finite complete rewriting systems have been exhibited elsewhere [22, §7.4].

A small cancellation group

The free group \mathcal{F}_3 with basis $\{a, b, c\}$ has a complete rewriting system of the form (1.1), whose rules consist of all free reductions.

The normal forms of the cyclic subgroups generated by \overline{c} and $\overline{cb^{-1}a^{-1}ba}$ are, respec-tively, the sets

$$\{c^k: k \in \mathbb{Z}\} \quad \text{and} \quad \{(cb^{-1}a^{-1}ba)^k: k \in \mathbb{Z}\},\$$

and so by Theorem 2.2.3, the HNN-group with the presentation

$$\langle a, b, c, d ; a^{-1}b^{-1}abc^{-1}d^{-1}cd \rangle.$$
 (2.4)

obtained from \mathcal{F}_3 by associating these cyclic subgroups has the following λ -complete rewriting system:

alphabet: $a, a^{-1}, b, b^{-1}, c, c^{-1}, d, d^{-1}$ **rules:** $aa^{-1} = 1, a^{-1}a = 1, d^{-1}c = cb^{-1}a^{-1}bad^{-1}, bb^{-1} = 1, b^{-1}b = 1, d^{-1}c^{-1} = a^{-1}b^{-1}abc^{-1}d^{-1}, cc^{-1} = 1, c^{-1}c = 1, dcb^{-1}a^{-1}ba = cd, dd^{-1} = 1, d^{-1}d = 1, da^{-1}b^{-1}abc^{-1} = c^{-1}d.$

But (2.4) is the standard presentation of the fundamental group of a closed orientable surface of genus 2, which was shown by Dehn himself to be (what we now call) a Dehn presentation (see [62, §6.1.4]). Therefore this group admits a different λ -complete rewriting system of the form (2.1) which is length reducing.

A group whose isoperimetric function grows faster than any simple exponential.

For each pair (p,q) of integers, the Baumslag-Solitar group $G_{p,q}$ with presentation (2.3) has a finite complete rewriting system [22, §7.4] on the alphabet x, x^{-1}, t, t^{-1} , moreover where the normal forms of the cyclic subgroups generated by \overline{x} and \overline{t} are, respectively,

 $\{x^k: k \in \mathbb{Z}\}$ and $\{t^k: k \in \mathbb{Z}\}.$

Again we can use Theorem 2.2.3 to construct λ -compete rewriting systems for the HNN-groups formed by associating these subgroups. In particular, the group

$$G_{1,2}*_{\langle \overline{x} \rangle \cong \langle \overline{t} \rangle}$$

has a finite λ -complete rewriting system. This group has presentation

$$\langle x, t, s ; txt^{-1} = x^2, t = sxs^{-1} \rangle$$

from which we can remove the generator t by a type 2 Tietze transformation to get the presentation

$$\langle x, s ; x^{(x^s)} = x^2 \rangle$$

where we use the notation x^t to denote the word txt^{-1} . This presentation was shown by Gersten [24] to have an *isoperimetric function*¹ growing faster than the function E_r for all $r \ge 0$, where E_r is the function $\mathbb{N} \longrightarrow \mathbb{N}$ defined recursively by writing

$$E_r(n) = 2^{E_{r-1}(n)}; \qquad E_0(n) = n$$

2.2.3 Remarks concerning Open Question 2.1.8

The Anick-Groves-Squier Theorem (Theorem 1.4.4) tells us that if a group has a finite complete rewriting system then it is of type FP_{∞} . Therefore, in order to answer Open Question 2.1.8, one might attempt to find a finite λ -complete rewriting system for a non- FP_{∞} group.

Proposition 2.13 (b) of Bieri's book [5] states that if a group G is of type FP_n and has isomorphic subgroups of type FP_{n-1} , then an HNN extension $G*_{A\cong B}$ obtained by associating these subgroups is itself of type FP_n . As a consequence, the construction used in Theorem 2.2.3 will only produce λ -complete rewriting systems for groups which are of type FP_{∞} : the base group is always of type FP_{∞} , since it is required to have a finite complete rewriting system, as are the cyclic subgroups which are associated (see, for example, [5, Proposition 2.7]).

However, we do not need the associated subgroups to be cyclic, or indeed to have a finite complete rewriting system for the base group; the requirements of Theorem 2.2.3 just allow a general result to be given. We might hope to use the same method, that is, using Britton's Lemma to examine the irreducible words congruent to the empty word, in specific cases of non- FP_{∞} groups. The author has tried (but not succeeded) to produce a λ -complete rewriting system for Abels' group in this way (see the appendix, §A.1).

¹Isoperimetric functions give some idea of the complexity of word problems (see [26]): for λ complete rewriting systems this function describes, in terms of the length of any word w satisfying $\overline{w} = 1$, the minimum number of applications of rules that one needs to allow in order to be certain
that w can be rewritten to the empty word.

Chapter 3

A 3-dimensional complex and its homology

Abstract

We begin by attaching 3-cells to the 2-complex $\mathcal{D}^{\mathbf{p}}$ to form a 3-complex, which we denote $\overline{\mathcal{D}^{\mathbf{p}}}$. We show that the second homology group of this new 3-complex has, like the first homology of \mathcal{D} , a natural ($\mathbb{Z}S,\mathbb{Z}S$)-bimodule structure. In §3.3 we exhibit this bimodule as the kernel of a presentation of Π (the short exact sequence (4)), and in §3.4 we give the short exact sequence (5).

3.1 A 3-dimensional complex

Let \mathcal{D} be the Squier complex and let \mathbf{p} be a set of closed paths. Recall from §1.3.1 that $\mathcal{D}^{\mathbf{p}}$ is the 2-complex obtained by attaching 2-cells [w, p, w'] $(p \in \mathbf{p}, w, w' \in F)$ along all closed paths of the form

 $\partial[w, p, w'] = w.p.w'.$

We now extend $\mathcal{D}^{\mathbf{p}}$ to form the 3-complex $\overline{\mathcal{D}^{\mathbf{p}}}$: the 2-skeleton is just $\mathcal{D}^{\mathbf{p}}$, and the construction is completed by adding 3-cells as follows. For each positive edge f and each 2-cell σ , where $\partial \sigma = e_1^{\varepsilon_1} \dots e_n^{\varepsilon_n}$ ($e_i \in \mathbf{e}^+$, $\varepsilon_i = \pm 1, i = 1, \dots, n$), 3-cells $[f, \sigma]$ and $[\sigma, f]$ are attached to the 2-skeleton by mapping their boundaries to, respectively:



Figure 3.1: Boundary of the 3-cell $[f, \sigma]$



Figure 3.2: A 2-cell of \mathcal{D}

- 1. the 2-cells $\iota f.\sigma$ and $-\tau f.\sigma$, together with 2-cells $\varepsilon_i[f, e_i]$ for $1 \le i \le n$ according to the diagram Figure 3.1, where $\varepsilon_i[f, e_i]$ is the 2-cell shown in Figure 3.2.
- 2. 2-cells $\sigma.\iota f$ and $-\sigma.\tau f$ together with 2-cells $-\varepsilon_i[e_i, f]$ for $1 \le i \le n$.

Remark 3.1.1 Note the similarities to the construction of the 2-complex \mathcal{D} , where for each pair of positive edges e and f, 2-cells [e, f] are attached to Γ by mapping their boundaries to the closed paths composed of the edges $e.\iota f$ and $-e.\tau f$ together with the edges $\tau e.f$ and $-\iota e.f$.

Remark 3.1.2 For $e, f, g \in e^+$ the added 3-cells [e, [f, g]] and [[e, f], g] are cubes, as shown in Figure 3.3.



Figure 3.3: A cube

The 2-sided action of F again extends naturally to the 3-cells. For $[f, \sigma]$ and $[\sigma, f]$ $(f \in e^+, \sigma \text{ a 2-cell})$, and $u, v \in F$,

 $u.[f,\sigma].v = [u.f,\sigma.v] \quad \text{ and } \quad u.[\sigma,f].v = [u.\sigma,f.v] \;.$

3.2 The 2-dimensional homology of the 3-complex $\overline{\mathcal{D}^{p}}$

We now expand the chain complex $C(\mathcal{D})$ in §1.2.2 to get the chain complex

$$\mathbf{C}(\overline{\mathcal{D}^{\mathbf{p}}}): \qquad 0 \longrightarrow C_3^{\mathbf{p}} \xrightarrow{\partial_3} C_2 \oplus C_2^{\mathbf{p}} \xrightarrow{\tilde{\partial}_2} C_1 \xrightarrow{\partial_1} C_0 \longrightarrow 0$$

associated to $\overline{\mathcal{D}^{\mathbf{p}}}$. Here $C_3^{\mathbf{p}}$ is the free abelian group with basis the set of all 3-cells and $C_2^{\mathbf{p}}$ is the free abelian group with basis the set of all 2-cells [u, p, v] $(p \in \mathbf{p}, u, v \in F)$. The map $\tilde{\partial}_2$ restricted to C_2 is ∂_2 , and a 2-cell [u, p, v] (where $u, v \in F$ and $p = f_1^{\delta_1} f_2^{\delta_2} \dots f_m^{\delta_m}$. $(\delta_i = \pm 1, f_i \in \mathbf{e}^+, i = 1, \dots, m)$ is a path in \mathbf{p}) is mapped to

$$\sum_{i=1}^m \delta_i u.f_i.v \in C_1 .$$

We define ∂_3 as follows. If $f \in \mathbf{e}^+$ and σ is any 2-cell with

$$\tilde{\partial}_2 \sigma = \sum_{i=1}^n \varepsilon_i e_i$$
 $(\varepsilon_i = \pm 1, e_i \in \mathbf{e}^+, i = 1, ..., n)$

59

then

$$\partial_3[f,\sigma] = (\iota f - \tau f).\sigma + \sum_{i=1}^n \varepsilon_i[f,e_i]$$
(3.1)

and

$$\partial_3[\sigma, f] = \sigma.(\iota f - \tau f) - \sum_{i=1}^n \varepsilon_i[e_i, f].$$
(3.2)

Again these chain groups are (free) $(\mathbb{Z}F,\mathbb{Z}F)$ -bimodules, where the *F*-action is inherited from the two-sided action of *F* on $\overline{\mathcal{D}^{\mathbf{p}}}$, and the boundary maps are $(\mathbb{Z}F,\mathbb{Z}F)$ bimodule homomorphisms. For future reference we note in particular that $C_2^{\mathbf{p}}$ is free with basis

$$\hat{\mathbf{p}} = \{ \hat{p} = [1, p, 1] : p \in \mathbf{p} \}.$$
(3.3)

We now examine the second homology group

$$H_2(\overline{\mathcal{D}^{\mathbf{p}}}) = \frac{Z_2(\overline{\mathcal{D}^{\mathbf{p}}})}{B_2(\overline{\mathcal{D}^{\mathbf{p}}})}$$

where $Z_2(\overline{\mathcal{D}^p}) = \operatorname{Ker}\tilde{\partial}_2$ and $B_2(\overline{\mathcal{D}^p}) = \operatorname{Im}\partial_3$. Again $H_2(\overline{\mathcal{D}^p})$ is a $(\mathbb{Z}F, \mathbb{Z}F)$ -bimodule, the *F*-action induced by the two-sided action of *F* on the bases of the chain groups. Recall that *J* denotes the kernel of the ring homomorphism $\mathbb{Z}F \longrightarrow \mathbb{Z}S$.

Lemma 3.2.1 $J.H_2(\overline{\mathcal{D}^p}) = 0$ and $H_2(\overline{\mathcal{D}^p}).J = 0$.

Proof: We will only show that $J.H_2(\overline{\mathcal{D}^p}) = 0$; the other equality is obtained similarly. Suppose that ξ is a 2-cycle, say

$$\xi = \sum_{i=1}^{n} \varepsilon_i \sigma_i \in Z_2(\overline{\mathcal{D}^p}) \qquad (\varepsilon_i = \pm 1 \text{ and } \sigma_i \text{ a 2-cell}, i = 1, \dots, n),$$

where for $i = 1, \ldots, n$

$$\tilde{\partial}_2 \sigma_i = \sum_{j=1}^{k_i} \delta_{i_j} e_{i_j} \qquad (\delta_{i_j} = \pm 1, e_{i_j} \in \mathbf{e}^+, j = 1, \dots, k_i).$$



Figure 3.4: Homologous 2-chains

Then

$$0 = \tilde{\partial}_2 \xi = \sum_{i=1}^n \varepsilon_i \sum_{j=1}^{k_i} \delta_{i_j} e_{i_j}.$$
(3.4)

By (3.1), if $f \in e^+$ then $(\iota f - \tau f)\sigma_i$ is homologous to the 2-chain

$$-\sum_{j=1}^{k_i}\delta_{i_j}[f,e_{i_j}].$$

(see Figure 3.4). and it follows that

$$(\iota f - \tau f) \cdot \xi + B_2(\overline{\mathcal{D}^{\mathbf{p}}}) = -\sum_{i=1}^n \varepsilon_i \sum_{j=1}^{k_i} \delta_{i_j}[f, e_{i_j}] + B_2(\overline{\mathcal{D}^{\mathbf{p}}}).$$

But by comparison with (3.4),

$$\sum_{i=1}^{n} \varepsilon_i \sum_{j=1}^{k_i} \delta_{i_j}[f, e_{i_j}] = 0,$$

and so $(\iota f - \tau f).\xi$ is null-homologous. By Lemma 1.1.6, J is generated as an abelian group by elements of the form $\iota f - \tau f$ $(f \in e^+)$, and it follows immediately that $J.H_2(\overline{\mathcal{D}P}) = 0.$

It follows that there is an induced $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodule structure on the second homology with the action

$$\bar{u}(\xi + B_2(\overline{\mathcal{D}^{\mathbf{p}}}))\bar{v} = u.\xi.v + B_2(\overline{\mathcal{D}^{\mathbf{p}}}) \qquad (\xi \in Z_2(\overline{\mathcal{D}^{\mathbf{p}}}), u, v \in F).$$

We denote this bimodule by $\Pi_2 = \Pi_2(\mathcal{P}, \mathbf{p})$.

Clearly
$$B_1(\mathcal{D}) \subseteq B_1(\overline{\mathcal{D}^p}) \subseteq Z_1(\mathcal{D}) = Z_1(\overline{\mathcal{D}^p})$$
. Note that the quotient

$$\Lambda(\overline{\mathcal{D}^{\mathbf{p}}}) = \frac{B_1(\overline{\mathcal{D}^{\mathbf{p}}})}{B_1(\mathcal{D})}$$

has a $(\mathbb{Z}F,\mathbb{Z}F)$ -bimodule structure, and in fact as a submodule of $H_1(\mathcal{D})$ we can deduce from (1.4) that

$$J.\Lambda(\overline{\mathcal{D}^{\mathbf{p}}}) = \Lambda(\overline{\mathcal{D}^{\mathbf{p}}}).J = 0,$$

so that $\Lambda(\overline{\mathcal{D}^{\mathbf{p}}})$ also has an induced $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodule structure. This module is generated by the elements

 $\tilde{\partial}_2[1, p, 1] + B_1(\mathcal{D}) \ (p \in \mathbf{p}),$

and so we have an epimorphism

 $\nu: \mathbb{Z}S.\mathbf{p}.\mathbb{Z}S \longrightarrow \Lambda(\overline{\mathcal{D}^{\mathbf{p}}}) \qquad p \longmapsto \tilde{\partial}_2[1, p, 1] + B_1(\mathcal{D}).$

Clearly $\Lambda(\overline{\mathcal{D}^{\mathbf{p}}}) = \Pi$ if and only if $H_1(\overline{\mathcal{D}^{\mathbf{p}}}) = 0$, that is, if and only if \mathbf{p} is a homology trivializer.

3.3 The basic short exact sequence

Theorem 3.3.1 If the homology classes of the paths in \mathbf{p} give rise to a set of bimodule generators of Π then there is a short exact sequence

$$0 \longrightarrow \Pi_2 \stackrel{\Phi}{\longrightarrow} \mathbb{Z}S.\mathbf{p}.\mathbb{Z}S \stackrel{\nu}{\longrightarrow} \Pi \longrightarrow 0. \tag{3.5}$$

of $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodules.

The proof we give here was inspired by the methods used by Kobayashi and Otto [40] to give the exact sequence (1) in the introduction.

Let $\mathbb{Z}S.\mathbf{p}.\mathbb{Z}S$ be the free $(\mathbb{Z}S,\mathbb{Z}S)$ -bimodule with basis \mathbf{p} , which we can consider as a $(\mathbb{Z}F,\mathbb{Z}F)$ -bimodule via the homomorphism of F onto S. Then since $C_2^{\mathbf{p}}$ is the free $(\mathbb{Z}F,\mathbb{Z}F)$ -bimodule with basis the set $\hat{\mathbf{p}}$ of 2-cells attached along paths in \mathbf{p} (3.3), we can define a $(\mathbb{Z}F,\mathbb{Z}F)$ -homomorphism

$$\varphi: C_2 \oplus C_2^{\mathbf{p}} \longrightarrow \mathbb{Z}S.\mathbf{p}.\mathbb{Z}S \tag{3.6}$$

which maps C_2 to 0 and a 2-cell $u.\hat{p}.v = [u, p, v]$ $(u, v \in F$ and $p \in \mathbf{p})$ is mapped to $\bar{u}.p.\bar{v}$. Let $K^{\mathbf{p}}$ denote the $(\mathbb{Z}F, \mathbb{Z}F)$ -bimodule Ker φ .

Lemma 3.3.2 We can write $K^{\mathbf{p}} = C_2 + J.\hat{\mathbf{p}}.\mathbb{Z}F + \mathbb{Z}F.\hat{\mathbf{p}}.J.$

Proof: This follows immediately from Lemma 1.2.4.

It is clear from the definition of the boundary map ∂_3 that $B_2(\overline{\mathcal{D}^p}) \subseteq K^p$ (see (3.1) and (3.2)). Also, the restriction of $\tilde{\partial}_2$ to K^p sends K^p onto $B_1(\mathcal{D})$ (if we apply $\tilde{\partial}_2$ to (3.1) where $\sigma = [1, p, w]$ ($w \in F, p \in \mathbf{p}$) the left-hand-side is mapped to 0, giving

$$(\iota f - \tau f).\tilde{\partial}_2 \hat{p}.w = -\tilde{\partial}_2 \sum_{i=1}^n \varepsilon_i [f, e_i] \in B_1(\mathcal{D})$$

so that $\tilde{\partial}_2(J.\hat{\mathbf{p}}.\mathbb{Z}F) \subseteq B_1(\mathcal{D})$; similarly, $\tilde{\partial}_2(\mathbb{Z}F.\hat{\mathbf{p}}.J) \subseteq B_1(\mathcal{D})$ and, moreover, since $\tilde{\partial}_2\partial_3 = 0, B_2(\overline{\mathcal{D}^p})$ is in the kernel of this homomorphism. Therefore we have a complex

$$0 \longrightarrow B_2(\overline{\mathcal{D}^{\mathbf{p}}}) \xrightarrow{incl.} K^{\mathbf{p}} \xrightarrow{\tilde{\partial}_2} B_1(\mathcal{D}) \longrightarrow 0.$$
(3.7)

We shall show that if **p** is a homology trivializer then $B_2(\overline{\mathcal{D}^p})$ is precisely the kernel of the epimorphism $\tilde{\partial}_2 : K^p \longrightarrow B_1(\mathcal{D})$, so that (3.7) is a short exact sequence. We first prove the following lemma.

Lemma 3.3.3 Let \mathbf{q} be another homology trivializer of \mathcal{D} . If the sequence

$$0 \longrightarrow B_2(\overline{\mathcal{D}^{\mathbf{q}}}) \xrightarrow{incl} K^{\mathbf{q}} \xrightarrow{\hat{\partial}_2} B_1(\mathcal{D}) \longrightarrow 0$$

is exact, then so is (3.7).

٦
Proof: Let ξ be a 2-cycle in $K^{\mathbf{p}}$. By Lemma 3.3.2 we can write ξ as a sum

$$\xi = \xi' + \sum_{i=1}^{m} \varepsilon_i (\iota e_i - \tau e_i) . [1, p_i, w_i] + \sum_{i=m+1}^{m+n} \varepsilon_i [w_i, p_i, 1] . (\iota e_i - \tau e_i)$$

where ξ' is a 2-chain in $C_2(\mathcal{D})$ and, for i = 1, ..., m+n, $\varepsilon_i = \pm 1$, $w_i \in F$, e_i is a positive edge in \mathcal{D} and $p_i \in \mathbf{p}$.

Suppose that, for some $i \in \{1, ..., m + n\}$, $\partial p_i = f_1^{\delta_1} \dots f_k^{\delta_k}$, where for j = 1, ..., k $\delta_j = \pm 1$ and f_j is a positive edge in \mathcal{D} . If $i \leq m$ then because of the 3-cell $[e_i, [1, p_i, w_i]]$ we have

$$(\iota e_i - \tau e_i).[1, p_i, w_i] = -\sum_{j=1}^k \delta_j [e_i, f_j]. w_i \qquad (\text{modulo } B_2(\overline{\mathcal{D}^p})),$$

and if m < i then because of the 3-cell $[[w_i, p_i, 1], e_i]$ we have

$$[w_i, p_i, 1].(\iota e_i - \tau e_i) = \sum_{j=1}^k \delta_j w_i.[f_j, e_i] \qquad (\text{modulo } B_2(\overline{\mathcal{D}^p})).$$

Thus ξ is homologous to some 2-cycle ζ , say, in $C_2(\mathcal{D})$, and since $C_2(\mathcal{D}) \subseteq K^q$ by hypothesis we can write ζ as the boundary of some 3-chain ω in $C_3(\overline{\mathcal{D}^q})$.

But because **p** is also a homology trivializer, for each $q \in \mathbf{q}$ we can choose a 2-chain

$$\sum_{i=1}^{t} \alpha_i \varsigma_i \qquad (\alpha_i = \pm 1 \text{ and } \varsigma_i \text{ a 2-cell, for } i = 1, \dots, t)$$

in $C_2(\overline{\mathcal{D}^p})$ with boundary the 1-cycle arising from the closed path q, and if in the 3-chain ω we replace each 3-cell of the form [f, [u, q, v]] or [[u, q, v], f] $(u, v \in F, f$ a positive edge) with the 3-chain

$$\sum_{i=1}^t lpha_i [f, [u, \varsigma_i, v]] \quad ext{ or } \quad \sum_{i=1}^t lpha_i [[u, \varsigma_i, v], f]$$

respectively, then we obtain a 3-chain in $C_3(\overline{\mathcal{D}^p})$ with the same boundary ζ . Therefore $\xi \in B_2(\overline{\mathcal{D}^p})$, and the sequence (3.7) is exact.



Figure 3.5: Typical 3-cells in $\overline{\mathcal{D}^{\mathbf{p}}}$

Proposition 3.3.4 If \mathbf{p} is a homology trivializer, then the sequence (3.7) is exact.

Special case: We assume that $\mathcal{P} = [\mathbf{x}; \mathbf{r}]$ is a *complete* rewriting system such that \mathbf{r} is compatible with some length-lexicographical ordering $<_{llex}$ of F, and \mathbf{p} is a trivializer obtained by choosing resolutions of the critical pairs of \mathbf{r} (see Remark 1.4.14).

Suppose ξ is a 2-cycle in $K^{\mathbf{p}}$. We will show that ξ is null-homologous by an inductive argument, for which we first need to order the 2-cells in $\overline{\mathcal{D}^{\mathbf{p}}}$. In what follows we shall use the notation for 2-cells introduced in Remark 1.4.14. We give the 2-cells of $\mathcal{D}^{\mathbf{p}}$ the noetherian total order described by writing [w; (e, f)] < [w'; (e', f')] if:

- (i) $w <_{llex} w'$; or
- (ii) w = w' and $f <_w f'$; or
- (iii) w = w', f = f' and $e <_w e'$.

Remark 3.3.5 Note that each 3-cell in $\overline{\mathcal{D}^{\mathbf{p}}}$ is of one of the two types shown in Figure 3.5, where $w \in F$ and $e, f, g \in \mathbf{e}^+$ with $e, f \in star^+(w)$. The boundary of each 3-cell

will therefore consist of three 2-cells with maximal vertex $w.\iota g$ or $\iota g.w$ respectively, together with 2-cells with lesser maximal vertices under the length-lexicographic ordering of F.

We can express ξ uniquely as a sum

$$\xi = \sum_{i=1}^{m} n_i \sigma_i$$

where $m \ge 0, \sigma_1, \ldots, \sigma_m$ are distinct 2-cells, and n_1, \ldots, n_m are non-zero integers.

Suppose that ξ is non-zero, that is m > 0, and that the 2-cell σ_1 is maximal, so that $\sigma_i < \sigma_1$ for i = 2, ..., m. Then according to the following procedure we can always replace $n_1\sigma_1$ with a 2-chain composed of lesser 2-cells to obtain a homologous 2-cycle ξ' , say. Because $\xi - \xi' \in B_2(\overline{D^p}) \subseteq K^p$, we have $\xi' = \xi - (\xi - \xi') \in K^p$ also. If we inductively assume that any such 2-cycle (whose maximal 2-cell with non-zero coefficient is beneath σ_1 in our ordering) is null-homologous, then we immediately deduce that ξ itself is null-homologous. The special case of the proposition follows by noetherian induction (Lemma 1.4.2) on the set of all 2-cycles in K^p , ordered by their maximal 2-cell with non-zero coefficient.

(I) Suppose first that we can write $\sigma_1 = \iota e.\sigma'_1$ where $e \in e^+$ and $\sigma'_1 = [w; (f,g)]$ for some $w \in F$ and $f, g \in star^+(w)$. Then

$$\partial_3[e,\sigma_1'] = \sigma_1 + [\iota e.w; (e.w, \iota e.f)] - [\iota e.w; (e.w, \iota e.g)] + \zeta$$

where ζ is a 2-chain composed of 2-cells with maximal vertices which are beneath $\iota e.w$ in the length-lexicographical order on F, and so σ_1 is homologous to the 2-chain

$$-[\iota e.w; (e.w, \iota e.f)] + [\iota e.w; (e.w, \iota e.g)] - \zeta$$

composed of lesser 2-cells under the ordering described above. Therefore up to homology we can remove $n_1\sigma_1$ from ξ , replacing it with a 2-chain ξ' , say, composed of lesser 2-cells.

(II) If we cannot write σ_1 as in (I), and if $\sigma_1 = [w; (e, f)]$ for some $w \in F$, then there is no $g \in star^+(w)$ satisfying $g <_w e$ and such that g is disjoint from e. Suppose instead that there is an edge $g \in star^+(w)$ satisfying $f <_w g$ with f and g disjoint. In this case σ_1 is homologous to the 2-chain

$$[w;(e,g)]-[w;(f,g)]+\zeta$$

where ζ is a 2-chain composed of 2-cells with lesser maximal vertices. Since we now have $[w; (e, g)], [w; (f, g)] > \sigma_1$ we do not want to simply replace $n_1\sigma_1$ as in (I). However, we know that $\tilde{\partial}_2 \xi = 0$, and since σ_1 is maximal it follows that the elementary 1-chains e and -f arising from σ_1 must be cancelled in the boundary map by edges from 2-cells with the same maximal vertex. In fact, part of ξ must consist of a 2-chain

$$\xi' = n_1 \sigma_1 + \sum_{i=1}^k \varepsilon_i [w; (e_i, f_i)]$$
 $(\varepsilon_i = \pm 1, e_i, f_i \in star^+(w), i = 1, ..., k)$

of 2-cells with maximal vertex w, with

$$n_1(e-f) + \sum_{i=1}^k \varepsilon_i(e_i - f_i) = 0.$$
(3.8)

Again since σ_1 is maximal, we cannot have $f <_w f_i$ for any $i \in \{1, ..., k\}$, and so ξ' is homologous to a 2-chain

$$n_1([w; (e,g)] - [w; (f,g)]) + \sum_{i=1}^k \varepsilon_i([w; (e_i,g)] - [w; (f_i,g)]) + \zeta^k$$

where ζ' is a 2-chain consisting of 2-cells whose maximal vertices are beneath w in the length-lexicographic order. But because of (3.8), we can cancel the terms in the first part of this 2-chain to be left with ζ' , and thus we may again (up to homology) replace part of ξ including $n_1\sigma_1$ with a 2-chain composed of lesser 2-cells.

(III) Next, suppose $\sigma_1 = [w; (e, g)]$ and there is an edge $f \in star^+(w)$ disjoint from both e and g and satisfying $e <_w f <_w g$ (so e and g must themselves be disjoint, and σ_1 is a 2-cell from \mathcal{D}), but that there is no such edge satisfying $f <_w e$ or $g <_w f$. In this case we have

$$[w; (e, f)] < \sigma_1 < [w; (f, g)].$$

As in part (II), because all the edges in $star^+(w)$ have to cancel under the boundary map, part of ξ is a 2-chain

$$\xi' = n_1 \sigma_1 + \sum_{i=1}^k \varepsilon_i[w; (e_i, g)] \qquad (\varepsilon_i = \pm 1, \text{ and } e_i \in star^+(w), i = 1, \dots, k)$$

consisting of all 2-cells with maximal vertex w and with their boundary including g, and where $e_i <_w e$ for all i = 1, ..., k since σ_1 is maximal. We then have

$$n_1 + \sum_{i=1}^k \varepsilon_i = 0,$$

as this sum expresses the coefficient of the edge g in $\tilde{\partial}_2 \xi$ (more precisely, in $\tilde{\partial}_2(-\xi)$). Then ξ' is homologous to the 2-chain

$$n_{1}([w; (e, f)] + [w; (f, g)]) + \sum_{i=1}^{k} \varepsilon_{i}([w; (e_{i}, f)] + [w; (f, g)]) + \zeta$$
$$= (n_{1} + \sum_{i=1}^{k} \varepsilon_{i})[w; (f, g)] + n_{1}[w; (e, f)] + \sum_{i=1}^{k} \varepsilon_{i}[w; (e_{i}, f)] + \zeta$$
$$= n_{1}[w; (e, f)] + \sum_{i=1}^{k} \varepsilon_{i}[w; (e_{i}, f)] + \zeta,$$

where ζ is a 2-chain consisting of 2-cells with maximal vertices beneath w in the length-lexicographical order. Therefore we can again (up to homology) replace a part of ξ including $n_1\sigma_1$ with a 2-chain composed of lesser 2-cells only.

(IV) Next suppose that $\sigma_1 = [w; (f, g)]$ where $f, g \in star^+(w)$ are again disjoint, but that now there are no other edges in $star^+(w)$ which are disjoint from both fand g. The coefficient of g in $\tilde{\partial}_2 \xi$ must be 0, and therefore 2-cells of the form [w; (e, g)] must also be represented in ξ , where $e <_w f$ and e and f arise from some critical pair. Then σ_1 is homologous to a 2-chain

$$[w; (e, g)] - [w; (e, f)] + \zeta$$

where ζ is a 2-chain consisting of 2-cells with maximal vertices beneath w in the length-lexicographical order. Therefore we can again (up to homology) replace $n_1\sigma_1$ in ξ with a 2-chain composed of lesser 2-cells only.

(V) The remaining possibility is that the maximal 2-cell σ_1 can be written in the form u.[w; (e, f)].v where $w, u, v \in F$ with u and v irreducible, and (e, f) is a critical pair. Since $\varphi \xi = 0$ it follows that $\xi - n_1 \sigma_1$ must include a 2-chain of the form

$$\sum_{j=1}^{k} \varepsilon_{j} u_{j} [w; (e, f)] . v_{j} \qquad (\varepsilon_{j} = \pm 1 \text{ and } u_{j}, v_{j} \in F \text{ for } j = 1, \dots, k)$$

where $\bar{u}_j = \bar{u}$ and $\bar{v}_j = \bar{v}$ for all j = 1, ..., k, and such that

$$\sum_{j=1}^{k} \varepsilon_j \bar{u}_j \cdot [w; (e, f)] \cdot \bar{v}_j = -n_1 \bar{u} \cdot [w; (e, f)] \cdot \bar{v}$$

But for each j = 1, ..., k we must have either $u_j \neq u$ or $v_j \neq v$, and because \mathcal{P} is complete either $u <_{llex} u_j$ and $v \leq_{llex} v_j$, or $u \leq_{llex} u_j$ and $v <_{llex} v_j$, from which it follows that $\sigma_1 < u_j \cdot [w; (e, f)] \cdot v_j$, a contradiction.

Remark 3.3.6 The above proof does not require the set \mathbf{r} to be finite, since we may in any case assume that the set $star^+(w)$ of any $w \in F$ is well-ordered. This is because w has at most a finite number of subwords, and if there is a pair of positive edges (u, r, +1, v), (u, r', +1, v) (where $u, v \in F$ and $r, r' \in \mathbf{r}$) in $star^+(w)$ which rewrite the same subword then

$$(u, r, +1, v) <_w (u, r', +1, v)$$
 if $r_{-1} <_{llex} r'_{-1}$,

so that the ordering of such edges in $star^+(w)$ is well-founded. It follows that the 2-cells of $\overline{\mathcal{D}^{\mathbf{p}}}$ are well-ordered even if the set of rules **r** and the homology trivializer **p** are both infinite.

General case: Let $\mathcal{P} = [\mathbf{x}; \mathbf{r}]$ be a finite rewriting system, with homology trivializer **p**. We can assume that **r** is compatible with some length-lexicographical order on F and we can complete **r** using the Knuth-Bendix completion procedure (§1.4.1), obtaining a complete (but possibly infinite) rewriting system $\mathcal{P}^{\infty} = [\mathbf{x}; \mathbf{r}^{\infty}]$ with $\mathbf{r} \subseteq \mathbf{r}^{\infty}$ and where \mathbf{r}^{∞} is also compatible with the order on F. Then the 2-complex $\mathcal{D}(\mathcal{P}^{\infty})$ has trivializer \mathbf{p}^{∞} obtained by choosing resolutions of all the critical pairs of \mathbf{r}^{∞} , and by the Special Case and Remark 3.3.6 the sequence

$$0 \longrightarrow B_2(\overline{\mathcal{D}(\mathcal{P}^{\infty})^{\mathbf{p}^{\infty}}}) \xrightarrow{incl.} K^{\infty} \xrightarrow{\tilde{\partial}_2} B_1(\mathcal{D}(\mathcal{P}^{\infty})) \longrightarrow 0$$
(3.9)

is exact, where $K^{\infty} = C_2(\mathcal{D}(\mathcal{P}^{\infty})) + J.\hat{\mathbf{p}}^{\infty}.\mathbb{Z}F + \mathbb{Z}F.\hat{\mathbf{p}}^{\infty}.J$. We will deduce from this and Lemma 3.3.3 that the corresponding sequence for $\overline{\mathcal{D}(\mathcal{P})^{\mathbf{p}}}$ is also exact.

There is a natural inclusion of $\mathcal{D}(\mathcal{P})$ into $\mathcal{D}(\mathcal{P}^{\infty})$, with a retraction ρ of $\mathcal{D}(\mathcal{P}^{\infty})$ onto $\mathcal{D}(\mathcal{P})$ given by some choice of a path in $\mathcal{D}(\mathcal{P})$ from r_{+1} to r_{-1} for each rule $r \in (\mathbf{r}^{\infty} - \mathbf{r})$:

if $\rho(u, r, +1, v) = e_1^{\varepsilon_1} \dots e_m^{\varepsilon_m}$ and $\rho(u', r', +1, v') = f_1^{\delta_1} \dots f_n^{\delta_n}$ for $u, v, u', v' \in F$ and $r, r' \in \mathbf{r}^{\infty}$ $(e_i, f_j$ positive edges in \mathcal{D} and $\varepsilon_i, \delta_j = \pm 1$ for $i = 1, \dots, m$ and $j = 1, \dots, n$) then the 2-cell [(u, r, +1, v), (u', r', +1, v')] is mapped by ρ to the subcomplex composed of the 2-cells

 $\{[e_i,f_j]:1{\leq}i{\leq}m,1{\leq}j{\leq}n\}.$



Figure 3.6: The subcomplex in Example 3.3.7

Example 3.3.7 As as illustration, if say m = 3 and n = 2, with $\varepsilon_1 = \varepsilon_3 = \delta_1 = 1$ and $\varepsilon_2 = \delta_2 = -1$, then the subcomplex will be as shown in Figure 3.6, where as in §3.1 the sign indicates the orientation of the 2-cell.

Let **q** denote the set of closed paths $\{\rho(p) : p \in \mathbf{p}^{\infty}\}$. The retraction ρ can be extended to a map of $\overline{\mathcal{D}(\mathcal{P}^{\infty})^{\mathbf{p}^{\infty}}}$ onto $\overline{\mathcal{D}(\mathcal{P})^{\mathbf{q}}}$. A 2-cell [u, p, v] $(u, v \in F, p \in \mathbf{p})$ will be mapped to the 2-cell $[u, \rho(p), v]$, and for a positive edge f in $\mathcal{D}(\mathcal{P}^{\infty})$ mapped to $\rho f = e_1^{\varepsilon_1} \dots e_m^{\varepsilon_m}$ ($\varepsilon_i = \pm 1$ and e_i a positive edge in $\mathcal{D}(\mathcal{P})$ for $i = 1, \dots, m$) the 3cells [f, [u, p, v]] and [[u, p, v], f] will be mapped to the subcomplexes composed of the 3-cells

$$\{[e_i, [u, \rho(p), v]] : 1 \le i \le m\} \text{ and } \{[[u, \rho(p), v], e_i] : 1 \le i \le m\}$$

respectively. Similarly, those 3-cells whose boundaries consist only of 2-cells in $\mathcal{D}(\mathcal{P}^{\infty})$ are mapped to subcomplexes composed of 3-cells whose boundaries consist only of 2-cells in $\mathcal{D}(\mathcal{P})$. We do not need to describe the map ρ completely as we are mainly concerned with the induced chain map

 $\hat{\rho}: \mathbf{C}(\overline{\mathcal{D}(\mathcal{P}^{\infty})^{\mathbf{p}^{\infty}}}) {\longrightarrow} \mathbf{C}(\overline{\mathcal{D}(\mathcal{P})^{\mathbf{q}}})$

which is as follows (where for brevity the chain groups and boundary maps of

 $C(\overline{\mathcal{D}(\mathcal{P}^{\infty})^{\mathbf{p}^{\infty}}})$ are distinguished by the superscript $-^{\infty}$):

$$\mathbf{C}(\overline{\mathcal{D}(\mathcal{P}^{\infty})^{\mathbf{p}^{\infty}}}): 0 \longrightarrow (C_{3}^{\infty})^{\mathbf{p}^{\infty}} \xrightarrow{\partial_{3}^{\infty}} C_{2}^{\infty} \oplus (C_{2}^{\infty})^{\mathbf{p}^{\infty}} \xrightarrow{\tilde{\partial}_{2}^{\infty}} C_{1}^{\infty} \xrightarrow{\partial_{1}^{\infty}} C_{0} \longrightarrow 0$$

$$\downarrow^{\hat{\rho}} \qquad \qquad \downarrow^{\hat{\rho}_{3}} \qquad \qquad \downarrow^{\hat{\rho}_{2}} \qquad \qquad \downarrow^{\hat{\rho}_{1}} \qquad \qquad \parallel$$

$$\mathbf{C}(\mathcal{D}(\mathcal{P}), \mathbf{q}): 0 \longrightarrow C_{3}^{\mathbf{q}} \xrightarrow{\partial_{3}} C_{2} \oplus C_{2}^{\mathbf{q}} \xrightarrow{\tilde{\partial}_{2}} C_{1} \xrightarrow{\partial_{1}} C_{0} \longrightarrow 0$$

For a positive edge e in $\mathcal{D}(\mathcal{P}^{\infty})$ with $\rho e = e_1^{\varepsilon_1} \dots e_m^{\varepsilon_m}$ ($\varepsilon_i = \pm 1$ and e_i a positive edge in $\mathcal{D}(\mathcal{P})$ for $i = 1, \dots, m$)

$$\hat{\rho}_1 e = \sum_{i=1}^m \varepsilon_i e_i,$$

and if f is another positive edge in $\mathcal{D}(\mathcal{P}^{\infty})$ with $\rho f = f_1^{\delta_1} \dots f_n^{\delta_m}$ ($\delta_j = \pm 1$ and f_j a positive edge in $\mathcal{D}(\mathcal{P})$ for $j = 1, \dots n$) then

$$\hat{\rho}_2[e,f] = \sum_{i=1}^m \sum_{j=1}^n \varepsilon_i \delta_j[e_i, f_j].$$

To complete the description of $\hat{\rho}_2$, a free abelian group generator [u, p, v] of $(C_2^{\infty})^{\mathbf{p}^{\infty}}$ $(u, v \in F, p \in \mathbf{p}^{\infty})$ is mapped to $[u, \rho(p), v]$. Next, in $\overline{\mathcal{D}(\mathcal{P}^{\infty})^{\mathbf{p}^{\infty}}}$ let e be a positive edge and σ a 2-cell with

$$\hat{\rho}_1 e = \sum_{i=1}^m \varepsilon_i e_i$$
 and $\hat{\rho}_2 \sigma = \sum_{j=1}^n \delta_j \sigma_j$

where (for i = 1, ..., m and j = 1, ..., n) $\varepsilon_i, \delta_j = \pm 1$, and e_i is a positive edge and σ_j a 2-cell in $\overline{\mathcal{D}(\mathcal{P})^q}$. Then $\hat{\rho}_3$ is defined by putting

$$\rho_3[e,\sigma] = \sum_{i=1}^m \sum_{j=1}^n \varepsilon_i \delta_j[e_i,\sigma_j] \quad \text{and} \quad \rho_3[\sigma,e] = \sum_{i=1}^m \sum_{j=1}^n \varepsilon_i \delta_j[\sigma_j,e_i].$$

Since the retraction ρ respects the two-sided action of F on the chain groups, there is an induced $(\mathbb{Z}S,\mathbb{Z}S)$ -bimodule epimorphism $\Pi(\mathcal{P}^{\infty}) \longrightarrow \Pi(\mathcal{P})$. Therefore **q** is a homology trivializer of $\mathcal{D}(\mathcal{P})$, and we will use the exactness of (3.9) to show that

$$0 \longrightarrow B_2(\overline{\mathcal{D}(\mathcal{P})^{\mathbf{q}}}) \xrightarrow{incl.} K^{\mathbf{q}} \xrightarrow{\tilde{\partial}_2} B_1(\mathcal{D}(\mathcal{P})) \longrightarrow 0$$
(3.10)

is also exact.

Suppose that ξ is a 2-cycle in $K^{\mathbf{q}}$, written as

$$\xi = \xi' + \sum_{i=1}^{m} \varepsilon_i (\iota e_i - \tau e_i) \cdot [1, q_i, w_i] + \sum_{i=m+1}^{m+n} \varepsilon_i [w_i, q_i, 1] \cdot (\iota e_i - \tau e_i)$$

where $\xi' \in C_2(\mathcal{D}(\mathcal{P}))$ and (for i = 1, ..., m + n) $\varepsilon_i = \pm 1$, e_i is a positive edge in $\mathcal{D}(\mathcal{P})$, $w_i \in F$ and $q_i \in \mathbf{q}$.

Now for each $q \in \mathbf{q}$ choose some $\tilde{q} \in \mathbf{p}^{\infty}$ such that $\rho \tilde{q} = q$. For each edge f of \tilde{q} , $\rho(f)f^{-1}$ is a closed path in $\mathcal{D}(\mathcal{P}^{\infty})$ and so gives rise to the 1-cycle $\hat{\rho}_1(f) - f$. Hence, since \mathbf{p}^{∞} is a homology trivializer, there is a 2-chain $c_2(q)$ in $C_2(\overline{\mathcal{D}(\mathcal{P}^{\infty})^{\mathbf{p}^{\infty}}})$ with boundary

$$\tilde{\partial}_2^{\infty} c_2(q) = \sum_{f \in \tilde{q}} (\hat{\rho}_1 f - f).$$

Note that $\hat{\rho}_2 c_2(q)$ is a 2-cycle, because

$$ilde{\partial}_2 \hat{
ho}_2 c_2(q) = \hat{
ho}_1 ilde{\partial}_2^\infty c_2(q) = \sum_{f \in ilde{q}} (\hat{
ho}_1 f - \hat{
ho}_1 f) = 0.$$

Consider the 2-chain

$$\zeta = \xi' + \sum_{i=1}^{m} \varepsilon_i (\iota e_i - \tau e_i) . ([1, \tilde{q}_i, 1] + c_2(q_i)) . w_i$$
$$+ \sum_{i=m+1}^{m+n} \varepsilon_i w_i . ([1, \tilde{q}_i, 1] + c_2(q_i)) . (\iota e_i - \tau e_i)$$

in K^{∞} . Since for any $q \in \mathbf{q}$

$$\tilde{\partial}_{2}^{\infty}([1,\tilde{q},1]+c_{2}(q)) = \sum_{f \in \tilde{q}} f + \sum_{f \in \tilde{q}} (\hat{\rho}_{1}f - f) = \sum_{f \in \tilde{q}} (\hat{\rho}_{1}f) = \tilde{\partial}_{2}[1,q,1]$$

we have

$$\tilde{\partial}_2^{\infty}\zeta = \tilde{\partial}_2\xi = 0$$

so that $\zeta \in K^{\infty}$ is a 2-cycle.

Moreover,

$$\hat{\rho}_{2}\zeta = \xi + \sum_{i=1}^{m} \varepsilon_{i} (\iota e_{i} - \tau e_{i}) \cdot \hat{\rho}_{2} c_{2}(q_{i}) \cdot w_{i} + \sum_{i=m+1}^{m+n} \varepsilon_{i} w_{i} \cdot \hat{\rho}_{2} c_{2}(q_{i}) \cdot (\iota e_{i} - \tau e_{i})$$
$$= \xi \quad \text{modulo} \ (B_{2}(\overline{\mathcal{D}(\mathcal{P})^{\mathbf{q}}})),$$

because, as in the proof of Lemma 3.2.1, if z is any 2-cycle and e a positive edge, then both $(\iota e - \tau e).z$ and $z.(\iota e - \tau e)$ are null-homologous.

As the sequence (3.9) is exact we know that ζ is the boundary of some 3-chain ω , say, in $C_3^{\infty} = C_3(\overline{\mathcal{D}(\mathcal{P}^{\infty})^{\mathbf{p}^{\infty}}})$. But then the 3-chain $\hat{\rho}_3 \omega$ in $C_3 = C_3(\overline{\mathcal{D}^q})$ has boundary

$$\partial_3 \hat{\rho}_3 \omega = \hat{\rho}_2 \partial_3^\infty \omega = \hat{\rho}_2 \zeta$$

which is homologous to ξ , and therefore ξ is a 2-boundary and the sequence (3.10) is exact also. The proposition now follows because of Lemma 3.3.3.

Proof of Theorem 3.3.1: If $H_1(\mathcal{D}^p) = 0$ then we have the following commutative diagram:

where all vertical maps are inclusions and both rows are exact. From the Snake Lemma (Lemma 1.3.2) we immediately have the short exact sequence (3.5), with

$$\Phi: \Pi_2 = \operatorname{coker}(B_2 \to Z_2) \longrightarrow \operatorname{coker}(K^{\mathbf{p}} \to C_2 \oplus C_2^{\mathbf{p}}) = \mathbb{Z}S.\mathbf{p}.\mathbb{Z}S$$

 $z_2 + B_2 \longrightarrow \varphi(z_2) \quad (z_2 \in Z_2(\overline{\mathcal{D}^p})).$

where φ is the homomorphism (3.6), and

$$\nu: \mathbb{Z}S.\mathbf{p}.\mathbb{Z}S = \operatorname{coker}(K^{\mathbf{p}} \to C_2 \oplus C_2^{\mathbf{p}}) \longrightarrow \operatorname{coker}(B_1 \to Z_1) = \Pi$$

is the natural homomorphism taking each $p \in \mathbf{p}$ to the homology class of the corresponding 1-cycle.

3.4 Killing the left or right action

Let Π_2^l (respectively Π_2^r) denote the left $\mathbb{Z}S$ -module $\Pi_2 \otimes_{\mathbb{Z}S} \mathbb{Z}$ (respectively, the right $\mathbb{Z}S$ -module $\mathbb{Z} \otimes_{\mathbb{Z}S} \Pi_2$).

Theorem 3.4.1 If the homology classes of the paths in \mathbf{p} give rise to a set of bimodule generators of Π then there is a short exact sequence

 $0 \longrightarrow \Pi_2^l \longrightarrow \mathbb{Z}S.\mathbf{p} \longrightarrow \Pi^l \longrightarrow 0 \tag{3.11}$

of left $\mathbb{Z}S$ -modules.

Remark 3.4.2 A similar sequence of right $\mathbb{Z}S$ -modules involving Π_2^r can also be obtained.

We first prove several lemmas, and we need the following definition.

Definition 3.4.3 (Kobayashi and Otto [40]) The leftmost path $p_l(w)$ from w to the minimal element \hat{w} in the congruence class of w with respect to some chosen length-lexicographical ordering of F is defined as follows. First for each pair (\hat{w}, x) $(w \in F, x \in \mathbf{x})$ we choose an arbitrary path $p_l(\hat{w}x)$ from $\hat{w}x$ to $\hat{w}x$. The leftmost paths are then defined recursively as follows:

- (i) if $w = \hat{w}$ then $p_l(w)$ is the empty path at w; or
- (ii) if w≠ŵ and w = ux for some u∈F and x∈x then p_l(w) is defined to be (p_l(u).x)p_l(ûx).

Lemma 3.4.4 (Kobayashi and Otto [40]) The leftmost paths have the property that for all $w_1, w_2 \in F$,

$$p_l(w_1w_2) = (p_l(w_1).w_2)p_l(\widehat{w}_1w_2). \tag{3.12}$$

Proof: This identity is proved using induction on the length n of w_2 . Firstly, if n = 0 then (3.12) reduces to $p_l(w_1) = p_l(w_1)$. Suppose on the other hand that $n \ge 1$ and that by inductive hypothesis (3.12) holds whenever the length of the second word is less than n. We can write $w_2 = w'_2 x$ where $x \in \mathbf{x}$ and $w'_2 \in F$. Then

$$p_{l}(w_{1}w_{2}) = (p_{l}(w_{1}w'_{2}).x)p_{l}(\widehat{w_{1}w'_{2}}x)$$

= $((p_{l}(w_{1}).w'_{2})(p_{l}(\widehat{w_{1}}w'_{2})).x)p_{l}(\widehat{w_{1}w'_{2}}x)$
= $(p_{l}(w_{1}).w_{2})p_{l}(\widehat{w_{1}}w_{2}).$

The lemma follows by induction.

Example 3.4.5 Consider the following complete rewriting system for the free abelian group of rank 2.

alphabet:
$$a, a^{-1}, b, b^{-1}$$

rules: $aa^{-1} = 1$, $b^{-1}b = 1$, $ba^{-1} = a^{-1}b$,
 $a^{-1}a = 1$, $ba = ab$, $b^{-1}a = ab^{-1}$,
 $bb^{-1} = 1$, $b^2a^2 = a^2b^2$, $b^{-1}a^{-1} = a^{-1}b^{-1}$.

This rewriting system is compatible with the length-lexicographic ordering induced by the order $b^{-1} \triangleright b \triangleright a^{-1} \triangleright a$, and the normal forms consist of all words of the form $w = a^m b^n$ where m and n are integers. For each such word there are unique paths $p_l(wa^{\varepsilon})$ and $p_l(wb^{\varepsilon})$ ($\varepsilon = \pm 1$) to the corresponding normal form - these paths do not use the rule $b^2a^2 = a^2b^2$.

The leftmost path $p_l(b^2a^2)$ from the vertex b^2a^2 to the normal form a^2b^2 is the path

 $b\underline{ba}a \rightarrow \underline{ba}ba \rightarrow ab\underline{ba} \rightarrow ab\underline{ba} \rightarrow a^2b^2$

(Since $p_l(b^2a)$ is the path

 $b\underline{b}\underline{a} \rightarrow \underline{b}\underline{a}\underline{b} \rightarrow ab^2$).

This example shows that the leftmost path is not generally the same as the *path* composed of left principal edges (see § 1.2.1), which in this case is the single edge

$$\underline{b^2a}^2 \rightarrow a^2b^2$$

L

Lemma 3.4.6 The left $\mathbb{Z}F$ -homomorphism

$$\hookrightarrow \otimes 1: Z_2(\overline{\mathcal{D}^{\mathbf{p}}}) \otimes_{\mathbb{Z}F} \mathbb{Z} \longrightarrow (C_2 \oplus C_2^{\mathbf{p}}) \otimes_{\mathbb{Z}F} \mathbb{Z}$$

is injective, where **p** is a homology trivializer of \mathcal{D} and \hookrightarrow denotes the inclusion

$$Z_2(\overline{\mathcal{D}^p}) \longrightarrow (C_2 \oplus C_2^p).$$

Proof: For each edge e there is a 1-cycle $z_1(e)$ corresponding to the closed path $p_l(\iota e)^{-1}ep_l(\tau e)$, and for any $w \in F$ we have

$$z_1(e.w) = z_1(e).w .$$

(By Lemma 3.4.4, $p_l(\iota e.w) = (p_l(\iota e).w)p_l(\hat{\iota}ew)$ and $p_l(\tau e.w) = (p_l(\tau e).w)p_l(\hat{\tau}ew)$, so that in writing down the 1-cycle $z_1(e.w)$ we can cancel the edges corresponding to the paths $p_l(\hat{\iota}ew) = p_l(\hat{\tau}ew)$.)

Next, since **p** is a homology trivializer, for each right-reduced positive edge f = (u, r, +1, 1) ($u \in F, r \in \mathbf{r}$) we can choose a 2-chain $c_2(f)$ with boundary $\tilde{\partial}_2 c_2(f) = z_1(f)$, and, for $\varepsilon = \pm 1, v \in F$, we define $c_2(f^{\varepsilon}.v)$ to be $\varepsilon c_2(f).v$. Then to each 2-cell σ of $\overline{\mathcal{D}^{\mathbf{p}}}$ with boundary

$$\tilde{\partial}_2 \sigma = \sum_{i=1}^n \delta_i e_i$$
 $(e_i \in \mathbf{e}^+ \text{ and } \delta_i = \pm 1 \text{ for } i = 1, ..., n)$

we can associate a 2-cycle

$$z_2(\sigma) = \sigma - \sum_{i=1}^n \delta_i c_2(e_i)$$

with the property that $z_2(\sigma.w) = z_2(\sigma).w$ for all $w \in F$. The homomorphism of abelian groups $\kappa : C_2 \oplus C_2^{\mathbf{p}} \longrightarrow Z_2(\overline{\mathcal{D}^{\mathbf{p}}})$ defined by mapping σ to $z_2(\sigma)$ is therefore a homomorphism of right $\mathbb{Z}F$ -modules.

Suppose that

$$\xi = \sum_{i=1}^{n} \varepsilon_i \sigma_i \in Z_2(\overline{\mathcal{D}^{\mathbf{p}}})$$

where (for i = 1, ..., n) $\varepsilon_i = \pm 1$ and σ_i is a 2-cell with

$$\tilde{\partial}_2 \sigma_i = \sum_{j=1}^{k_i} \delta_{i_j} e_{i_j}$$
 $(e_{i_j} \in \mathbf{e}^+ \text{ and } \delta_{i_j} = \pm 1 \text{ for } j = 1, \dots, k_i).$

Then since

$$\sum_{i=1}^{n} \sum_{j=1}^{k_i} \varepsilon_i \delta_{i_j} e_{i_j} = \tilde{\partial}_2 \xi = 0$$

we have

$$(\kappa)(\hookrightarrow)\xi = \sum_{i=1}^{n} \varepsilon_i \sigma_i - \sum_{i=1}^{n} \sum_{j=1}^{k_i} \varepsilon_i \delta_{i_j} c_2(e_{i_j}) = \sum_{i=1}^{n} \varepsilon_i \sigma_i = \xi$$

and so κ is a retraction, with $\kappa \hookrightarrow$ the identity map of $Z_2(\overline{\mathcal{D}^p})$. Because κ is a right $\mathbb{Z}F$ -homomorphism, we have a group homomorphism

$$\kappa \otimes 1 : (C_2 \oplus C_2^{\mathbf{p}}) \otimes_{\mathbb{Z}F} \mathbb{Z} \longrightarrow Z_2(\overline{\mathcal{D}^{\mathbf{p}}}) \otimes_{\mathbb{Z}F} \mathbb{Z}.$$

But then $(\kappa \otimes 1)((\hookrightarrow) \otimes 1) = (\kappa(\hookrightarrow) \otimes 1)$ is the identity map of $Z_2(\overline{\mathcal{D}^p}) \otimes_{\mathbb{Z}F} \mathbb{Z}$ and therefore $(\hookrightarrow) \otimes 1$ is injective.

Therefore, we have the following commutative diagram with exact rows:

Let L denote the kernel of the left $\mathbb{Z}F$ -homomorphism

 $\hookrightarrow \otimes 1: B_1(\mathcal{D}) \otimes_{\mathbb{Z}F} \mathbb{Z} \longrightarrow Z_1(\mathcal{D}) \otimes_{\mathbb{Z}F} \mathbb{Z}$

where \hookrightarrow denotes the inclusion $B_1(\mathcal{D}) \longrightarrow Z_1(\mathcal{D})$. By the Snake Lemma there is a homomorphism

$$d: L \longrightarrow \Pi_2^l (\cong \operatorname{coker}(B_2(\overline{\mathcal{D}^p}) \otimes_{\mathbb{Z}F} \mathbb{Z} \longrightarrow Z_2(\overline{\mathcal{D}^p}) \otimes_{\mathbb{Z}F} \mathbb{Z}))$$

where for $b_1 \in B_1(\mathcal{D})$

 $d(b_1 \otimes 1) = (z_2 + B_2(\overline{\mathcal{D}^p})) \otimes 1$

where $z_2 \in Z_2(\overline{\mathcal{D}^{\mathbf{p}}})$ is such that $z_2 \otimes 1$ has the same image in $(C_2 \oplus C_2^{\mathbf{p}}) \otimes_{\mathbb{Z}F} \mathbb{Z}$ as some lift of $b_1 \otimes 1$ in $K^{\mathbf{p}} \otimes_{\mathbb{Z}F} \mathbb{Z}$, and there is an exact sequence

$$L \xrightarrow{d} \Pi_2^l(\overline{\mathcal{D}^p}) \longrightarrow \mathbb{Z}S.\mathbf{p} \longrightarrow \Pi^l(\mathcal{D}) \longrightarrow 0.$$
(3.13)

Lemma 3.4.7 L is generated as a left $\mathbb{Z}F$ -module by elements of the form

$$\xi.(\iota f - \tau f) \otimes 1$$

where ξ is a 1-cycle and f is a positive edge.

Proof: First note that $\xi (\iota f - \tau f)$ is a 1-boundary, for if

$$\xi = \sum_{i=1}^{n} \varepsilon_i e_i$$
 $(e_i \in \mathbf{e}^+ \text{ and } \varepsilon_i = \pm 1 \text{ for } i = 1, \dots n)$

then

$$\partial_2 \sum_{i=1}^n \varepsilon_i[e_i, f] = \sum_{i=1}^n \varepsilon_i(e_i . (\iota f - \tau f) - (\iota e_i - \tau e_i) . f)$$

= $\xi . (\iota f - \tau f) - \partial_1 \xi . f$
= $\xi . (\iota f - \tau f) .$

Also $\xi (\iota f - \tau f) \otimes 1 \in L$, since in $Z_1(\mathcal{D}) \otimes_{\mathbb{Z}F} \mathbb{Z}$ we can write

$$\xi.(\iota f - \tau f) \otimes 1 = \xi.\iota f \otimes 1 - \xi.\tau f \otimes 1$$
$$= \xi \otimes 1 - \xi \otimes 1$$
$$= 0.$$

Now suppose that some 1-boundary

$$\zeta = \partial_2 \sum_{i=1}^n \varepsilon_i [e_i, f_i] \in B_1(\mathcal{D}) \qquad (e_i, f_i \in \mathbf{e}^+ \text{ and } \varepsilon_i = \pm 1 \text{ for } i = 1, \dots, n)$$

is such that $0 = \zeta \otimes 1$ holds in $Z_1(\mathcal{D}) \otimes_{\mathbb{Z}F} \mathbb{Z}$. In $C_1(\mathcal{D}) \otimes_{\mathbb{Z}F} \mathbb{Z}$ we may write

$$0 = \zeta \otimes 1$$

= $\sum_{i=1}^{n} \varepsilon_i (e_i . (\iota f_i - \tau f_i) - (\iota e_i - \tau e_i) . f_i) \otimes 1$
= $\sum_{i=1}^{n} \varepsilon_i (\iota e_i - \tau e_i) . f'_i \otimes 1$

where if (for i = 1, ..., n) f_i is the edge $(u_i, r_i, +1, v_i)$ $(u_i, v_i \in F, r_i \in \mathbf{r})$ then we use f'_i to denote the right-reduced edge $(u_i, r_i, +1, 1)$. Now as a $(\mathbb{Z}F, \mathbb{Z}F)$ -bimodule, $C_1(\mathcal{D})$ is freely generated by edges of the form (1, r, +1, 1) $(r \in \mathbf{r})$, and so $C_1(\mathcal{D}) \otimes_{\mathbb{Z}F} \mathbb{Z}$ is naturally isomorphic to the free left $\mathbb{Z}F$ -module $\mathbb{Z}F.\mathbf{r}$, the isomorphism mapping $\zeta \otimes 1$ to

$$\sum_{i=1}^{n} \varepsilon_i (\iota e_i - \tau e_i) u_i \cdot r_i = 0.$$

It follows that we can partition the indexing set $\{1, ..., n\}$ as a disjoint union

$$I_1 \cup I_2 \cup \ldots \cup I_m$$

where $i, j \in I_k$ $(k \in \{1, ..., m\})$ if and only if $r_i = r_j$, so that for each k we must have

$$\sum_{i \in I_k} \varepsilon_i (\iota e_i - \tau e_i) u_i = 0 \tag{3.14}$$

and

$$0 = \partial_1 \partial_2 \sum_{(i \in I_k)} \varepsilon_i[e_i, f'_i]$$

= $\partial_1 \sum_{(i \in I_k)} \varepsilon_i(e_i.u_i((r_i)_{+1} - (r_i)_{-1}) - (\iota e_i - \tau e_i).f'_i)$
= $\partial_1 \sum_{(i \in I_k)} \varepsilon_i e_i.u_i((r_i)_{+1} - (r_i)_{-1})$ (by (3.14))
= $(\partial_1 \sum_{(i \in I_k)} \varepsilon_i e_i.u_i).((s_k)_{+1} - (s_k)_{-1})$

where $s_k = ((s_k)_{+1}, (s_k)_{-1}) = r_i$ for all $i \in I_k$. Since $\mathbb{Z}F$ has no zero-divisors it follows that $\sum_{I_k} \varepsilon_i e_i . u_i$ is a 1-cycle, which we denote by ξ_k .

But consider the 1-boundary

$$\hat{\zeta} = \partial_2 \sum_{i=1}^n \varepsilon_i [e_i, f'_i].$$

In $B_1(\mathcal{D}) \otimes_{\mathbb{Z}F} \mathbb{Z}$ we have

$$\hat{\zeta} \otimes 1 = \sum_{i=1}^{n} \partial_2 \varepsilon_i [e_i, f_i'] \otimes 1 = \sum_{i=1}^{n} \partial_2 \varepsilon_i [e_i, f_i'] . v_i \otimes 1 = \zeta \otimes 1$$

and since

$$\hat{\zeta} = \sum_{k=1}^{m} \xi_k \cdot ((s_k)_{+1} - (s_k)_{-1})$$

we can write

$$\zeta \otimes 1 = \hat{\zeta} \otimes 1 = \sum_{k=1}^{m} \xi_k \cdot ((s_k)_{+1} - (s_k)_{-1}) \otimes 1.$$

Proof of Theorem 3.4.1: It suffices to show that the map d in the exact sequence (3.13) applied to a generator $\xi(\iota f - \tau f) \otimes 1$ (ξ a 1-cycle and $f \in e^+$) of L is zero. Since **p** is a homology trivializer of \mathcal{D} there is some 2-chain $\zeta \in C_2 \oplus C_2^{\mathbf{p}}$ such that $\tilde{\partial}_2 \zeta = \xi$. Then $\zeta . (\iota f - \tau f) \in K^{\mathbf{p}}$ is such that $\zeta . (\iota f - \tau f) \otimes 1$ satisfies

$$(\tilde{\partial}_2 \otimes 1)(\zeta \cdot (\iota f - \tau f) \otimes 1) = \xi \cdot (\iota f - \tau f) \otimes 1.$$

But in $(C_2 \oplus C_2^{\mathbf{p}}) \otimes_{\mathbb{Z}F} \mathbb{Z}$

$$\begin{aligned} \zeta.(\iota f - \tau f) \otimes 1 &= \zeta.\iota f \otimes 1 - \zeta.\tau f \otimes 1 \\ &= \zeta \otimes 1 - \zeta \otimes 1 \\ &= 0. \end{aligned}$$

It follows easily from the definition of d that $d(\xi.(\iota f - \tau f) \otimes 1) = 0$

Chapter 4

Some new finiteness conditions

Abstract

In this chapter we introduce new finiteness conditions FDT_2 and FHT_2 , and prove that they are invariants of finite presentations, finite trivializers and of retractions. In §4.4 we show that monoids with finite complete rewriting systems are FDT_2 and FHT_2 .

4.1 The properties FDT_2 and FHT_2

Definition 4.1.1 We shall say that the finite rewriting system \mathcal{P} is of second order finite derivation type (FDT_2) if:

- 1. it is of type FDT;
- 2. for some finite homotopy trivializer \mathbf{p} of \mathcal{D} the 3-complex $\overline{\mathcal{D}^{\mathbf{p}}}$ has a finite set \mathbf{X} of spherical subcomplexes such that attaching 3-cells to the set $F.\mathbf{X}.F$ gives a 3-complex with trivial second homotopy groups.

Definition 4.1.2 We shall say that the finite rewriting system \mathcal{P} is of second order finite homological type (FHT_2) if:

1. it is of type FHT;

2. for some finite homology trivializer **p** of \mathcal{D} there is a finite set **Y** of 2-cycles whose homology classes generate the $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodule $\Pi_2 = H_2(\overline{\mathcal{D}P})$.

It is immediate from the definitions that if a finite rewriting system has the property FDT_2 then it has the property FHT_2 also.

Remark 4.1.3 Suppose that for some rewriting system \mathcal{P} we choose a subset Υ of F containing a unique representative of each congruence class. If \mathbf{p} is some set of closed paths in \mathcal{D} then there is the Hurewicz homomorphism [48]

$$\bigoplus_{w \in \Upsilon} \pi_2(\overline{\mathcal{D}^{\mathbf{p}}}, w) \longrightarrow H_2(\overline{\mathcal{D}^{\mathbf{p}}})$$
(4.1)

which maps the homotopy class of a continuous map of S^2 into $\overline{\mathcal{D}^p}$ based at $w \in \Upsilon$ to the homology class of the corresponding 2-cycle. Furthermore if **p** is a homotopy trivializer, so that $\overline{\mathcal{D}^p}$ is simply-connected, then by the Hurewicz Isomorphism Theorem [48] the above homomorphism is in fact an isomorphism.

Thus if **p** is a homotopy trivializer of \mathcal{D} and if **X** is a set of spherical subcomplexes of $\overline{\mathcal{D}^{\mathbf{p}}}$, then the 3-complex obtained by attaching 3-cells according to the set $F.\mathbf{X}.F$ has trivial second homotopy groups if and only if the homology classes of the 2-cycles arising from the set **X** generate the bimodule $\Pi_2 = H_2(\overline{\mathcal{D}^{\mathbf{p}}})$. In particular, we deduce that

for rewriting systems which are of type FDT, the properties FDT_2 and FHT_2 are equivalent.

Consequently,

if the properties FDT and FHT turn out to be equivalent¹, then the properties FDT_2 and FHT_2 are also equivalent.

¹A recent example of Pride and Otto shows that this is not the case.

4.2 Invariance properties

4.2.1 Invariance of finite presentation and trivializer

We shall say that a finitely presented monoid S has the property FDT_2 (respectively, FHT_2) if some finite rewriting system $\mathcal{P} = [\mathbf{x}; \mathbf{r}]$ of S is of type FDT_2 (respectively, FHT_2). Taking account of Remark 4.1.3, this amounts to saying that for some finite homotopy (respectively, homology) trivializer \mathbf{p} of \mathcal{D} , the ($\mathbb{Z}S, \mathbb{Z}S$)-bimodule $\Pi_2(\mathcal{D}, \mathbf{p})$ is finitely generated.

Theorem 4.2.1 The properties FDT_2 and FHT_2 are monoid invariants, that is, they are independent of the choice of finite rewriting system and finite trivializer.

Proof: Suppose that $Q = [\mathbf{y}; \mathbf{s}]$ is another finite rewriting system presenting S, with finite homotopy or homology trivializer \mathbf{q} . By a result of Ivanov [32, Proposition 1.7] the $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodules

$$\mathcal{M}(\mathcal{P}) \oplus \mathbb{Z}S.\mathbf{y}.\mathbb{Z}S$$
 and $\mathcal{M}(\mathcal{Q}) \oplus \mathbb{Z}S.\mathbf{x}.\mathbb{Z}S$

are isomorphic, where \mathcal{M} denotes the relation bimodule (see §1.2.2). By adding free summands to the sequence (1) for \mathcal{P} and then splicing with (4) we obtain an exact sequence:

$$0 \longrightarrow \Pi_2(\mathcal{P}, \mathbf{p}) \longrightarrow \mathbb{Z}S.\mathbf{p}.\mathbb{Z}S \longrightarrow \mathbb{Z}S.\mathbf{r}.\mathbb{Z}S \oplus \mathbb{Z}S.\mathbf{y}.\mathbb{Z}S \longrightarrow \cdots$$

 $\cdots \longrightarrow \mathcal{M}(\mathcal{P}) \oplus \mathbb{Z}S.\mathbf{y}.\mathbb{Z}S \longrightarrow 0$

A similar sequence can be constructed for the pair (Q, q). Then applying the generalized Schanuel Lemma (Lemma 1.3.1) to these two sequences, we deduce that

 $\Pi_2(\mathcal{P},\mathbf{p}) \oplus \mathbb{Z}S.\mathbf{q}.\mathbb{Z}S \oplus \mathbb{Z}S.\mathbf{r}.\mathbb{Z}S \oplus \mathbb{Z}S.\mathbf{y}.\mathbb{Z}S$

and

$$\Pi_2(\mathcal{Q},\mathbf{q}) \oplus \mathbb{Z}S.\mathbf{p}.\mathbb{Z}S \oplus \mathbb{Z}S.\mathbf{s}.\mathbb{Z}S \oplus \mathbb{Z}S.\mathbf{x}.\mathbb{Z}S$$

are isomorphic as $(\mathbb{Z}S,\mathbb{Z}S)$ -bimodules. Therefore $\Pi_2(\mathcal{P},\mathbf{p})$ is finitely generated as a $(\mathbb{Z}S,\mathbb{Z}S)$ -bimodule if and only if $\Pi_2(\mathcal{Q},\mathbf{q})$ is.

4.2.2 Invariance under retraction

Theorem 4.2.2 A retract of an FDT_2 (respectively, FHT_2) monoid is also of this type.

Proof.Suppose that $\pi: S \longrightarrow R$ is a retraction of monoids. If S is finitely presented then we can choose finite rewriting systems $\mathcal{P} = [\mathbf{x}; \mathbf{r}]$ and $\mathcal{P}_0 = [\mathbf{x}_0; \mathbf{r}_0]$ for S and R respectively, and such that $\mathbf{x}_0 \subseteq \mathbf{x}, \mathbf{r}_0 \subseteq \mathbf{r}$ and there is a homomorphism $\rho: F(\mathbf{x}) \longrightarrow F(\mathbf{x}_0)$ such that $\rho(x) = x$ for all $x \in x_0$ and also $(\rho(r_{+1}), \rho(r_{-1})) \in \mathbf{r}_0$ for each $r \in \mathbf{r}$ (see [66, Theorem 3.3]). We can therefore extend ρ to a retraction

$$\rho: \mathcal{D} = \mathcal{D}(\mathcal{P}) \longrightarrow \mathcal{D}_0 = \mathcal{D}(\mathcal{P}_0)$$

of 2-complexes.

Suppose that **p** is a (finite) set of closed paths in \mathcal{D} , and let $\mathbf{p}_0 = \rho(\mathbf{p})$. By enlarging **p** if necessary, we can assume that $\mathbf{p}_0 \subseteq \mathbf{p}$. By [66, Lemma 3.3], if **p** is a homotopy (respectively, homology) trivializer for \mathcal{D} , then \mathbf{p}_0 is a homotopy (respectively, homology) trivializer for \mathcal{D}_0 .

We can then extend ρ again to give a retraction

$$\rho: (\overline{\mathcal{D}^{\mathbf{p}}}) \longrightarrow (\overline{\mathcal{D}_0^{\mathbf{p}_0}})$$

which induces a chain map from the chain complex of $\overline{\mathcal{D}^{\mathbf{p}}}$ to the chain complex of $\overline{\mathcal{D}_{0}^{\mathbf{p}_{0}}}$, and then induced homomorphisms on homology. In particular, we have a surjective (group) homomorphism

 $\rho_*: \Pi_2(\mathcal{P}, \mathbf{p}) \longrightarrow \Pi_2(\mathcal{P}_0, \mathbf{p}_0)$

which respects the bimodule structures: for any $s_1, s_2 \in S$ and $[\xi] \in \Pi_2(\mathcal{P}, \mathbf{p})$ we have

$$\rho_*(s_1.[\xi].s_2) = \pi(s_1).\rho_*[\xi].\pi(s_2).$$

It follows that the image of a (finite) set of generators of the $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodule $\Pi_2(\mathcal{P}, \mathbf{p})$ will generate $\Pi_2(\mathcal{P}_0, \mathbf{p}_0)$ as a $(\mathbb{Z}R, \mathbb{Z}R)$ -bimodule.

4.3 The properties FDT_2 and FHT_2 for groups

For an FHT monoid S we may splice the exact sequences (5) and (1.8) of left ZSmodules arising from some choice of finite rewriting system and homology trivializer to obtain the partial resolution

$$0 \longrightarrow \Pi_2^l \longrightarrow \mathbb{Z}S.\mathbf{p} \longrightarrow \mathbb{Z}S.\mathbf{r} \longrightarrow \mathbb{Z}S.\mathbf{x} \longrightarrow \mathbb{Z}S \longrightarrow \mathbb{Z} \longrightarrow 0$$

$$(4.2)$$

of the trivial left $\mathbb{Z}S$ -module \mathbb{Z} . As in §1.3.2 we can use the generalized Schanuel's Lemma (Lemma 1.3.1) to deduce that

S is of type left FP_4 if and only if Π_2^l is finitely generated as a left $\mathbb{Z}S$ -module.

Theorem 4.3.1 For finitely presented groups, the properties FDT_2 , FHT_2 and FP_4 are equivalent.

Proof: Since for finitely presented groups the properties FDT and FHT are equivalent §1.3.2), by Remark 4.1.3 the properties FDT_2 and FHT_2 are also equivalent.

We now show the equivalence of FHT_2 and FP_4 . Let \mathcal{P} be a finite rewriting system defining a group S, and let \mathcal{D} have a finite homotopy trivializer \mathbf{p} . As remarked above, S is FP_4 if and only if Π_2^l is finitely generated as a left module. The bimodule Π_2 has a decomposition analogous to (1.5), namely as a direct sum of the second homology groups of each connected component of $\overline{\mathcal{D}^{\mathbf{p}}}$, so as in §1.3.4, we can apply Lemma 1.3.5 to see that Π_2 is finitely generated as a bimodule (that is, S is FHT_2) if and only if Π_2^l is finitely generated as a left module, and the result follows.

4.4 Monoids with finite complete rewriting systems are FDT_2

This section contains two proofs of the following result:

Theorem 4.4.1 A monoid with a finite complete rewriting system is of type FDT_2 .

The first uses homology, which was seen in Remark 4.1.3 to be an adequate description of the second homotopy of $\overline{\mathcal{D}^{\mathbf{p}}}$ when **p** is a homotopy trivializer. The second proof uses *pictures* to study the homotopy directly.

4.4.1 2-cycles at critical triples

Throughout this section, we assume that \mathcal{P} is a finite complete rewriting system, and that **p** is some finite homotopy trivializer arising from a choice of resolution of each critical pair. Because of the choice of **p**, the boundary of each 2-cell in $\overline{\mathcal{D}^{\mathbf{p}}}$ consists of a pair of positive paths with common initial and terminal vertices, and we shall again use the notation [w; (e, f)] for 2-cells, where $w \in F$ and $e, f \in star^+(w)$, which was introduced in Remark 1.4.14. Recall that $\succ_{\mathbf{r}}$ denotes the reduction ordering induced by the rules of \mathcal{P} ; thus for $v, w \in F$, we write

 $w \succ_{\mathbf{r}} v$ if $w \rightarrow^*_{\mathbf{r}} v$ but $v \neq w$.

We shall extend this notation to 2-cells and 2-chains, writing $\sigma \succ_{\mathbf{r}} \sigma'$ for 2-cells σ and σ' if their respective maximal vertices w and v, say, satisfy $w \succ_{\mathbf{r}} v$, and $\xi \succ_{\mathbf{r}} \zeta$ for 2-chains ξ and ζ if for each 2-cell σ' represented in ζ there is some σ represented in ξ such that $\sigma \succ_{\mathbf{r}} \sigma'$.

From the construction of $\overline{\mathcal{D}^{\mathbf{p}}}$, for any $w \in F$ and triple of edges $e, f, g \in star^+(w)$ with at least one disjoint from the remaining pair, there is a 3-cell whose boundary gives rise to a 2-cycle $\xi_{(e,f,g)}$ of the form

$$\xi_{(e,f,g)} = [w; (e,f)] + [w; (f,g)] - [w; (e,g)] + \zeta_{(e,f,g)}$$

$$(4.3)$$

where $\zeta_{(e,f,g)}$ is a 2-chain satisfying $\xi_{(e,f,g)} \succ_{\mathbf{r}} \zeta_{(e,f,g)}$ (see Remark 3.3.5). Moreover, we can suppose that the number of terms in $\zeta_{(e,f,g)}$ is not greater than k-1, where $k \ge 4$ is the maximal length of the boundary of any 2-cell in $\overline{\mathcal{D}^{\mathbf{p}}}$.

The critical pairs of a rewriting system are described in Definition 1.4.6; similarly, we may consider the *critical triples*:

Definition 4.4.2 A triple of positive edges with the same initial vertex form a critical triple if either:

- 1. One of the triple is both left- and right-reduced; or
- 2. One of the triple is left-reduced but not right-reduced, one of the remaining pair of edges is right-reduced but not left-reduced, and no single edge is disjoint from the other two.

Example 4.4.3 The rewriting system given in Example 2.1.5 has a single critical triple (of type 2):



We want to show that there is also a 2-cycle with boundary of the form (4.3) for each critical triple of edges (Definition 4.4.2), but we first need the following technical definition and lemma.

For any $w \in F$ we shall say that P(w) holds if for any pair $p = e_1 \dots e_m$ and $q = f_1 \dots f_n$ of positive paths from w to Irr(w) there is a 2-chain whose boundary is the 1-cycle $\sum_{i=1}^m e_i - \sum_{j=1}^n f_j$ arising from the closed path pq^{-1} , and which can be written as either

1.
$$[w; (e_1, f_1)] + \zeta$$
 (if we assume that $e_1 <_w f_1$), or

2.
$$\zeta$$
 (if $e_1 = f_1$),

where ζ is a 2-chain which, if non-zero, contains only 2-cells with maximum vertices beneath w in the reduction order \succ_r .

Lemma 4.4.4 For all $w \in F$, P(w) holds.

Proof: The proof is by noetherian induction on F using the reduction order $\succ_{\mathbf{r}}$ induced by the rules \mathbf{r} .

Let $w \in F$, and assume inductively that P(w') holds for each $w' \in F$ satisfying $w \succ_{\mathbf{r}} w'$. We want to deduce that P_w holds.

If w = Irr(w) then P(w) holds since we may just take the trivial 2-chain 0.

So suppose that $w \neq Irr(w)$, and that $p = e_1 \dots e_m$ and $q = f_1 \dots f_n$ are positive paths from w to Irr(w). Thus $m, n \ge 1$, and so either $e_1 = f_1$, in which case by our inductive hypothesis there is a suitable 2-chain with boundary corresponding to the closed path $e_2 \dots e_m f_n^{-1} \dots f_2^{-1}$, or there is a 2-cell $[w; (e_1, f_1)]$ with boundary $e_1 p'(q')^{-1} f_1^{-1}$, say, for some positive paths p', q'. If we choose a positive path r from $\tau p' = \tau q'$ to Irr(w) then by inductive assumption there are 2-chains ζ_{e_1} and ζ_{f_1} whose boundaries are 1-cycles arising from the closed paths $e_2 \dots e_m r^{-1}(p')^{-1}$ and $f_2 \dots f_n r^{-1}(q')^{-1}$ respectively, and with $[w; (e_1, f_1)] \succ_r \zeta_{e_1}, \zeta_{f_1}$. (This construction is described by Figure 1.2.)

Then $[w; (e_1, f_1)] + \zeta_{e_1} - \zeta_{f_1}$ is a suitable 2-chain for this pair of paths , and so P(w) holds. The lemma now follows by noetherian induction (Lemma 1.4.2).

We shall now show that there is also a 2-cycle with boundary of the form (4.3) for each critical triple of edges.

Suppose that (e_1, e_2, e_3) is a critical triple of edges at $w \in F$ with $e_1 <_w e_2 <_w e_3$. Then for $1 \le i < j \le 3$ there is a 2-cell $[w; (e_i, e_j)]$ with boundary $e_i p_{i,j} q_{i,j}^{-1} e_j^{-1}$ say, where $p_{i,j}$ and $q_{i,j}$ are positive paths. If for $1 \le i < j \le 3$ we also choose positive paths $r_{i,j}$ from $\tau p_{i,j}$ to Irr(w), then by Lemma 4.4.4 there exist 2-chains ζ_1, ζ_2 and ζ_3 whose boundaries are 1-cycles corresponding to the closed paths $p_{1,3}r_{1,3}r_{1,2}^{-1}p_{1,2}^{-1}$, $q_{1,2}r_{1,2}r_{2,3}^{-1}p_{2,3}^{-1}$ and $q_{2,3}r_{2,3}r_{1,3}^{-1}q_{1,3}^{-1}$ respectively, from which we can construct the 2-cycle

 $\xi_{(e_1,e_2,e_3)} = [w;(e_1,e_2)] - [w;(e_1,e_3)] + [w;(e_2,e_3)] + \zeta_1 + \zeta_2 + \zeta_3,$

which is of the form (4.3), because $\xi_{(e_1,e_2,e_3)} \succ_{\mathbf{r}} \zeta_1 + \zeta_2 + \zeta_3$.

Let X be a set consisting of such a 2-cycle $\xi_{(e,f,g)}$ for each critical triple (e, f, g)of positive edges. For a pair of 2-cycles z, z' we shall write $z \sim_{\mathbf{X}} z'$ if z - z' is in the sub-bimodule of the $(\mathbb{Z}F, \mathbb{Z}F)$ -bimodule $Z_2(\overline{\mathcal{D}P})$ generated by $B_2(\overline{\mathcal{D}P})$ and X.

Any triple of positive edges at some vertex, where no edge is disjoint from the other pair, is a translate of a critical triple by the two-sided *F*-action. Therefore, by our choice of the set X there is now a 2-cycle $\xi_{(e,f,g)}$ of the form (4.3) for any $w \in F$ and triple (e, f, g) of positive edges in $star^+(w)$ $(w \in F)$ such that

$$\xi_{(e,f,g)} = [w; (e,f)] + [w; (f,g)] - [w; (e,g)] + \zeta_{(e,f,g)} \sim_{\mathbf{X}} 0,$$

where $\xi_{(e,f,g)} \succ_{\mathbf{r}} \zeta_{(e,f,g)}$. Moreover, there is a constant c (depending on the choice of set **X** and the maximal length of the boundary of any 2-cell in $\overline{\mathcal{D}^{\mathbf{p}}}$) which bounds the number of terms in each such 2-cycle.

4.4.2 Proof of Theorem 4.4.1

We shall show that the homology classes of the set \mathbf{X} generate Π_2 .

Let ξ be a 2-cycle. We can write ξ as

$$\xi = \sum_{i=1}^{m} n_i \sigma_i$$

where $m \ge 0, \sigma_1, \ldots, \sigma_m$ are distinct 2-cells, and $n_1, \ldots, n_m \in \mathbb{Z} - \{0\}$. Each 2-cell can be written as $\sigma_i = [w_i; (f_i, g_i)]$ for $i = 1, \ldots, m$, where $w_i \in F$ and $f_i, g_i \in star^+(w_i)$.

Denote by $\Omega = \Omega(\xi)$ the set of 2-cells of the form [u; (e, f)] where $u \in F$ is a *descendant* of one of the vertices w_1, \ldots, w_m , that is

$$u \in \bigcup_{i=1}^{m} \{ v \in F : w_i \to^* v \},$$

and $e, f \in star^+(u)$. Because Γ is locally finite, it follows from Lemma 1.2.2 that Ω is finite. We partially order Ω as follows: For $u_i \in F$ and positive edges $e_i, e'_i \in star^+(u_i)$ (i = 1, 2) we write $[u_1; (e_1, e'_1)] < [u_2; (e_2, e'_2)]$ if (i) $u_2 \succ_{\mathbf{r}} u_1$; or

(ii) $u_1 = u_2$ and $e'_1 <_{u_1} e'_2$; or

(iii) $u_1 = u_2, e'_1 = e'_2$ and $e_1 <_{u_1} e_2$.

Next, choose some bijection

$$\psi: \Omega \longrightarrow \{1, 2, \ldots, |\Omega|\}$$

which is compatible with this ordering, and define

$$\Psi(\xi) = \sum_{i=1}^m c^{\psi \sigma_i}.$$

Suppose that ξ is non-zero (that is, m > 0), and assume that $\psi(\sigma_1) > \psi(\sigma_i)$ for all $1 < i \le m$. If the edge f_1 is not the left principal edge e at w_1 , then we can replace $n_1\sigma_1$ with a 2-chain

$$n_1([w_1;(e,g_1)] - [w_1;(e,f_1)] - \zeta_{(e,f_1,g_1)})$$

where $\sigma_1 \succ_{\mathbf{r}} \zeta_{(e,f_1,g_1)}$, giving a 2-cycle ξ' with $\xi' \sim_{\mathbf{X}} \xi$ and $\Psi(\xi') < \Psi(\xi)$.

Otherwise f_1 is left principal, and so because σ_1 is maximal it is the only 2-cell represented in ξ with the edge g_1 in its boundary. Since ξ is a 2-cycle it follows that $n_1 = 0$, a contradiction. By induction, $\xi \sim_{\mathbf{X}} 0$.

4.5 An alternative proof of Theorem 4.4.1 using homotopy

Again let \mathcal{P} be a finite complete rewriting system, and \mathbf{p} a finite trivializer arising from a choice of resolution of each critical pair. To study the second homotopy of $\overline{\mathcal{D}^{\mathbf{p}}}$ we shall use *pictures*:

4.5.1 Pictures

In this subsection we shall only review the material that we shall need; more details and proofs can be found in, for example, [6, 31]. A picture \mathbb{P} over $\overline{\mathcal{D}^{p}}$ consists of the following:

- 1. A disc D^2 .
- 2. Disjoint discs d_1, \ldots, d_m in the interior of D^2 .
- 3. A finite number of disjoint arcs $\alpha_1, \ldots, \alpha_n$. Each arc lies in the closure of $D^2 (d_1 \cup d_2 \cup \cdots \cup d_m)$ and is either a simple closed curve having trivial intersection with the boundaries of the discs D^2, d_1, \ldots, d_m , or is a simple non-closed curve whose intersection with the boundaries of these discs consists of its endpoints only.
- 4. Each arc has a normal orientation indicated by a transverse arrow and is labelled by a positive edge of $\overline{\mathcal{D}^{p}}$.
- 5. Travelling around the boundary ∂d_i of each interior disc d_i reading the edges labelling the arcs with endpoints on the boundary will give the boundary of a 2-cell in $\overline{\mathcal{D}^{\mathbf{p}}}$. If we cross an arc labelled by a positive edge e in the direction of its normal orientation then we read e; otherwise we read e^{-1} .
- 6. The regions of \mathbb{P} are the connected components of

$$D^2 - (\partial D^2 \cup \bigcup_{i=1}^n d_i \cup \bigcup_{j=1}^m \alpha_j)$$

and each region is labelled by an element of F: If we travel across an arc labelled by a positive edge e in the direction of its normal orientation then we move from a region labelled ιe to a region labelled τe .

We define $\partial \mathbb{P}$ to be ∂D^2 . By travelling around this boundary we read a closed path in the 1-skeleton Γ of $\overline{\mathcal{D}^p}$. We say that \mathbb{P} is *spherical* if this is the empty path, which is to say that no arc meets $\partial \mathbb{P}$: When a picture is spherical we call the region surrounding $\partial \mathbb{P}$ the *outer region*, and $\partial \mathbb{P}$ is not drawn. The *area* of \mathbb{P} is the number m of interior discs, and the *components* of \mathbb{P} are the connected components of

$$\bigcup_{i=1}^n d_i \cup \bigcup_{j=1}^m \alpha_j.$$

We say that \mathbb{P} is *connected* if it has at most one component.

Suppose we embed a circle in $\mathbb{P} - \bigcup_i d_i$ so that it meets the arcs in only finitely many transverse intersections. Then the part of \mathbb{P} enclosed by this circle is said to be a *subpicture* of \mathbb{P} . There is a 2-sided action of F on the collection of pictures induced by the action of F on the labels of the regions and arcs.

Each picture describes a continuous map of the disc into the 2-skeleton of $\overline{\mathcal{D}^{p}}$ in the following way. Each interior disc maps onto the 2-cell whose boundary corresponds to the boundary of the disc, each arc maps onto the mid-point of the edge labelling it, and each region is mapped in some way to the vertex w labelling it and a subset of

 $\{e, e^{-1} : e \in star^+(w)\}.$

If \mathbb{P} is a spherical picture then we may assume that all of $\partial \mathbb{P}$ is mapped to the vertex $w \in F$ which labels the outer region, so that \mathbb{P} describes a continuous map of the 2-sphere S^2 into $\overline{\mathcal{D}^p}$. The homotopy class of \mathbb{P} is the corresponding element of the second homotopy group $\pi_2(\overline{\mathcal{D}^p}, w)$, and in fact every element of the second homotopy groups of $\overline{\mathcal{D}^p}$ can be represented in this way by a spherical picture. We shall use $\xi(\mathbb{P})$ to denote the 2-chain obtained by reading the labels on each interior disc of a picture \mathbb{P} .

Example 4.5.1 The connected spherical picture in Figure 4.1 describes a map of S^2 onto the boundary of the 3-cell [e, [f, f']] $(e, f, f' \in e^+)$, and $\xi(\mathbb{P})$ is the 2-cycle

$$[e, f] \cdot \iota f' - \tau e \cdot [f, f'] - [e, \iota f \cdot f'] + [e, \tau f \cdot f'] - [e, f] \cdot \tau f' + \iota e \cdot [f, f']$$

= $(\iota e - \tau e) \cdot [f, f'] + [e, \tau f \cdot f'] - [e, \iota f \cdot f'] + [e, f] \cdot (\iota f' - \tau f'),$



Figure 4.1: A spherical picture

where the coefficient gives the orientation of the 2-cell. The outer region is labelled $\iota e\iota f\iota f'$, and the labelling of the unmarked edges and regions follows from the labelling of the interior discs.

Suppose that Υ is a subset of F containing a unique representative of each congruence class. The Hurewicz homomorphism (4.1) maps the homotopy class of a spherical picture \mathbb{P} with outer region labelled by some $w \in F$ to the homology class of the 2-cycle $\xi(\mathbb{P})$. Since \mathbf{p} is a homotopy trivializer then the above homomorphism is in fact an isomorphism, as noted in §4.1.

4.5.2 3-cells at critical triples

Since a spherical picture over $\overline{\mathcal{D}^{\mathbf{p}}}$ describes a continuous map of S^2 into the 2-skeleton of the augmented Squier complex, we can describe the attaching map of each 3-cell in $\overline{\mathcal{D}^{\mathbf{p}}}$ in this way.

From our choice of trivializer **p**, and by the construction of $\overline{\mathcal{D}^{\mathbf{p}}}$, for any $w \in F$ and triple of edges $e, f, g \in star^+(w)$ with at least one disjoint from the remaining pair



Figure 4.2: Spherical picture at triple (e_1, e_2, e_3)

there is a 3-cell whose attaching map can be described by a spherical picture of the form shown in Figure 4.2 for some subpicture \mathbb{Q} , with the outer region labelled w, and all other regions labelled by vertices v_i satisfying $\vartheta v_i < \vartheta w$, where ϑ is the height function (Definition 1.2.3). Furthermore, we can assume that the area of this picture is not greater than k + 2, where $k \ge 4$ is the maximum length of the boundary of any 2-cell.

We shall now give a set of spherical pictures describing the attaching maps of a set of 3-cells at each critical triple (Definition 4.4.2), and which have the same form as in Figure 4.2. We need the following lemma:

Lemma 4.5.2 For any pair of positive paths $p = e_1 \dots e_m$ and $q = f_1 \dots f_n$ from the same initial vertex $w \in F$ to Irr(w) there is a picture over $\overline{D^p}$ with boundary pq^{-1} with only one region labelled w and such that any other region is labelled by some $v \in F$ with $\vartheta v < \vartheta w$, where ϑ is the height function.

Proof: The proof is by noetherian induction on F with the reduction order $\succ_{\mathbf{r}}$ induced by the rules \mathbf{r} .

Let $w \in F$ and assume inductively that for any $w' \in F$ satisfying $w \succ_{\mathbf{r}} w'$ and pair of paths r, s from w' to Irr(w') there is a picture over $\overline{\mathcal{D}^{\mathbf{p}}}$ with boundary rs^{-1} with only one region labelled w' and such that any other region is labelled by some $v \in F$ with $\vartheta v < \vartheta w'$. We want to construct a suitable picture for w.



Figure 4.3: Construction used in the proof of Lemma 4.5.2

If w = Irr(w) then p and q are both the empty path at w and we just take the empty picture with a single region labelled w.

So suppose that $w \neq Irr(w)$, and that $p = e_1 \dots e_m$ and $q = f_1 \dots f_n$ are a pair of positive paths from w to Irr(w), with $m, n \ge 1$. If $e_1 = f_1$ then there is by inductive hypothesis a picture \mathcal{P} with boundary $e_2 \dots e_m f_n^{-1} \dots f_2^{-1}$ where a single region is labelled τe_1 , and all other regions labelled by some $v \in F$ with $\vartheta v < \vartheta \tau e_1 < \vartheta w$. A suitable picture can be obtained from \mathcal{P} by simply adding an arc with label $e_1 = f_1$ as shown in Figure 4.3.

Otherwise, there is a 2-cell $[w; (e_1, f_1)]$ with boundary $e_1 p'(q')^{-1} f_1^{-1}$ for some pair of positive paths p' and q'. If we choose a positive path r from $\tau p' = \tau q'$ to Irr(w) then by our inductive assumption we have pictures \mathbb{P} and \mathbb{Q} with boundaries $\partial \mathbb{P} = e_2 \dots e_m r^{-1} (p')^{-1}$ and $\partial \mathbb{Q} = q' r f_n^{-1} \dots f_2^{-1}$ giving the picture shown in Figure 4.4. The lemma follows by noetherian induction (Lemma 1.4.2).

Proposition 4.5.3 We can construct a spherical picture of the form shown in Figure 4.2 for each critical triple (e_1, e_2, e_3) of positive edges at $w \in F$.

Proof. For $1 \le i < j \le 3$ there is a 2-cell $[w; (e_i, e_j)]$ with boundary $e_i p_{i,j} q_{i,j}^{-1} e_j^{-1}$ say, where $p_{i,j}$ and $q_{i,j}$ are positive paths. If we also choose positive paths $r_{i,j}$ from $\tau p_{i,j}$ to Irr(w) then by using Lemma 4.5.2 we can construct a spherical picture of the form shown in Figure 4.5 from pictures $\mathbb{P}_1, \mathbb{P}_2, \mathbb{P}_3$ which have boundaries $p_{1,3}r_{1,3}r_{1,2}^{-1}p_{1,2}^{-1}$,



Figure 4.4: Second construction used in the proof of Lemma 4.5.2



Figure 4.5: Spherical picture at a critical triple.



Figure 4.6: Cancelling pair

 $q_{1,2}r_{1,2}r_{2,3}^{-1}p_{2,3}^{-1}$ and $q_{2,3}r_{2,3}r_{1,3}^{-1}q_{1,3}^{-1}$ respectively. Then there is only one region labelled w, and every other region is labelled by some $v \in F$ satisfying $\vartheta v < \vartheta w$.

We choose such a spherical picture for each critical pair, and we claim that 3-cells attached according to this set of pictures together with their translates under the 2-sided action of F trivialize the second homotopy groups of $\overline{\mathcal{D}^p}$, so that S is of type FDT_2 . Note that this choice has given us a set \mathbf{X} of spherical pictures of the form shown in Figure 4.2 for every triple of positive edges in $star^+(w)$ for every $w \in F$.

4.5.3 Operations on pictures

Let X be the chosen set of spherical pictures, one for every triple of positive edges in $star^+(w)$ for some $w \in F$. By an X-picture we will mean either a picture \mathbb{P} from X, or $-\mathbb{P}$, the picture obtained by a planar reflection of \mathbb{P} .

We allow the following operations on spherical pictures.

- (A) Deletion of a closed arc which encircles no arcs or discs (a floating circle).
- (B) Deletion of a cancelling pair, a subpicture of the form shown in Figure 4.6.
- (C) Bridge move: see Figure 4.7.
- (D) Replace(X), the replacement of an interior disc with label a 2-cell [w; (f, f')]where $w \in F$ and $f, f' \in star^+(w)$ are not left principal edges. If we let e denote the left principal edge at w then this operation is carried out as follows. First



Figure 4.7: Bridge move



Figure 4.8: Insertion of X-picture

we insert the appropriate X-picture corresponding to the triple (e, f, f') into the original picture as shown in Figure 4.8. Next, a series of bridge moves can be applied to obtain the picture shown in Figure 4.9, and we then delete the cancelling pair to finish with the picture shown in Figure 4.10.

Operations (A), (B) and (C) do not change the homotopy class of the picture \mathbb{P} . This is also true of operation (D) if there are 3-cells whose attaching maps correspond to the X-pictures, for then we are essentially just pushing the image of \mathbb{P} across the appropriate 3-cell.

4.5.4 A 2-dimensional Dehn-type algorithm

Let \mathbb{P} be a spherical picture over $\mathcal{D}^{\mathbf{p}}$ whose regions are labelled by a set of vertices $w_1, w_2, \ldots, w_n \in F$. Following [6] we shall use a type of "2-dimensional" Dehn algorithm


Figure 4.9: Subpicture after performing bridge moves



Figure 4.10: Result of $replace(\mathbf{X})$

to show that \mathbb{P} is X-equivalent to the empty picture. The algorithm consists of a series of transformations of the original picture which we now describe.

Firstly we remove any floating circle in \mathbb{P} . If there is an interior disc with label a 2-cell [w; (f, f')] for some $w \in F$ and positive edges $f, f' \in star^+(w)$ neither of which are left principal then we can perform an operation (D) and we are done.

Otherwise, choose some connected spherical subpicture of \mathbb{P} , and in this subpicture choose a region that is labelled by some vertex (w_1, say) that is maximal with respect to the height function ϑ . If this region is bounded by a pair of discs and arcs of the form as in Figure 4.11 where $\sigma = [w_1; (e, f)]$ (for edges $e, f \in star^+(w_1)$ with eleft principal) has boundary $epq^{-1}f^{-1}$ where p, q are positive paths, then by a series of bridge moves this subpicture can be transformed to a subpicture containing a cancelling pair as shown in Figure 4.12. and we can now remove the cancelling pair, at the same time removing the region with label w_1 .



Figure 4.11: Subpicture composed of the discs labelled σ and $-\sigma$



Figure 4.12: New subpicture containing a cancelling pair

On the other hand it is possible that the region with label w_1 is bounded by more than 3 interior discs and arcs as shown in Figure 4.13, where the arcs are labelled alternately by the left principal edge e at w and by edges $f, f', \ldots \in star^+(w_1) - \{e\}$. In this case, we can perform a bridge move between two arcs labelled by e to obtain a subpicture of the form shown in Figure 4.11, which again leads to the removal of a cancelling pair.

It is also possible that the region labelled by w_1 is the outer region of the spherical subpicture. If this subpicture is of the form shown in Figure 4.14 for some subpicture \mathbb{Q} , then by applying bridge moves we can obtain the picture shown in Figure 4.15. from which we can remove the cancelling pair. If the subpicture is of the form shown in Figure 4.16 we can similarly apply bridge moves to isolate a cancelling pair which we then remove.

Lemma 4.5.4 For any spherical picture \mathbb{P} any sequence of transformations will terminate in a finite number of steps.



Figure 4.13: The maximal region bounded by more than two discs



Figure 4.14: The maximal region is the outer region



Figure 4.15: After performing bridge moves on Figure 4.14



Figure 4.16: The maximal region is the outer region (case 2)

Proof: For any given $u \in F$ let $\Delta(u)$ denote the subgraph of Γ formed by taking the union of all positive paths originating at u. As its vertices $\Delta(u)$ has the set

$$vert(\Delta(u)) = \{v \in F : u \to_{\mathbf{r}}^* v\}$$

of descendants of u and the edge set is

$$\{e, e^{-1}: e \in star(v), u \rightarrow_{\mathbf{r}}^* v\}.$$

Because Γ is locally finite, it follows from Lemma 1.2.2 that $\Delta(u)$ is finite. We denote by $\Omega = \Omega(\mathbb{P})$ the set of 2-cells of the form [u; (e, f)] with

$$u \in \bigcup_{i=1}^{n} vert(\Delta(w_i)) \subset F \text{ and } e, f \in star^+(u)$$

where $w_1, \ldots, w_n \in F$ are the labels of the regions of \mathbb{P} . Again this set is finite.

Note that each single transformation of a picture either removes a cancelling pair, or uses an X-picture to replace a subpicture with a single disc, with label a 2-cell σ , say, with a subpicture containing only discs labelled by 2-cells in Ω which are beneath σ in the partial ordering introduced in the proof of the special case of Proposition 3.3.4. We must show that this process terminates in a finite number of steps.

First, we choose some bijective function

 $\psi: \Omega \longrightarrow \{1, 2, \ldots, |\Omega|\}$

requiring only that for 2-cells $\sigma, \sigma' \in \Omega$ we have $\psi \sigma < \psi \sigma'$ if $\sigma < \sigma'$.

Based on the values assigned by ψ to each 2-cell of Ω we define another function

$$\Psi:\Upsilon\longrightarrow\mathbb{N}\cup\{0\}$$

by extending to the set of interior discs the function

$$\pm \sigma \mapsto k^{\psi \sigma}$$

where k + 1 is the maximum area of any X-picture.

To prove the lemma, note that applying a single transformation to any picture \mathbb{Q} in Υ will either reduce the value of $\Psi(\mathbb{Q})$ by $2k^i$ for some $1 \le i \le |\Omega|$ (if we have removed a cancelling pair), or will reduce the value of $\Psi(\mathbb{Q})$ by k^i while increasing the value by adding less than $k(k^{(i-1)})$.

Chapter 5

Alternative viewpoints

5.1 Competing finiteness conditions

5.1.1 An alternative definition of FHT_2

Our definition of FDT_2 seems to allow the possibility of further development of these ideas, for example we could study the third homotopy groups of a 4-complex obtained by attaching 3-cells to the 3-complex \overline{DP} in order to trivialize the second homotopy groups, together with certain "obvious" 4-cells.

On the other hand, in our definition of FHT_2 we only require that for some finite homology trivializer **p** the bimodule $\Pi_2(\mathcal{P}, \mathbf{p})$ is finitely generated. In this situation, where **p** may not be a homotopy trivializer, if the Hurewicz homomorphism (4.1) is not surjective then it may be that we are not in fact able to kill the second homology by attaching 3-cells $\overline{\mathcal{D}^{\mathbf{p}}}$.

Without being able to realize this geometrically, it is not clear how we could define a property FHT_3 by studying a 4-complex based on $\overline{\mathcal{DP}}$. An alternative definition of FHT_2 would require that not only the bimodule Π_2 was finitely generated, but that we could attach a finite set of 3-cells to trivialize the second homology. This problem does not exist with the definition of the property FHT because any 1-cycle arises from a (finite) set of closed paths.

104

5.1.2 The property bi- FP_n

Following Otto and Kobayashi [41], we say that a monoid S is of type bi- FP_n if the monoid ring $\mathbb{Z}S$ (thought of as a $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodule by left and right multiplication) has a partial resolution of length n by free $(\mathbb{Z}S, \mathbb{Z}S)$ -bimodules of finite rank.

Applying the functor $-\bigotimes_{\mathbb{Z}S}\mathbb{Z}$ to such a resolution will give a partial resolution of the trivial left $\mathbb{Z}S$ -module \mathbb{Z} ($\cong \mathbb{Z}S \bigotimes_{\mathbb{Z}S}\mathbb{Z}$) by finite rank free left $\mathbb{Z}S$ -modules [41]. Thus if a monoid is of type bi- FP_n then it is necessarily of type left- FP_n . Similarly, applying $\mathbb{Z} \bigotimes_{\mathbb{Z}S}$ - will show that such a monoid is also of type right- FP_n . On the other hand, Kobayashi and Otto [42] have given an example of a monoid which is both left and right FP_{∞} but is not of type bi- FP_3 .

In [41] it is shown that there is a partial resolution of $\mathbb{Z}S$ by free ($\mathbb{Z}S,\mathbb{Z}S$)bimodules analogous to (1.8), but constructed using the short exact sequence (3), giving the characterization

FHT monoids are those finitely presented monoids of type bi- FP_3 .

By splicing this partial resolution with the short exact sequence (4) arising from some choice of finite homology trivializer, we immediately have the analogous characterization

 FHT_2 monoids are those finitely presented monoids of type bi- FP_4 .

5.1.3 Homological finite derivation type

Alonso and Hermiller [2] have introduced the alternative finiteness conditions *n*dimensional homological finite derivation type $(HFDT_n)$ for all $n\geq 0$, and have shown that a monoid S of type left and right FP_n is necessarily of type $HFDT_n$. However, the example mentioned above due to Kobayashi and Otto of a monoid which is both left and right FP_{∞} but not of type FHT or FDT shows us that (in particular) the property $HFDT_3$ is not equivalent to either FHT or FDT, nor is the property $HFDT_4$ equivalent to FHT_2 or FDT_2 .

5.1.4 A finiteness condition for higher dimensional rewriting systems

In order to extend the notion of finite derivation type and finite homological type one dimension higher, we have defined a 3-complex $\overline{\mathcal{D}^{p}}$ based on \mathcal{D} and studied its second homotopy and homology. We now outline another approach (as suggested in [36, 43]) which is to consider some kind of finite derivation type on some new graph based on the derivations of a higher dimensional "rewriting system" which operates on paths in Γ .

In the next section (§5.2), we describe such a rewriting system on paths, or homotopy reduction system, which can be associated to any finite complete rewriting system. The new graph of derivations of this system (introduced in [17]) would have as its vertices the set of all paths in Γ , and there would be edges which correspond to pushing a subpath across a 2-cell in $\overline{\mathcal{D}^{p}}$.

A closed path in this new graph will correspond to a 2-cycle in $\overline{D^{p}}$, and so this method is related to our earlier study of the second homology of this 3-complex, and it is possible that they are equivalent. The advantage of our earlier approach is in obtaining the short exact sequence (4) which allows us to relate the properties FHT_2 and FDT_2 to homological finiteness conditions, and the invariance property (Theorem 4.2.1) can be proved easily using homological algebra together with known results.

5.2 A complete homotopy reduction system for complete rewriting systems

Suppose that \mathcal{P} is a finite complete rewriting system, and let \mathbf{p} be a homotopy trivializer given by resolutions of the critical pairs associated with the set of rules \mathbf{r} . As we have already noted (Remark 1.4.14), the boundary of each 2-cell σ consists of two positive paths arising from resolutions of either disjoint or critical pairs, according as to whether σ is in \mathcal{D} or not.

We now describe a homotopy reduction system for $\mathcal{D}^{\mathbf{p}}$: a "rewriting system" on paths in $\mathcal{D}^{\mathbf{p}}$, which operates by pushing subpaths across 2-cells. It will turn out that this homotopy reduction system is *complete* in the sense that in a finite number of steps it will rewrite any path to a unique irreducible path sharing the same endpoints (see Theorem 5.2.2).

A single-step homotopy reduction, applied to subpaths of paths in $\mathcal{D}^{\mathbf{p}}$, will be of one of two types:

- 1. For any $e \in e^+$ we allow reductions of the form
 - (a) $ee^{-1} \rightsquigarrow 1_{\iota e}$ (where $1_{\iota e}$ denotes the empty path at ιe); and (b) $e^{-1}e \rightsquigarrow 1_{\tau e}$.
- 2. For any pair $e, f \in star^+(w)$ there is a 2-cell in \mathcal{D}^p with boundary of the form

$$epq^{-1}f^{-1}$$

where p and q are positive paths ending at some vertex z, say, which resolve the pair (e, f). If $e <_w f$ holds we allow the reductions

- (a) $f \rightsquigarrow epq^{-1}$, and
- (b) $f^{-1} \rightsquigarrow q p^{-1} e^{-1}$,

illustrated by Figure 5.1.

We use the notation \rightsquigarrow^* to denote the reflexive and transitive closure of the relation \rightsquigarrow on paths in Γ .

Proposition 5.2.1 The homotopy reduction system just given is noetherian, that is, we can only perform a finite number of single-step homotopy reductions on any given path p.



Figure 5.1: Single-step homotopy reduction (type 2)

Proof: First note that each single-step homotopy reduction either removes an edgepair $e^{\varepsilon}e^{-\varepsilon}$ ($e \in e^+$, $\varepsilon = \pm 1$) (a Type 1 reduction), or replaces a subpath f^{ε} ($f \in e^+$, $\varepsilon = \pm 1$) with a subpath consisting of one edge e^{ε} where $e \in star^+(\iota f)$ satisfies $e <_{\iota \varepsilon} f$, together with edges of the form g^{δ} ($g \in e^+$, $\delta = \pm 1$) satisfying $\vartheta \iota g < \vartheta \iota f$, where ϑ denotes the height function (Definition 1.2.3) (a Type 2 reduction). We show that this process must terminate in a finite number of steps.

For the given path p, by Lemma 1.2.2 and the fact that Γ is locally finite, the set

$$A = A(p) = \{ v \in F : v \prec_{\mathbf{r}} w \text{ for some vertex } w \text{ of } p \}$$

of descendants of vertices in p is finite, and again because Γ is locally finite the set

$$\Omega = \Omega(p) = \{e \in \mathbf{e}^+ : \iota e, \tau e \in A\}$$

of positive edges between vertices in A is also finite. Firstly, we can order A in some arbitrary way as $A = (v_1, v_2, \dots v_{|A|})$, but requiring that if $\vartheta v_i < \vartheta v_j$ then i < j, and then we can order Ω as

$$\Omega = (e_1, e_2, \dots e_{|\Omega|})$$

= $(e_{v_1,1}, e_{v_1,2}, \dots, e_{v_1,k_{v_1}}, e_{v_2,1}, \dots, e_{v_{|A|},k_{v_{|A|}}})$

where the edges indexed by $v_i \in A$ are those edges with initial vertex v_i , and then according to the ordering of edges in $star^+(v_i)$:

if $e_{v_i,j} <_{v_i} e_{v_i,k}$, then we require that j < k.

Now we consider the set $\Xi = \Xi(p)$ of all finite paths with the same initial and terminal vertices as our original path p and composed of edges e^{ε} with $e \in \Omega$ and $\varepsilon = \pm 1$, and define a function

$$\psi: \Xi \longrightarrow \mathbb{N}_0 \qquad e_{i_1}^{\epsilon_k} \dots e_{i_k}^{\epsilon_k} \mapsto \sum_{j=1}^k n^{i_j}$$

where n is the maximum length of the boundary of any 2-cell in $\mathcal{D}^{\mathbf{p}}$.

To prove the proposition, note that applying a Type 1 single-step homotopy reduction to any path q in $\Xi(p)$ will reduce the value of $\psi(q)$ by $2n^i$ for some i > 0, and that a Type 2 reduction will reduce the value $\psi(q)$ by n^i for some i > 1 while increasing the value by adding less than $n(n^{(i-1)})$; clearly this process must terminate, as we have associated with any sequence of single-step reductions on an arbitrary path a strictly decreasing sequence in \mathbb{N}_0 .

Theorem 5.2.2 The system is complete.

Proof: The only irreducible paths are those composed of left principal edges and that have no spurs (subpaths of the form $\ldots e^{\varepsilon}e^{-\varepsilon}\ldots$, where $e \in e^+$ and $\varepsilon = \pm 1$); these are the unique edge-paths of shortest length between two points in a connected component of the maximal forest of left principal edges in Γ .

Remark 5.2.3 Homotopy reduction systems were introduced by Kobayashi [36], who considered more general complete homotopy reduction systems. The system that we have associated to a complete rewriting system is a variant of the *Guba-Sapir* reduction system [29, 39], in which the type 2 single step homotopy reductions are restricted to only those reductions which push an edge across a 2-cell from \mathcal{D} . The advantage of extending this as we have done is that the overlapping rules of the new homotopy reduction system are easily described (§5.2.1), and then there is a nice partial resolution (5.1) of Z extending the partial resolution (1.9).

5.2.1 Critical pairs of the homotopy reduction system

Associated to any finite complete rewriting system \mathcal{P} we have described a complete homotopy reduction system based on a natural homotopy trivializer \mathbf{p} of \mathcal{D} given by choosing resolutions of the critical pairs. We shall now examine the *overlapping pairs* of the homotopy reduction system, that is, the pairs of single-step homotopy reductions that can operate on non-disjoint subpaths. The analogue of the result (1.10) that a noetherian rewriting system is complete if and only if it is confluent holds for homotopy reduction systems, as does the Diamond Lemma (Lemma 1.4.3), and therefore each overlapping pair can be resolved by positive homotopy reductions to some common path in Γ .

Overlaps of Type 1 reductions

These overlaps occur on subpaths of the form $ee^{-1}e$ or $e^{-1}ee^{-1}$ $(e \in e^+)$, and are immediately resolved (to e or e^{-1} respectively).

Overlaps between Type 1 and Type 2 reductions

Suppose that we can perform a type 2 reduction

$$f \rightsquigarrow epq^{-1}$$

on a positive edge f, where $e \in e^+$ and p and q are positive paths. Then, for example, we have overlaps between the reductions

$$ff^{-1} \rightsquigarrow 1_{\iota f}$$
 and $ff^{-1} \rightsquigarrow epq^{-1}f^{-1}$.

This overlap is resolved easily as follows:

$$epq^{-1}f^{-1} \rightsquigarrow^* epq^{-1}qp^{-1}e^{-1} \rightsquigarrow^* epp^{-1}e^{-1} \rightsquigarrow^* ee^{-1} \rightsquigarrow^* 1_{\iota f}.$$

Overlaps of Type 2 reductions

These overlapping pairs are of four types (see [37]), and occur when we have edges e, f, g in $star^+(w)$ with $e <_w f <_w g$, so that there is a choice of two different Type

2 reductions at g (and, similarly, at g^{-1}). The first three are *inessential* overlapping pairs and are shown to be easily resolved, and the final type of overlapping pair will be seen to correspond precisely to translates of the critical triples of \mathcal{P} .

1. Suppose firstly that e, f and g are all disjoint. Then we can write w as $w = (\iota e')(\iota f')(\iota g')$ for some $e', f', g' \in e^+$ such that $e = e' \iota f' \iota g', f = \iota e' \cdot f' \cdot \iota g'$ and $g = \iota e' \cdot \iota f' \cdot g'$, and the overlapping pair can be resolved along the following pair of positive reductions:

$$h_1: g \rightsquigarrow e.(\tau e' \iota f'.g')(e' \iota f' \tau g')^{-1}$$
$$\rightsquigarrow e(\tau e'.f'.\iota g')(\tau e' \tau f'.g')(\tau e'.f'.\tau g')^{-1}(e'.\iota f' \tau g')^{-1}$$

and

$$h_{2}: g \rightsquigarrow f(\iota e'\tau f'.g')(\iota e'.f'.\tau g')^{-1} \\ \sim e(\tau e'.f'.\iota g')(e'.\tau f'\iota g')^{-1}(\iota e'\tau f'.g')(\iota e'.f'.\tau g')^{-1} \\ \sim e(\tau e'.f'.\iota g')(e'.\tau f'\iota g')^{-1}(e'.\tau f'\iota g') \\ (\tau e'\tau f'.g')(e'.\tau f'\tau g')^{-1}(\iota e'.f'.\tau g')^{-1} \\ \sim e(\tau e'.f'.\iota g')(\tau e'\tau f'.g')(e'.\tau f'\tau g')^{-1}(\iota e'.f'.\tau g')^{-1} \\ \sim e(\tau e'.f'.\iota g')(\tau e'\tau f'.g')(e'.\tau f'\tau g')^{-1} \\ (e'.\tau f'\tau g')(e'.\tau f'\tau g')^{-1}(e'.\iota f'\tau g')^{-1} \\ e(\tau e'.f'.\iota g')(\tau e'\tau f'.g')(\tau e'.f'.\tau g')^{-1}(e'.\iota f'\tau g')^{-1}$$

which is illustrated in Figures 5.2 and 5.3 (the reduction h_1 is on the left and h_2 is on the right).

2. Next suppose that e is disjoint from f and g, which arise from some critical pair. We can write w as $w = \iota e'w'$ for some $w' \in F$ and $e' \in e^+$ such that e = e'.w', and



 $\iota e'\iota f'\iota g' \longrightarrow \iota e'\iota f'\tau g'$











Figure 5.2: Resolution of type 1 overlap

....



Figure 5.3: Resolution of type 1 overlap (continued)

there are also edges $f', g' \in star^+(w')$ such that $f = \iota e'.f'$ and $g = \iota e'.g'$. By our choice of **p**, the edges f' and g' are resolved along positive paths $p = f_1 \dots f_m$ and $q = g_1 \dots g_n$ on the boundary of a 2-cell in \mathcal{D}^p to some vertex z. We may resolve this overlapping pair along positive homotopy reductions as follows (see figure 5.4):

$$h_1: g \rightsquigarrow e(\tau e'.g')(e'.\tau g')^{-1}$$
$$\rightsquigarrow e(\tau e'.f'pq^{-1})(e'.\tau g')^{-1}$$

and

$$\begin{split} h_2: \ g & \rightsquigarrow \ f(\iota e'.pq^{-1}) \\ & & \sim \ e(\tau e'.f')(e'.\tau f')^{-1}(\iota e.f_1f_2...f_mg_n^{-1}g_1^{-1}) \\ & & \sim \ e(\tau e'.f')(e'.\tau f')^{-1}(e'.\tau f')(\tau e'.f_1)(e'.\tau f_1)^{-1} \\ & & (\iota e.f_2...f_mg_n^{-1}g_1^{-1}) \\ & & \sim \ e(\tau e'.f')(e'.\tau f')^{-1}(e'.\tau f')(\tau e'.f_1)(e'.\tau f_1)^{-1} \\ & & (e'.\tau f_1)(\tau e'.f_2)(e'.\tau f_2)^{-1}(\iota e.f_3...f_mg_n^{-1}g_1^{-1}) \\ & & \sim^* \ e(\tau e'.f')(e'.\tau f')^{-1}(e'.\tau f')(\tau e'.f_1)(e'.\tau f_1)^{-1}(e'.\tau f_1) \\ & & (\tau e'.f_2)(e'.\tau f_2)^{-1}...(e'.\tau g_1)(\tau e'.g_1)^{-1}(e'.\iota g_1)^{-1} \\ & & \sim^* \ e(\tau e'.f')(\tau e'.f_1f_2...f_mg_n^{-1}...g_1^{-1})(e'.\iota g_1)^{-1} \\ & & = \ e(\tau e'.(f'pq^{-1}))(e'.\tau g')^{-1} \end{split}$$

3. Next suppose that g is disjoint from both e and f, which arise from some critical pair. We can write w as $w'\iota g'$ for some $w \in F$ and $g' \in e^+$ such that g = w'.g', and there are edges $e', f' \in star^+(w')$ such that $e = e'.\iota g'$ and $f = f'.\iota g'$. Again we suppose that the edges e' and f' at w' are resolved along positive paths



Figure 5.4: Resolution of type 2 overlap

 $p = e_1 \dots e_m$ and $q = f_1 \dots f_n$ on the boundary of a 2-cell to some vertex z. This overlapping pair can be can be resolved along the following positive homotopy reductions:

$$h_{1}: g \rightsquigarrow e(\tau e'.g')(e'.\tau g')^{-1}$$
$$\rightsquigarrow e(e_{1}.\iota g')(\tau e_{1}.g')(e_{1}.\tau g')^{-1}(e'.\tau g')^{-1}$$
$$\rightsquigarrow^{*} e(p.\iota g')(z.g')(p^{-1}.\tau g')(e'.\tau g')^{-1}$$

and

$$h_{2}: g \rightsquigarrow f(\tau f'.g')(f'.\tau g')^{-1}$$

$$\rightsquigarrow^{*} f(q.\iota g')(z.g')(q^{-1}.\tau g')(f'.\tau g')^{-1}$$

$$\rightsquigarrow^{*} e(pq^{-1}q.\iota g')(z.g')(q^{-1}.\tau g')(e'pq^{-1}.\tau g')^{-1}$$

$$\rightsquigarrow^{*} e(p.\iota g')(z.g')(p^{-1}.\tau g')(e'.\tau g')^{-1}$$

4. (critical overlap) Lastly suppose that none of the edges e, f or g are disjoint from the other two. Then we have the pair of homotopy reductions shown









in Figure 5.6 across 2-cells arising from the trivializer \mathbf{p} . Since the reduction system is complete this pair can be resolved, and in fact there is a pair of positive homotopy reductions

 $g \rightsquigarrow^* er.$

where r is the irreducible path which we obtain by performing homotopy reductions on the path pq^{-1} .

Remark 5.2.4 Note that the critical overlaps of this homotopy reduction system correspond precisely to the critical triples (Definition 4.4.2) of the original rewriting system, together with their translates under the two-sided action of F.

5.2.2 Homology

In [37], the critical overlaps which are not translates of others under the two-sided action of F are called the *substantial critical pairs* of the homotopy reduction system. It is shown that in the case when the homotopy reduction system obtained from \mathbf{p} is complete, we can extend the partial resolution (1.9) to dimension 4:

$$\mathbb{Z}S.\mathbf{c} \xrightarrow{\partial_4} \mathbb{Z}S.\mathbf{p} \xrightarrow{\partial_3} \mathbb{Z}S.\mathbf{r} \xrightarrow{\partial_2} \mathbb{Z}S.\mathbf{x} \xrightarrow{\partial_1} \mathbb{Z}S \xrightarrow{\partial_0} \mathbb{Z} \longrightarrow 0$$
(5.1)

where c denotes, as above, the set of substantial critical pairs of the homotopy reduction system.

Since the substantial critical pairs correspond to the set of critical triples of the rewriting system, we have the corollary that a monoid with a finite complete rewriting system is of type left FP_4 (and also right FP_4). While this is clearly a weaker result than that given in the Anick-Groves-Squier Theorem (Theorem 1.4.4), it is still interesting in itself: in (5.1) the basis **p** corresponds to the critical pairs of the rewriting system, whereas the basis of the free module one dimension higher corresponds to the "critical pairs" of a higher dimensional rewriting system.

We shall not discuss this in detail; we just mention it as an interesting alternative to our method of extending the partial resolution (1.9) to dimension 4 in Chapter 3 by introducing a new 3-complex based on \mathcal{D} and studying its second homology.

Appendix A Appendix

A.1 Abels' group

Abels' group [1] is a finitely presented group which is not of type FP_3 , and so by the Anick-Groves-Squier Theorem (Theorem 1.4.4) it has no finite complete rewriting system. We begin this subsection by showing that one of its subgroups does have a finite complete rewriting system, and we can use this as a base to give a finite λ -complete rewriting system for an HNN-group.

Using the abbreviation [x, y] for the word $x^{-1}y^{-1}xy$,

is a presentation for the group of matrices with integer entries and 1's on the diagonal [1]. We can add by a type 1 Tietze transformation the new generator t = [a, d] = [b, e]. These generators correspond to matrices with 1's on the diagonal and a 1 in the entry labelled by the corresponding capital letter:

$$\left(\begin{array}{rrrrr} 1 & A & B & T \\ & 1 & C & D \\ & & 1 & E \\ & & & 1 \end{array}\right)$$

so that, for example, we identify \overline{b} with the matrix

$$\left(\begin{array}{rrrr}1 & 0 & 1 & 0\\ & 1 & 0 & 0\\ & & 1 & 0\\ & & & 1\end{array}\right)$$

From this we can derive the rewriting system $\mathcal{P}_{\mathcal{H}}$ for \mathcal{H} , consisting of the alphabet

$$a, a^{-1}, b, b^{-1}, c, c^{-1}, d, d^{-1}, e, e^{-1}, t, t^{-1}, d, d^{-1}, e, e^{-1}, t, t^{-1}, d, d^{-1}, d, d^{-1}$$

together with the following set of rules:

where the superscripts \ddagger and \flat may take the values -1 or +1 on each side of a rule, together with the set of rules of the form

$$x^{\varepsilon}x^{-\varepsilon} = 1 \qquad (\varepsilon = \pm 1, x \in \{\mathbf{g}, b, c, d, e, t\})$$

describing free reductions.

Proposition A.1.1 The rewriting system $\mathcal{P}_{\mathcal{H}}$ for is complete.

Proof: The rules have been chosen in order that any word on the generators is rewritten in a finite number of steps to a word of the form

$$a^{\alpha}b^{\beta}c^{\gamma}t^{\tau}d^{\delta}e^{\epsilon} \tag{A.1}$$

where $\alpha, \beta, \gamma, \delta, \epsilon, \tau \in \mathbb{Z}$. To show that it is noetherian we can check that the rules are compatible with recursive path order from the left (Definition 1.1.3) induced by the partial order

 $e^{\pm 1} \triangleright d^{\pm 1} \triangleright t^{\pm 1} \triangleright c^{\pm 1} \triangleright b^{\pm 1} \triangleright a^{\pm 1}$

on the alphabet. For example, $ca \succ ab^{-1}c$ holds because $c \triangleright a$ and $ca \succ b^{-1}c$, the latter being true because $c \triangleright b^{-1}$ and $ca \succ c$, again the latter holding because $a \succ 1$.

The fact that $\mathcal{P}_{\mathcal{H}}$ is complete follows easily by checking that all the critical pairs are resolved. For example caa^{-1} can be rewritten to either c or $ab^{-1}ca^{-1}$, thus forming a critical pair which can be resolved by the path

$$ab^{-1}\underline{ca}^{-1} \to a\underline{b}^{-1}\underline{a}^{-1}bc \to \underline{aa}^{-1}b^{-1}bc \to \underline{b}^{-1}\underline{b}c \to c.$$

Remark A.1.2 If we wanted to verify that $\mathcal{P}_{\mathcal{H}}$ is a rewriting system for the group of matrices we are studying we would also need to show that each matrix can be represented by a word of the form (A.1) - this is not difficult to check. The standard presentation of the group of 3×3 upper triangular integral matrices with 1's on the diagonal (the *integral Heisenberg group*) is derived in a similar manner in [33, §5.4].

We want to use $\mathcal{P}_{\mathcal{H}}$ to produce a λ -complete rewriting system by forming an HNN extension.

Lemma A.1.3 The subgroups $\langle a_{\mathbf{p}}b, c^{\mathbf{p}}, e \rangle$ and $\langle a^{\mathbf{p}}, b, c, e \rangle$ of H are isomorphic.

Proof: Any word on the generators a, b, c^p and e will be rewritten to an irreducible word of the form

$$a^{\alpha}b^{\beta}c^{p\gamma}t^{\tau}d^{p\delta}e^{\epsilon}$$

where $\alpha, \beta, \gamma, \delta, \epsilon, \tau \in \mathbb{Z}$, whereas any word on the generators a^p , b, c and e will be rewritten to an irreducible word of the form

$$a^{plpha}b^{eta}c^{\gamma}t^{ au}d^{\delta}e^{\delta}$$

۲

122

where $\alpha, \beta, \gamma, \delta, \epsilon, \tau \in \mathbb{Z}$.

It is then straightforward to show that these subgroups are, respectively, the groups of matrices of the form

$$\begin{pmatrix} 1 & * & * & * \\ 1 & p* & p* \\ & 1 & * \\ & & 1 \end{pmatrix} \quad \text{and} \quad \begin{pmatrix} 1 & p* & * & * \\ & 1 & * & * \\ & & 1 & * \\ & & & 1 \end{pmatrix}$$

with integers in the entries marked with a *. There is an isomorphism

$$\langle a, b, c^p, e \rangle \longrightarrow \langle a^p, b, c, e \rangle$$

given by

$$\begin{pmatrix} 1 & \alpha & \beta & \tau \\ & 1 & p\gamma & p\delta \\ & & 1 & \epsilon \\ & & & 1 \end{pmatrix} \mapsto \begin{pmatrix} 1 & p\alpha & \beta & \tau \\ & 1 & \gamma & \delta \\ & & 1 & \epsilon \\ & & & 1 \end{pmatrix}$$

again where $\alpha, \beta, \gamma, \delta, \epsilon, \tau \in \mathbb{Z}$.

We can then form the HNN-extension

$$G = \langle H \mid x \; ; \; x^{-1}ax = a^p, [x, b], \; xcx^{-1} = c^p, \; [x, e] \rangle$$

with base group H and stable letter x associating the subgroups $\langle a, b, c^p, e \rangle$ and $\langle a^p, b, c, e \rangle$ according to the isomorphism described in Lemma A.1.3. We can extend $\mathcal{P}_{\mathcal{H}}$ to obtain a rewriting system \mathcal{P}_G for G by adding the letters x, x^{-1} , together with the set of rules

(again \sharp and \flat may take the values +1 or -1).

*

Г

Proof: First, the rewriting system noetherian, since it is compatible with the recursive path ordering from the left induced by the partial order

$$a^{\pm 1} \triangleleft b^{\pm 1} \triangleleft c^{\pm 1} \triangleleft t^{\pm 1} \triangleleft d^{\pm 1} \triangleleft e^{\pm 1} \triangleleft x^{\pm 1}.$$

Now suppose that w is a non-empty irreducible word such that $\overline{w} = 1$. Since \mathcal{P}_H is complete, we can assume that w contains instances of the letters x, x^{-1} . By Brittons' Lemma (Lemma 2.2.2) w must contain a subword of the form

$$x^{-1}a^{\alpha}b^{\beta}c^{p\gamma}t^{\tau}d^{p\delta}e^{\epsilon}x$$

or of the form

$$xa^{p\alpha}b^{\beta}c^{\gamma}t^{\tau}d^{\delta}e^{\epsilon}x^{-1},$$

where $\alpha, \beta, \gamma, \delta, \epsilon, \tau \in \mathbb{Z}$, and this contradicts the assumption that w is an irreducible word.

Remark A.1.5 Note that $\mathcal{P}_{\mathcal{G}}$ is not complete. For example, the two single edges

 $\underline{xca} \rightarrow c^p xa$ and $\underline{xca} \rightarrow xab^{-1}c$

rewrite the word xca to two distinct irreducible words.

Remark A.1.6 Abels [1] has shown that the group with the presentation

$$\mathcal{A} = \langle \mathcal{G} \mid z \; ; \; z^{-1}cz = c^p, \; zcz^{-1} = c^p \rangle$$

is not of type FP_3 , and therefore by the Anick-Groves-Squier Theorem it has no finite complete rewriting system. On the other hand \mathcal{A} is a presentation of the group of matrices of the form

where the entries marked with * take values in the ring of rational numbers with denominator a power of p, and with positive units on the diagonal. As such, the group has solvable word problem.

Exhibiting a finite λ -complete rewriting system for this group would therefore give an example of a group with such a rewriting system but without any finite complete rewriting system, thereby answering Open Question 2.1.8. Unfortunately, \mathcal{A} does not define an HNN-extension with base group the group presented by \mathcal{G} , and so we cannot use Britton's Lemma to derive a λ -complete rewriting system for Abels' group using $\mathcal{P}_{\mathcal{G}}$. This statement can be justified by considering the word $w = [a, [ab, z^{-1}cz]]$ which is not equivalent to 1 in \mathcal{G} . In \mathcal{A} , however, it is equivalent to 1, since because z corresponds to the matrix

 $\left(\begin{array}{rrrr} 1 & 0 & 0 & 0 \\ 1 & 0 & 0 \\ & 1/p & 0 \\ & & 1 \end{array}\right)$

w describes the commutation of the matrices

$$\left(egin{array}{ccccc} 1 & 1 & 0 & 0 \ 1 & 0 & 0 \ & & 1 & 0 \ & & & 1 \end{array}
ight) \qquad ext{and} \qquad \left(egin{array}{ccccccc} 1 & 0 & 1/p & 0 \ 1 & 0 & 0 \ & & 1 & 0 \ & & & 1 \end{array}
ight).$$

Bibliography

- H. ABELS, An example of a finitely presented solvable group, in: Homological Group Theory, C. T. C. Wall, ed., no. 36 in LMS Lecture Note Series, CUP, Cambridge, 1979, pp. 205-211.
- [2] J. ALONSO AND S. HERMILLER, Homological finite derivation type, preprint, Department of Mathematics, University of Nebraska, Lincoln NE, 2000. (available at: http://www.math.unl.edu/~smh/ppt/hfdt.ps).
- [3] D. J. ANICK, On the homology of associative algebras, Trans. Amer. Math. Soc., 296 (1986), pp. 641–659.
- [4] G. BAUMSLAG, R. BIERI, C. F. MILLER III, AND H. SHORT, Finitely presented subgroups of automatic groups and their isoperimetric functions, J. London Math. Soc., 56 (1997), pp. 292-304.
- [5] R. BIERI, Homological Dimension of Discrete Groups, Queen Mary College Mathematics Notes, 1976.
- [6] W. A. BOGLEY AND S. J. PRIDE, Calculating generators of π₂, in: Two-Dimensional Homotopy and Combinatorial Group Theory, C. Hog-Angeloni, W. Metzler, and A. J. Sieradski, eds., vol. 197 of LMS Lecture Note Series, CUP, 1992, pp. 157–188.
- [7] R. BOOK AND F. OTTO, String-Rewriting Systems, Springer-Verlag, New York, 1993.

- [8] R. V. BOOK, Thue systems as rewriting systems, J. Symbolic Comp., 3 (1987), pp. 39-68.
- [9] W. W. BOONE, The word problem, Ann. of Math., 70 (1959), pp. 207–265.
- [10] J. L. BRITTON, The word problem, Ann. of Math., 77 (1963), pp. 16-32.
- [11] K. BROWN, Cohomology of Groups, Springer-Verlag, New York, 1982.
- [12] K. S. BROWN, The geometry of rewriting systems: a proof of the Anick-Groves-Squier theorem, in: Proceedings of the Workshop on Algorithms, Word Problems and Classification in Combinatorial Group Theory, vol. 23 of MSRI series, Springer, New York, 1992, pp. 125–178.
- [13] D. E. COHEN, A monoid which is right FP_{∞} but not left FP_1 , Bull. London Math. Soc., 24 (1992), pp. 340-342.
- [14] —, String rewriting a survey for group theorists, in: Geometric Group Theory, Volume 1, G. A. Niblo and M. A. Roller, eds., no. 181 in LMS Lecture Notes, CUP, 1993, pp. 37-47.
- [15] D. J. COLLINS, R. I. GRIGORCHUK, P. F. KURCHANOV, AND H. ZIESCHANG, Combinatorial Group Theory and Applications to Geometry, Springer, Berlin, 1998.
- [16] R. CREMANNS, Finiteness conditions for re-writing systems, PhD thesis, Universität-Gh Kassel, 1995.
- [17] R. CREMANNS AND F. OTTO, Finite derivation type implies the homological finiteness condition FP₃, J. Symbolic Comput., 18 (1994), pp. 91–112.
- [18] —, For groups the property of having finite derivation type is equivalent to the homological finiteness condition FP₃, J. Symbolic Comput., 22 (1996), pp. 155– 177.

- [19] M. DEHN, Uber die Topology des dreidimensional Raumes, Math. Ann., 69 (1910), pp. 137-168.
- [20] D. V. DEKOV, Finite complete rewriting systems for groups, Comm. in Alg., 25 (1997), pp. 4023-4028.
- [21] N. DERSHOWITZ, Orderings for term-rewriting systems, Th. Comput. Science, 17 (1982), pp. 279–301.
- [22] D. B. A. EPSTEIN ET AL, Word Processing in Groups, Jones and Bartlett, New York, 1992.
- [23] P. FREYD, D. YETTER, J. HOSTE, W. LICKORISH, K. MILLET, AND A. OC-NEANU, A new polynomial invariant of knots and links, Bull. Amer Math. Soc., 12 (1985), pp. 239-246.
- [24] S. M. GERSTEN, Dehn functions and l₁-norms of finite presentations, in: Algorithms and Classification in Combinatorial Group Theory, vol. 23 of MSRI series, Springer, New York, 1995, pp. 15–31.
- [25] M. GREENDLINGER, Dehn's algorithm for the word problem, Comm. Pure and Appl. Math., 13 (1960), pp. 67-83.
- [26] M. GROMOV, Hyperbolic groups, in: Essays in Group Theory, vol. 8 of MSRI Series, Springer, New York, 1987, pp. 75-263.
- [27] J. R. J. GROVES, Rewriting systems and homology of groups, in: Groups -Canberra 1989, L. G. Kovacs, ed., vol. 1456 of Lecture Notes in Mathematics, Springer, Berlin, 1991, pp. 114-141.
- [28] J. R. J. GROVES AND G. C. SMITH, Soluble groups with a finite rewriting system, Proc. Edin. Math. Soc., 36 (1993), pp. 283-288.
- [29] V. GUBA AND M. SAPIR, Diagram groups, Memoirs Amer. Math. Soc., 620 (1997), pp. 1–117.

- [30] G. HIGMAN, B. H. NEUMANN, AND H. NEUMANN, Embedding theorems for groups, J. London Math. Soc., 24 (1949), pp. 247–254.
- [31] J. HUEBSCHMANN, Aspherical 2-complexes and an unsettled problem of J. H. C. Whitehead, Math. Ann., 258 (1981), pp. 17–37.
- [32] S. V. IVANOV, Relation modules and relation bimodules of groups, semigroups and associative algebras, Int. J. Algebra and Comput., 1 (1991), pp. 89–114.
- [33] D. L. JOHNSON, Presentations of Groups, CUP, Cambridge, 1990.
- [34] V. KILIBARDA, On the algebra of semigroup diagrams, Int. J. Algebra and Comput., 7 (1997), pp. 313–338.
- [35] D. KNUTH AND P. BENDIX, Simple word problems in universal algebra, in: Computational Problems in Abstract Algebra, J. Leech, ed., Pergamon Press, New York, 1970, pp. 263-297.
- [36] Y. KOBAYASHI, Complete rewriting systems and homology of monoid algebras,
 J. Pure Appl. Algebra, 65 (1990), pp. 263-275.
- [37] —, Homotopy reduction systems for monoid presentations: Asphericity and low-dimensional homology, J. Pure Appl. Algebra, 130 (1998), pp. 159–195.
- [38] —, Finite homotopy bases of one-relator monoids, J. Algebra, 229 (2000), pp. 547-569.
- [39] —, Homotopy reduction systems for monoid presentations II: the Guba-Sapir reduction and homotopy modules, in: Algorithmic Problems in Groups and Semigroups, J. C. Birget, S. Margolis, J. Meakin, and M. Sapir, eds., Birkhäuser, Boston, 2000, pp. 143-159.
- [40] Y. KOBAYASHI AND F. OTTO, Some exact sequences for the homotopy (bi-) module of a monoid, Mathematische Schriften 18, Universität-GH Kassel, 2000 (Int. J. Alg. and Comp., to appear).

••

- [41] —, For finitely presented monoids the homological finiteness conditions FHT and bi-FP₃ coincide, Mathematische Schriften 4, Universität-GH Kassel, 2001.
- [42] ____, On homotopical and homological finiteness conditions for finitely presented monoids, Int. J. Algebra and Comput., 11 (2001), pp. 391-403.
- [43] Y. LAFONT, A finiteness condition for monoids presented by complete rewriting systems, J.Pure Appl.Alg., 98 (1995), pp. 229-244.
- [44] R. C. LYNDON AND P. E. SCHUPP, Combinatorial Group Theory, Springer, New York, 1977.
- [45] K. MADLENER AND F. OTTO, Groups presented by certain classes of finite length-reducing string rewriting systems, in: Rewriting Techniques and Applications (Bordeaux 1987), P. Lescanne, ed., vol. 256 of Lecture Notes in Computer Science, Springer, Berlin, 1987, pp. 133-144.
- [46] W. MAGNUS, Das identitätsproblem für Gruppen mit einer definierenden Relation, Math. Ann., 106 (1932), pp. 295–307.
- [47] W. MAGNUS, A. KARRASS, AND D. SOLITAR, Combinatorial Group Theory, Dover, New York, 2 ed., 1976.
- [48] C. R. F. MAUNDER, Algebraic Topology, Van Nostrand Reinhold, London, 1970.
- [49] M. H. A. NEWMAN, On theories with a combinatorial definition of equivalence, Ann. Math., 43 (1942), pp. 223-243.
- [50] P. S. NOVIKOV, On the algorithmic unsolvability of the word problem, Trud Mat. Inst. Styklov, 44 (143), p. 1955.
- [51] F. OTTO, Finite complete rewriting systems for the Jantzen monoid and for the Greendlinger group, Th. Comput. Sc., 32 (1984), pp. 249-260.

**

- [52] F. OTTO, A. SATTLER-KLEIN, AND K. MADLENER, Automatic monoids versus monoids with finite convergent presentations, in: Rewriting Techniques and Applications (Tsukuba, 1998), vol. 1379 of Lecture Notes in Computer Science, Springer, Berlin, 1998, pp. 32-46.
- [53] C. D. PAPAKYRIAKOPOLOUS, On Dehn's lemma and the asphericity of knots, Ann. of Math., 66 (1957), pp. 1–26.
- [54] S. J. PRIDE, Low-dimensional homotopy theory for monoids, Int. J. Algebra and Comput., 5 (1995), pp. 631–649.
- [55] —, Low-dimensional homotopy theory for monoids II: Groups, Glas. J. Math., 41 (1999), pp. 1–11.
- [56] S. J. PRIDE AND J. WANG, Rewriting systems, finiteness conditions, and associated functions, in: Algorithmic Problems in Groups and Semigroups, J. M. J. C. Birget, S. Margolis and M. Sapir, eds., Birkhäuser, Boston, 2000, pp. 195–216.
- [57] J. J. ROTMAN, An Introduction to the Theory of Groups, Springer, New York, 1994.
- [58] H. SHORT (ED.), Notes on word hyperbolic groups, in: Group theory from a geometric viewpoint, E. Ghys, A. Haefliger, and A. Verjovski, eds., ICTP Trieste, World Scientific, Singapore, 1991, pp. 3-64.
- [59] C. C. SQUIER, Word problems and a homological finiteness condition for monoids, J. Pure Appl. Algebra, 49 (1987), pp. 201-216.
- [60] C. C. SQUIER, F. OTTO, AND Y. KOBAYASHI, A finiteness condition for rewriting systems, Theoret. Comput. Sci., 131 (1994), pp. 271–294.
- [61] J. R. STALLINGS, A finitely presented group whose 3-dimensional integral homology is not finitely generated., Amer. J. Math, 85 (1963), pp. 541-543.

- [62] J. STILLWELL, Classical Topology and Combinatorial Group Theory, Springer, New York, 1980.
- [63] G. S. TSEJTIN, Associative calculation with an unsolvable equivalence problem, Trudy Mat. Inst. Steklova, 52 (1958), pp. 172–89.
- [64] F. WALDHAUSEN, The word problem in fundamental groups of sufficiently large irreducible 3-manifolds, Ann. of Math., 63 (1968), pp. 272-280.
- [65] J. WANG, Rewriting Systems, Finiteness Conditions and Second Order Dehn Functions of Monoids, PhD thesis, Department of Mathematics, University of Glasgow, 1998.
- [66] X. WANG AND S. J. PRIDE, Second order dehn functions of groups and monoids, Int. J. Algebra Comput., 10 (2000), pp. 425-456.